Securities and Exchange Commission

Washington, DC 20549

Form 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported) June 18, 2004

PERFICIENT, INC.

(Exact Name of Registrant as Specified in its Charter)

Delaware

001-15169

74-2853258

(State or Other Jurisdiction of Incorporation)

(Commission File Number)

(IRS Employer Identification No.)

1120 South Capital of Texas Highway, Suite 220, Building 3, Austin, TX

(Address of Principal Executive Offices)

78746 (Zip Code)

Registrant's telephone number, including area code (512) 531-6000

(Former Name or Former Address, if Changed Since Last Report)

ITEM 2. ACQUISITION OR DISPOSITION OF ASSETS

On June 18, 2004, Perficient, Inc. ("Perficient") consummated the acquisition of Meritage Technologies, Inc. ("Meritage"), a Delaware corporation, by merging our wholly owned subsidiary, Perficient Meritage, Inc., a Delaware corporation with and into Meritage. Meritage survived the merger as our direct wholly owned subsidiary under the name "Perficient Meritage, Inc."

Perficient paid approximately \$7.1 million to the selling shareholders consisting of approximately \$2.9 million in cash and up to \$4.2 million worth of Perficient common stock subject to certain post-closing adjustments. A portion of the shares of common stock issued in the merger are subject to certain conditions in the first three years following the closing of the transaction. Perficient will also assume \$2.4 million of debt, \$1.5 million of which is related to an accounts receivable credit facility that will be retired upon the closing of the acquisition. The operating management of Meritage will receive only Perficient common stock and Perficient stock options in the transaction. The shares of common stock issued in connection with the merger were ascribed a value of \$3.595 per share, which was the average closing price of the Company's common stock for the 23 consecutive trading days ending on June 15, 2004. The merger was funded in part by a private placement of Perficient common stock and in part by an expansion of our credit facility with Silicon Valley Bank.

Prior to the acquisition, the assets of Meritage were used to provide information technology consulting services to its customers. The Company intends to continue such uses for the assets of Meritage.

The foregoing summary does not purport to be complete and is qualified in its entirety by reference to the full text of the Meritage Agreement and Plan of Merger, a copy of which is included herein as Exhibit 2.1.

In connection with the acquisition, on June 16, 2004 Perficient raised approximately \$2.5 million through a private placement of 800,000 shares of Perficient common stock at \$3.09 per share to a group of institutional investors led by Tate Capital Partners. The investors were also issued warrants for the purchase of an additional 160,000 shares of Perficient common stock, exercisable over the next 24 months at a purchase price of \$4.64 per share. The additional estimated proceeds to Perficient if all warrants are exercised is expected to be approximately \$742,400. A copy of the full text of the Securities Purchase Agreement is included herein as Exhibit 10.1.

In addition, Perficient entered into a Loan Modification Agreement effective June 8, 2004 with Silicon Valley Bank to establish a new acquisition credit facility. Funds drawn under the acquisition facility will be used to finance the cash portion of accretive acquisitions. The new facility increases the total size of the Company's Silicon Valley Bank credit facilities from \$6 million to \$10 million. The Company has drawn \$2.5 million under the new acquisition credit facility in connection with the acquisition of Meritage. A copy of the full text of the Loan Modification Agreement is included herein as Exhibit 10.2.

The press release announcing the acquisition and the related financing is included herein as Exhibit 99.1 to this Form 8-K.

2

Financial Statements relating to the acquisition described in this Form 8-K and required pursuant to Rule 3-05 of Regulation S-X are not included herein but will be filed by an amendment to this Form 8-K within sixty (60) days from July 3, 2004.

(b) PRO FORMA FINANCIAL INFORMATION

Pro forma financial information relating to the acquisitions described in this Form 8-K and required pursuant to Article 11 of Regulation S-X are not included herein but will be filed by an amendment to this Form 8-K within sixty (60) days from July 3, 2004.

(c) EXHIBITS

EXHIBIT NO.	DESCRIPTION
2.1*	Agreement and Plan of Merger, dated as of June 18, 2004, by and among Perficient, Inc., Perficient Meritage, Inc., Meritage Technologies, Inc., and Robert Honner, as Stockholder Representative.
10.1*	Securities Purchase Agreement, dated as of June 16, 2004, by and among Perficient, Inc., Tate Capital Partners Fund, LLC, Pandora Select Partners, LP, and Sigma Opportunity Fund, LLC.
10.2*	Loan Modification Agreement, dated as of June 8, 2004, by and among Perficient, Inc., Perficient Canada Corp. and Silicon Valley Bank.
99.1	Perficient, Inc. Press Release issued on June 21, 2004 regarding the acquisition of Meritage Technologies, Inc.
*	Perficient hereby agrees to furnish supplementally to the Securities and Exchange Commission upon request a copy of any exhibit or schedule omitted from Exhibits 2.1, 10.1 and 10.2 filed herewith.
	3

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated June 23, 2004

PERFICIENT, INC.

/s/ Michael D. Hill Michael D. Hill Chief Financial Officer Dated June 23, 2004

AGREEMENT AND PLAN OF MERGER

By and Among

PERFICIENT, INC.

PERFICIENT MERITAGE, INC.

MERITAGE TECHNOLOGIES, INC.

and

ROBERT HONNER, as STOCKHOLDER REPRESENTATIVE

Dated as of June 18, 2004

TABLE OF CONTENTS

ARTICLE I THE MERGER

<u>1.01</u>	The Merger
1.02	Plan of Merger
1.03	Effective Time
1.04	Effect of the Merger
1.05	Conversion of Company Preferred Stock; Cancellation of Company Common Stock
1.06	Conversion of Merger Sub Stock
1.07	Cash Holdback Amount; Stock Holdback Amount; Escrowed Shares
1.08	Certificate of Incorporation
1.09	<u>Bylaws</u>
1.10	Officers and Directors
	ARTICLE II
	PAYMENT OF MERGER CONSIDERATION
2.01	Exchange of Shares
2.02	Stock Restriction Agreements
2.03	No Assumption of Company Stock Options; Termination of Company Warrants
2.04	Adjustment of Merger Consideration
2.05	Dissenting Shares
	ARTICLE III
	REPRESENTATIONS AND WARRANTIES OF THE COMPANY
3.01	Corporate Organization and Qualification
3.02	Capitalization
3.03	Authority; No Violation
3.04	Consents and Approvals
<u>3.05</u>	<u>Financial Statements</u>
<u>3.06</u>	Absence of Certain Changes or Events
<u>3.07</u>	<u>Legal Proceedings</u>
<u>3.08</u>	Taxes and Tax Returns
<u>3.09</u>	Employee Benefit Plans
<u>3.10</u>	Compliance with Applicable Law; Certain Agreements; Licensing
<u>3.11</u>	<u>Certain Contracts</u>
3.12	Agreements with Regulatory Agencies
<u>3.13</u>	Environmental Matters
<u>3.14</u>	<u>Properties</u>
<u>3.15</u>	<u>Insurance</u>
<u>3.16</u>	<u>Labor Matters</u>
<u>3.17</u>	<u>Intellectual Property</u>
<u>3.18</u>	Broker's Fees
3.19	Bank Accounts

i

	ARTICLE IV
	REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB
<u>4.01</u>	Corporate Organization and Qualification
<u>4.02</u> <u>4.03</u>	Capitalization Authority; No Violations
<u>4.04</u>	Consents and Approvals
<u>4.05</u> <u>4.06</u>	Broker's Fees No Prior Activities
4.07 4.08	Labor Matters Reports; Financial Statements
<u>4.09</u>	No Undisclosed Material Liabilities
<u>4.10</u> <u>4.11</u>	<u>Legal Proceedings</u> <u>Continued Listing of Stock</u>
4.12	Valid Issuance of Parent Common Stock
	ARTICLE V COVENANTS RELATING TO CONDUCT OF BUSINESS
<u>5.01</u>	Covenants of the Company
<u>5.02</u> <u>5.03</u>	Covenants of Parent and Merger Sub All Necessary Action
<u>5.03</u>	Notification Notification
	ARTICLE VI
	ADDITIONAL AGREEMENTS
<u>6.01</u> <u>6.02</u>	Regulatory Matters Securities Matters
<u>6.03</u>	Stockholder Approval; Exemption from Registration
6.04 6.05	Access to Information; Confidentiality No Solicitation of Transactions
<u>6.06</u>	<u>Legal Conditions to Merger</u>
6.07 6.08	<u>Disclosure Supplements</u> <u>Tax Matters</u>
6.09 6.10	Employment Agreements Company Employees
<u>6.11</u>	Plans and Stock Option Grants
6.12 6.13	Non-Compete Agreements Publicity
6.14	<u>Indemnification</u>
6.15 6.16	<u>Termination of Company Warrants, Stock Agreement and Profit Sharing Plan</u> <u>Obligations of Surviving Corporation</u>
	ARTICLE VII
	CONDITIONS PRECEDENT
<u>7.01</u>	Conditions to Each Party's Obligation to Effect the Merger
	ii
<u>7.02</u>	Conditions to Obligations of Parent and Merger Sub
<u>7.03</u>	Conditions to Obligations of the Company
	ARTICLE VIII TERMINATION AND AMENDMENT
8.01 8.02	Termination Effect of Termination
8.02 8.03	Effect of Termination Expenses
8.04 8.05	Amendment Extension; Waiver
	ARTICLE IX INDEMNIFICATION
9.01 9.02	Agreement to Indemnify Survival of Indemnity
9.03	Additional Provisions
<u>9.04</u>	Claim Notice; Definitions; Third Party Claim Procedures

<u>3.20</u>

<u>Disclosure</u>

ARTICLE X STOCKHOLDERS REPRESENTATIVE

GENERAL PROVISIONS

10.01	Appointment of Stockholders Representative
10.02	<u>Authority</u>
10.03	Reliance
10.04	<u>Indemnification of Parent, Merger Sub and Their Affiliates</u>
<u>10.05</u>	Indemnification of Stockholder Representative

ARTICLE XI

11.01 **Notices** 11.02 **Interpretation** 11.03 **Counterparts** 11.04 **Entire Agreement Governing Law** 11.05 Enforcement of Agreement 11.06 Severability 11.07 11.08 **Assignment**

Amendment

11.09

iii

EXHIBIT LIST

EXHIBIT A Form of Stockholder Proxy **EXHIBIT B** Purchase Price Certificate **EXHIBIT C** Form of Escrow Agreement **EXHIBIT D** Form of Letter of Transmittal **EXHIBIT E** Form of Stockholder Representation Letter Form of Stock Restriction Agreement **EXHIBIT F EXHIBIT G** Form of Employment Agreement **EXHIBIT H** Form of Confidentiality and Intellectual Property Assignment Agreement **EXHIBIT I** Form of Stock Option Agreement **EXHIBIT J** Form of Non-Compete Agreement EXHIBIT K Form of Opinion of Counsel to the Company iv

AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER (the "Agreement") dated as of June 18, 2004, by and among Perficient, Inc., a Delaware corporation ("Parent"), Perficient Meritage, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("Merger Sub"), Meritage Technologies, Inc., a Delaware corporation (the "Company"), and Robert Honner ("Stockholder Representative").

WHEREAS, the respective Boards of Directors of Parent, Merger Sub and the Company have determined that it is advisable and in the best interests of their respective companies and their Stockholders to consummate the business combination transaction provided for herein in which the Merger Sub will, subject to the terms and conditions set forth herein, merge with and into the Company (the "Merger");

WHEREAS, as an inducement to Parent's and Merger Sub's entering into this Agreement and incurring the obligations set forth herein, certain Company Preferred Stockholders (the "Principal Stockholders") have executed a proxy in the form attached hereto as Exhibit A (the "Stockholder Proxy") in favor of Christopher Gianattasio who has delivered on the date hereof the actions by written consent of the stockholders of the Company (the "Stockholder") Consent") attached to the Stockholder Proxy; and

WHEREAS, Parent, Merger Sub and the Company desire to make certain representations, warranties and covenants in connection with the Merger.

NOW, THEREFORE, in consideration of the mutual covenants, representations, warranties and agreements contained herein, and intending to be legally bound hereby, the parties agree as follows:

ARTICLE I THE MERGER

- **1.01 The Merger.** Subject to the terms and conditions of this Agreement, in accordance with the Delaware General Corporation Law ("*DGCL*"), at the Effective Time (as hereinafter defined), the Merger Sub shall merge with and into the Company. The Company shall be the surviving corporation (hereinafter sometimes called the "*Surviving Corporation*") in the Merger, and shall continue its corporate existence under the laws of the State of Delaware. The name of the Surviving Corporation shall be "Perficient Meritage, Inc.", a Delaware corporation. Upon consummation of the Merger, the separate corporate existence of the Merger Sub shall terminate.
 - **1.02 Plan of Merger.** This Agreement shall constitute an agreement of merger for purposes of the DGCL.
- **1.03 Effective Time.** As promptly as practicable, but in no event later than the third (3rd) business day after all of the conditions set forth in <u>ARTICLE VII</u> shall have been satisfied or waived by the party or parties entitled to the benefit of the same, the Company and Merger

1

Sub shall duly execute and file a certificate of merger (the "Certificate of Merger") with the Secretary of State of the State of Delaware (the "Delaware Secretary") in accordance with the DGCL. The Merger shall become effective on the date (the "Effective Date" or the "Closing Date") and at the time (the "Effective Time") as the Certificate of Merger is filed with the Delaware Secretary or at such later date and time as is agreed in writing by the parties and specified in such Certificate of Merger. Subject to the terms and conditions of this Agreement, the closing of the Merger (the "Closing") shall be held at the offices of Vinson & Elkins L.L.P., Terrace 7, 2801 Via Fortuna, Suite 100, Austin, Texas 78746 or such other location as the parties may mutually agree upon.

1.04 Effect of the Merger. At the Effective Time, the effect of the Merger shall be as provided herein and as set forth in Section 259 of the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, (a) all the property, rights, privileges, powers and franchises of Merger Sub and the Company shall vest in the Surviving Corporation, (b) all debts, liabilities, obligations, restrictions, disabilities and duties of Merger Sub and the Company shall become the debts, liabilities, obligations, restrictions, disabilities and duties of the Surviving Corporation and (c) the Surviving Corporation shall become a wholly-owned subsidiary of Parent.

1.05 Conversion of Company Preferred Stock; Cancellation of Company Common Stock.

- (a) At the Effective Time, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company or the holders of any of the following securities:
- (i) Each issued and outstanding share of Series B Convertible Preferred Stock, par value \$0.001 per share, (the "*Company Series B Preferred Stock*"), shall convert into the right to receive and be exchangeable for the Series B Merger Consideration divided by the total number of shares of Company Series B Preferred Stock issued and outstanding immediately prior to the Effective Time;
- (ii) Each issued and outstanding share of Series A Convertible Preferred Stock, par value \$0.001 per share, (the "Company Series A Preferred Stock," and, together with the Company Series B Preferred Stock, the "Company Preferred Stock"), shall convert into the right to receive and be exchangeable for the Series A Merger Consideration divided by the total number of shares of Company Series A Preferred Stock issued and outstanding immediately prior to the Effective Time; and
- (iii) Each issued and outstanding share of Common Stock, par value par value \$0.001 per share, (the "Company Common Stock," and, together with the Company Preferred Stock, the "Company Securities"), shall be cancelled and extinguished without any conversion thereof and no payment or distribution shall be made with respect thereto.
 - (b) For purposes of this Agreement, the following terms have the following meanings:

5

- (i) "Closing Date Debt" means, without duplication, (A) subject to final determination based on the Final Closing Date Balance Sheet, the amount of all liabilities of the Company for borrowed money, including, but not limited to, any Company-guaranteed payment obligations (except for the Standby Letter of Credit #S030367300 issued by KeyBank National Association), and any other long term liabilities of the Company as specifically set forth on the Purchase Price Certificate attached hereto as Exhibit B, plus (B) the amount of \$700,000 (the "Estimated Payroll Tax Obligation") in respect of past-due payroll tax obligations relating to periods prior to the Closing Date, as reflected on the Estimated Closing Date Balance Sheet, without regard to transactions occurring after the Effective Time and outside the ordinary course of business.
- (ii) "Closing Date Net Working Capital" means, subject to final determination based on the Final Closing Date Balance
 Sheet, the amount of all current assets (net of reserves) of the Company, less the amount of all current liabilities (excluding those current liabilities included in
 Closing Date Debt) of the Company, as of the Effective Time, as reflected on the Estimated Closing Date Balance Sheet.
- (iii) "Estimated Closing Date Balance Sheet" means the estimated consolidated balance sheet of the Company as at the close of business on the Closing Date, prepared in accordance with generally accepted accounting principles ("GAAP"), and in accordance with the same principles and methods followed in preparing the Financial Statements referred to in Section 3.05 hereof, and delivered by the Company to Parent no later than three business days prior to the expected Closing Date.
 - (iv) "Final Closing Date Balance Sheet" has the meaning set forth in Section 2.04(a).
 - (v) "Merger Consideration" means, collectively, the Series A Merger Consideration and the Series B Merger Consideration.
- (vi) "*Merger Consideration Adjustment*" means, subject to final determination based on the Final Closing Date Balance Sheet, the aggregate of (A) the amount by which the Closing Date Debt exceeds \$2,000,000, and (B) either (1) increased by the amount, if any, by which the

Closing Date Net Working Capital is less than \$1,351,000, or (2) decreased by the amount, if any, by which the Closing Date Net Working Capital is in excess of \$1,351,000.

- (vii) "Parent Stock Per Share Price" means \$3.595.
- (viii) "Series A Merger Consideration" means, subject to final determination based on the Final Closing Date Balance Sheet, that number of shares of common stock, par value \$0.001 per share, of the Parent ("Parent Common Stock") equal to (A) \$4,200,000, less the Merger Consideration Adjustment, divided by (B) the Parent Stock Per Share Price.
- (ix) "*Series B Merger Consideration*" means, subject to final determination based on the Final Closing Date Balance Sheet, (A) a cash amount equal to \$3,300,000 less the Merger Consideration Adjustment, and (B) that number of shares of Parent

3

Common Stock equal to the Merger Consideration Adjustment divided by the Parent Stock Per Share Price.

- (c) Each share of Company Preferred Stock converted into the right to receive Merger Consideration pursuant to this <u>ARTICLE I</u> shall no longer be outstanding and shall automatically be canceled and retired and shall cease to exist, and each certificate (each a "*Certificate*," and collectively, the "*Certificates*") previously representing any such shares of Company Preferred Stock shall thereafter represent the right to receive Merger Consideration.
- (d) If, between the date of this Agreement and the Effective Time, the outstanding shares of Parent Common Stock shall be changed into a different number of shares by reason of any reclassification, recapitalization or exchange of shares or if a stock split, combination, stock dividend, stock rights or dividend thereon shall be declared with a record date within said period, the Parent Stock Per Share Price shall be correspondingly adjusted, as applicable. No fractional shares of Parent Common Stock will be issued, provided, however, that the Company shall provide cash in an amount equal to the value of such fractional share of Parent Common Stock based upon the Parent Stock Per Share Price.
- 1.06 Conversion of Merger Sub Stock. At the Effective Time, by virtue of the Merger and without any action on the part of Parent, Merger Sub, or the Company, each share of common stock, par value \$0.001 per share, of Merger Sub issued and outstanding immediately prior to the Effective Time shall be converted and exchanged for one validly issued, fully paid and nonassessable share of common stock, par value \$0.001 per share, of the Surviving Corporation. The stock certificate evidencing shares of common stock of Merger Sub shall then evidence ownership of the outstanding share of common stock of the Surviving Corporation.

1.07 Cash Holdback Amount; Stock Holdback Amount; Escrowed Shares.

- (a) \$250,000 of the cash portion of the Series B Merger Consideration (the "*Cash Holdback Amount*") to be delivered to the holders of Company Series B Preferred Stock (the "*Company Series B Preferred Stockholders*") at Closing shall be held back by counsel for Parent, on behalf of Parent, until such time as a Final Closing Date Balance Sheet has been determined pursuant to <u>Section 2.04(a)</u> hereof.
- (b) Such number of shares of Parent Common Stock to be issued to the holders of Company Series A Preferred Stock (the "Company Series A Preferred Stockholders," and, together with the Company Series B Preferred Stockholders, the "Company Preferred Stockholders") as Series A Merger Consideration equal to the Cash Holdback Amount divided by the Parent Stock Price Per Share (the "Stock Holdback Amount") to be delivered to the holders of Company Series A Preferred Stock at Closing shall be held back by counsel for Parent, on behalf of Parent, until such time as a Final Closing Date Balance Sheet has been determined pursuant to Section 2.04(a) hereof.
- (c) Such number of shares of Parent Common Stock to be issued to Company Series A Preferred Stockholders equal to \$1,125,000 divided by the Parent Stock Per Share Price (the "*Escrowed Shares*") shall be held in escrow for a period of one (1) year from the Closing Date, subject to the provisions of <u>ARTICLE IX</u> hereof, pursuant to the terms and subject to the

4

conditions set forth in an escrow agreement to be entered into among the parties hereto and Continental Stock Transfer & Trust Company, as Escrow Agent, in the form attached hereto as <u>Exhibit C</u>, with such modifications as may be reasonably acceptable to the Company and Parent, as requested by the Escrow Agent (the "*Escrow Agreement*").

- **1.08 Certificate of Incorporation.** Unless otherwise agreed to by the parties prior to the Effective Time, at and after the Effective Time, the Certificate of Incorporation of Merger Sub shall be the Certificate of Incorporation of the Surviving Corporation, until thereafter amended as provided by law and such Certificate of Incorporation.
- **1.09 Bylaws**. Unless otherwise agreed to by the parties prior to the Effective Time, at and after the Effective Time, the Bylaws of Merger Sub shall be the Bylaws of the Surviving Corporation, until thereafter amended as provided by law, the Certificate of Incorporation of the Surviving Corporation and such Bylaws.
- **1.10 Officers and Directors**. Unless otherwise agreed to by the parties prior to the Effective Time, the officers and directors of Merger Sub immediately prior to the Effective Time shall be the officers and directors of the Surviving Corporation immediately after the Effective Time, until thereafter elected as provided by law and the Certificate of Incorporation and Bylaws of the Surviving Corporation.

ARTICLE II PAYMENT OF MERGER CONSIDERATION

- (a) As promptly as practicable after the Effective Time, Parent shall mail to each record holder of Certificates a letter of transmittal, in the form attached hereto as Exhibit D (the "Letter of Transmittal") and instructions for use in surrendering such Certificates and receiving the applicable Merger Consideration pursuant to Section 1.05. From and after the Effective Time, upon surrender of all the Certificates representing all issued and outstanding shares of Company Preferred Stock to Parent (or affidavits and bonds relating thereto in accordance with Section 2.01(c)), together with a properly completed Letter of Transmittal, Stockholder Representation Letter, in the form attached hereto as Exhibit E (the "Stockholder Representation Letter"), and, if applicable, a Stock Restriction Agreement (as defined in Section 2.02) and a Non-Compete Agreement (as defined in Section 6.13), Parent shall deliver to each Company Preferred Stockholder such Stockholder's pro rata portion of the Merger Consideration that is not subject to holdback or placed in escrow, shall promptly deliver to each Company Preferred Stockholder such Stockholder's pro rata portion of the Merger Consideration so released from holdback or escrow.
- (b) After the date of this Agreement, there shall be no transfers on the stock transfer books of the Company of the shares of Company Securities which were issued and outstanding immediately prior to the date hereof.

5

- (c) In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by Parent, the posting by such person of a bond in such amount as Parent may direct as indemnity against any claim that may be made against it with respect to such Certificate, Parent will deliver in exchange for such lost, stolen or destroyed Certificate, the Merger Consideration specified in Section 2.01(a).
- **2.02 Stock Restriction Agreements.** Fifty percent (50%) of the shares of Parent Common Stock (the "*Restricted Shares*") to be issued pursuant to this Agreement to each Company Preferred Stockholder listed on <u>Schedule 2.02</u> that accepts an offer of employment to continue as an employee of the Parent or the Surviving Corporation as of and following the Closing (the "*Continuing Employee Stockholders*") shall be subject to forfeiture for a period of three (3) years from the Closing Date pursuant to the terms and conditions of a Stock Restriction Agreement to be entered into by the Parent and such Stockholder, in the form attached hereto as <u>Exhibit F</u> (the "*Stock Restriction Agreement*"). Any forfeiture of Restricted Shares by a Continuing Employee Stockholder pursuant to such Stock Restriction Agreement to the Merger Consideration payable to such Stockholder only.

2.03 No Assumption of Company Stock Options; Termination of Company Warrants.

- (a) No options to purchase Company Common Stock, whether granted under the Stock Plan (as defined in Section 3.02(b)) or otherwise, shall be assumed by Parent. Prior to the date hereof and subject to the Effective Time, the Company shall have effected the termination of the Stock Plan and shall have satisfied all requirements under the Stock Plan or otherwise to effect such termination, including, but not limited to, the giving of any notice and obtaining any consents, amendments or waivers required by the Stock Plan or any agreement relating to the Company Options. Company acknowledges that, immediately after the Effective Time, Parent and the Surviving Corporation shall take any and all actions necessary to cause the cancellation and termination of all Company Options that have not been exercised prior to the Effective Time.
- (b) Prior to the Effective Time, each warrant to acquire shares of Company Securities granted and outstanding immediately prior to the Effective Time (each, a "Company Warrant"), whether or not contingent or earned and whether exercisable or unexercisable, shall be terminated pursuant to its terms or pursuant to an agreement with the holder of such Company Warrant.

2.04 Adjustment of Merger Consideration.

(a) As soon as practicable but in no event later than thirty (30) days following the Closing Date, Parent, at its expense, shall cause the preparation of a final balance sheet of the Company, as at the close of business on the Closing Date (the "*Final Closing Date Balance Sheet*"), prepared in accordance with GAAP and in accordance with the same principles and methods followed in preparing the Financial Statements referred to in <u>Section 3.05</u> hereof. Parent shall share with the Stockholder Representative (as defined <u>Section 10.01</u>) such detailed calculations and supporting documents as the Stockholder Representative shall reasonably

6

request in connection with its review thereof. The Stockholder Representative may submit to Parent, not later than fifteen (15) days from the receipt of the Final Closing Date Balance Sheet from Parent, a list of any components of the Final Closing Date Balance Sheet with which the Stockholder Representative disagrees, if any (a "Dispute Notice"). If the Stockholder Representative does not issue a Dispute Notice prior to such date, the Final Closing Date Balance Sheet, as supplied to the Stockholder Representative, shall be deemed to have been accepted and agreed to by the Stockholder Representative on behalf of the Stockholders, and shall be final and binding on the parties to this Agreement and all Stockholders. In the event of a Dispute Notice by the Stockholder Representative, Parent and the Stockholder Representative shall thereafter for a period of up to twenty (20) days negotiate in good faith to resolve any items of dispute. Any items of dispute regarding the Final Closing Date Balance Sheet which are not so resolved shall be submitted to Deloitte (the "Arbitrating") Accountant") or another nationally recognized firm of public accountants mutually acceptable to Parent and the Stockholder Representative, who shall serve as an arbitrator hereunder, the expenses of which shall be shared one-half by Parent and one-half by the Stockholder Representative, on behalf of the Stockholders, unless otherwise determined by the Arbitrating Accountant. If Parent and the Stockholder Representative are unable to agree on an Arbitrating Accountant pursuant to the foregoing, each of (x) Parent and (y) the Stockholder Representative shall, within ten (10) days after delivery of the Dispute Notice select a disinterested arbitrator with relevant experience, and the two disinterested arbitrators so selected shall select, within ten (10) days of the selection of such arbitrating Accountant. In connection with the resolution of any dispute, the Arbitrating Accountant shall have access to all documents, records, work papers, facilities and personnel necessary to perform its function as arbitrator. The Arbitrating Accountant so selected shall render a written decision as promptly as practicable, but in no event later than thirty (30) days after submission of the matter to the Arbitrating Accountant. The decision of the Arbitrating Accountant shall be final and binding upon the parties and the Stockholders, and judgment may be entered on such decision in a court of competent jurisdiction. To the extent not otherwise provided herein, the commercial arbitration rules of the American Arbitration Association as in effect at the time of any arbitration shall govern such arbitration in all respects. The determination of the Arbitrating Accountant with respect to any and all disputes with respect to the Final Closing Date Balance Sheet shall be conclusive and binding upon all parties.

(b) If the Merger Consideration Adjustment, as finally determined based upon the Final Closing Date Balance Sheet (determined pursuant to the procedures set forth in Section 2.04(a)), exceeds the Merger Consideration Adjustment determined based upon the Estimated Closing Date Balance Sheet (such excess, if any, referred to herein as the "Additional Merger Consideration Adjustment"), then Parent shall, within ten (10) days after the Final Closing Date Balance Sheet is determined:
(i) deduct the amount of the Additional Merger Consideration Adjustment (the "Cash Reduction Amount") from the Cash Holdback Amount and pay the remaining balance, if any, of the Cash Holdback Amount to the Company Series B Preferred Stockholders, pro rata based on the number of shares of Company Series B Preferred Stock held by each Stockholder immediately prior to the Closing;
(ii) issue to the Company Series B Preferred Stockholders the number of shares of Parent Common Stock (the "Cash Replacement Shares") equal to the Cash
7
Reduction Amount divided by the Parent Stock Price Per Share, pro rata based on the number of shares of Company Series B Preferred Stock held by each Company Series B Preferred Stockholder immediately prior to the Closing;
(iii) deduct the number of shares of Parent Common Stock from the Stock Holdback Amount equal to the number of Cash Replacement Shares and deliver the remaining balance, if any, of the Stock Holdback Amount to the Company Series A Preferred Stockholders, pro rata based on the number of shares of Company Series A Preferred Stock held by each Company Series A Preferred Stockholder immediately prior to the Closing; and
(iv) to the extent the Additional Merger Consideration Adjustment is greater than the Cash Holdback Amount, claim such

- additional amount pursuant to **ARTICLE IX** of this Agreement.
- If the Merger Consideration Adjustment, as finally determined based upon the Final Closing Date Balance Sheet (determined pursuant to the procedures set forth in Section 2.04(a)), is less than the Merger Consideration Adjustment determined based upon the Estimated Closing Date Balance Sheet (such reduction, if any, referred to herein as the "Reduced Merger Consideration Adjustment"), then Parent shall, within ten (10) days after the Final Closing Date Balance Sheet is determined:
- pay the Cash Holdback Amount to the Company Series B Preferred Stockholders, pro rata based on the number of shares of Company Series B Preferred Stock held by each Stockholder immediately prior to the Closing:
- deliver the Stock Holdback Amount to the Company Series A Preferred Stockholders, pro rata based on the number of shares of Company Series A Preferred Stock held by each Stockholder immediately prior to the Closing; and
- pay the Reduced Merger Consideration Adjustment to the Company Series A Preferred Stockholders, pro rata based on (iii) the number of shares of Company Series A Preferred Stock held by each Stockholder immediately prior to the Closing.
- In the event the Surviving Corporation's liability for all past-due payroll tax obligations of the Company relating to periods prior to the Closing Date, including all costs associated with settling such payroll tax obligations, including reasonable attorneys' fees and disbursements and reasonable accountants' fees and disbursements, is less than the Estimated Payroll Tax Obligation, then promptly upon the settlement of such liability, Parent shall pay to the Company Series A Preferred Stockholders, pro rata based on the number of shares of Company Series A Preferred Stock held by each Stockholder immediately prior to the Closing, the amount by which the Estimated Payroll Tax Obligation exceeded the Surviving Corporation's liability for all such past-due payroll tax obligations.

2.05 **Dissenting Shares.**

Notwithstanding any provision of this Agreement to the contrary, shares of Company Securities that are outstanding immediately prior to the Effective Time and which are held by Stockholders (as defined in Section 3.02(a)) who have exercised and perfected

8

appraisal rights for such shares of Company Securities in accordance with the DGCL (collectively, the "Dissenting Shares") shall not be converted into or represent the right to receive Merger Consideration. Each such Stockholder shall be entitled to receive payment of the appraised value of such shares of Company Securities held by it in accordance with the DGCL, unless and until such Stockholder fails to perfect or effectively withdraws or otherwise loses its appraisal rights under the DGCL. All Dissenting Shares held by Stockholders who shall fail to perfect or who effectively shall waive, withdraw or otherwise lose their right to appraisal of such shares of Company Securities under the DGCL shall thereupon be deemed to have been converted into and to have become exchangeable for, as of the Effective Time, the right to receive the applicable portion of Merger Consideration, without any interest thereon, upon the surrender, in the manner provided in Section 2.01, of the corresponding Certificate.

The Company shall give Parent (i) prompt notice of any demands for appraisal received by the Company, withdrawals of such demands, and any other related instruments served pursuant to the DGCL and received by the Company and (ii) the opportunity to direct all negotiations and proceedings with respect to demands for appraisal under the DGCL. The Company shall not, except with the prior written consent of Parent, make any payment with respect to any demands for appraisal or offer to settle or settle any such demands.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to Parent and Merger Sub that the statements contained below are true and correct, except as set forth in the disclosure schedule (the "Company Disclosure Schedule") delivered by the Company to Parent and Merger Sub, on the date hereof and as of the Effective Date. The disclosures in any section or subsection of the Company Disclosure Schedule shall qualify other sections and subsections in this Article III where it should be reasonably apparent that such disclosure relates to other such sections and subsections. For purposes of this Agreement, "Company Material Adverse Effect" shall mean a material adverse effect on the Company, its business or financial condition, taken as a whole. When used herein, the term "to the knowledge of the Company" shall mean the actual knowledge of the executive officers of the Company.

3.01 Corporate Organization and Qualification.

(a) The Company is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. Each of the Company's Subsidiaries is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation. Each of the Company and its Subsidiaries has the corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted, and is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary. The Certificate of Incorporation and Bylaws of the Company and its Subsidiaries, copies of which have previously been delivered to Parent, are true, accurate and complete copies of such documents as in effect as of the date of this Agreement.

9

- (b) Except as set forth on Schedule 3.01(b), the Company has no direct or indirect Subsidiaries. Except as set forth on Schedule 3.01(b), the Company does not own, control or hold with the power to vote, directly or indirectly of record, beneficially or otherwise, any capital stock or any equity or ownership interest in any corporation, partnership, association, joint venture or other entity, except for less than five percent (5%) of any equity security registered under the Exchange Act. As used in this Agreement, the word "Subsidiary" means, with respect to any party, any corporation, partnership, limited liability company, or other organization, whether incorporated or unincorporated, which is or should be consolidated with such party for financial reporting purposes.
- (c) The minute books of each of the Company and its Subsidiaries contain records of all actions taken at meetings of, and corporate actions or written consents taken by its stockholders and board of directors (including committees thereof).

3.02 Capitalization.

- (a) The authorized capital stock of the Company consists of 54,000,000 shares of Company Common Stock, 1,000,000 shares of Preferred Stock, \$0.001 par value, of which 10,000 shares are designated as Company Series A Preferred Stock and 3,000 shares are designated as Company Series B Preferred Stock. As of the date of this Agreement, there are 28,097,727 shares of Company Common Stock, 8,148 shares of Company Series A Preferred Stock and 3,000 shares of Company Series B Preferred Stock issued and outstanding, all of which are owned by the holders and in the amounts as set forth in Schedule 3.02 annexed hereto (the "Stockholders"). All of the issued and outstanding shares of Company Securities have been duly authorized and validly issued and are fully paid, non-assessable and free of preemptive rights. The authorized and issued and outstanding capital stock of each Subsidiary of the Company is set forth on Schedule 3.02. All of the issued and outstanding shares of capital stock of each Subsidiary of the Company are owned by the Company, have been duly authorized and validly issued and are fully paid, non-assessable and free of preemptive rights. Except as set forth in Schedule 3.02 hereto and Section 3.02(b) hereof, the Company does not have and is not bound by any outstanding subscriptions, options, warrants, calls, commitments or agreements of any character calling for the purchase or issuance of any shares of Company Securities or any other equity security of the Company of its Subsidiaries or any securities representing the right to purchase or otherwise receive any shares of Company Securities or any other equity security of the Company other than as provided for in this Agreement. There are no bonds, debentures, notes, shares of preferred stock (other than the Company Preferred Stock) or other indebtedness of the Company having the right to vote (or convertible into, or exchangeable for securities
- (b) No shares of Company Common Stock have been issued pursuant to the exercise of options ("Company Options") granted under the Meritage Technologies, Inc. Nonqualified Stock Option Plan (the "Stock Plan") and (ii) 2,212,317 shares of Company Common Stock are subject to Company Options that have not yet been exercised. The Company Options have been granted to the holders in the amounts listed on <u>Schedule 3.02(b)</u>. The cancellation and termination of the Stock Plan in accordance with <u>Section 2.03</u> will not give rise or constitute a breach of the Stock Plan or any agreement entered into pursuant to such plan.

10

- (c) Except as disclosed on Schedule 3.02(c) hereto, there are no agreements or understandings with respect to the voting of any shares of Company Securities or of any Subsidiary of the Company or which restrict the transfer of such shares, to which the Company is a party and there are no such agreements or understandings to which the Company or any of its Subsidiaries is a party with respect to the voting of any such shares or which restrict the transfer of such shares, other than applicable federal and state securities laws.
 - (d) All dividends on Company Securities which have been declared prior to the date of this Agreement have been paid in full.

3.03 Authority; No Violation.

(a) The Company has full corporate power and authority to execute and deliver this Agreement and, subject to obtaining the necessary approvals of the Company Stockholders, to perform its obligations hereunder and to consummate the Merger and the other transactions contemplated hereby. The Board of Directors of the Company, at a meeting duly called and held, has unanimously approved and adopted the Merger, this Agreement and the other transactions contemplated hereby in accordance with the provisions of the DGCL and the Company's charter documents and directed that this Agreement and the Merger be submitted to the Company Stockholders for their approval and adoption. The execution and delivery of this Agreement and the consummation by the Company of the transactions contemplated by this Agreement have been duly and validly authorized by all requisite corporate action on the part of the Company and, except for the approval and adoption of this Agreement and the Merger by the Company Stockholders as required by the Company's Certificate of Incorporation and Bylaws and the filing of the Certificate of Merger, no other corporate proceedings on the part of the Company are necessary to approve this Agreement and to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by the Company and constitutes a valid and binding obligation of the Company, enforceable against each of the Company in accordance with its terms, subject to the effect of any applicable bankruptcy, reorganization, insolvency (including, without limitation, all laws relating to fraudulent transfers), moratorium or

similar laws affecting creditors' rights and remedies generally and subject, as to enforceability, to the effect of general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

(b) Except as set forth on Schedule 3.03(b)(i) hereto, neither the execution and delivery of this Agreement by the Company, nor the consummation by the Company of the transactions contemplated hereby, nor compliance by the Company with any of the terms or provisions, hereof, will (i) violate, conflict with or result in a breach of any provision of the Certificate of Incorporation or Bylaws of the Company, (ii) to the knowledge of the Company, violate any statute, code, ordinance, rule, regulation, judgment, order, writ, decree, license or injunction applicable to the Company or any of its Subsidiaries or any of their respective properties or assets, or (iii) violate, conflict with, result in a breach of any provisions of or the loss of any benefit under, constitute a default (or any event, which, with notice or lapse of time, or both would constitute a default) under, result in the termination of or a right of termination or cancellation under, accelerate the performance required by, or result in the creation of any lien, pledge, security interest, charge or other encumbrance upon any of the properties or assets of the

11

Company or any of its Subsidiaries under any Company Contract or other agreement entered into with the customers listed on <u>Schedule 3.03(b)(ii)</u> hereto (the "*Material Customers*").

3.04 Consents and Approvals. Except for (a) the filing of the Certificate of Merger with the Delaware Secretary pursuant to the DGCL to effect the Merger, (b) such filings under securities laws as may be necessary in connection with the offer and sale of the Parent Common Stock, (c) any facts or circumstances relating solely to Parent or Merger Sub, and (d) such filings, authorizations, consents or approvals as may be set forth on <u>Schedule 3.04</u> hereto, no consents or approvals of, or filings or registrations with, any court, administrative agency, regulatory agency or commission or other governmental authority or instrumentality (each a "*Governmental Entity*") or under any Company Contract or other agreement with any Material Customer are necessary in connection with the execution and delivery by the Company of this Agreement and the consummation by the Company of the Merger and the other transactions contemplated hereby.

3.05 Financial Statements.

- (a) Attached to Schedule 3.05(a) are copies of the audited consolidated balance sheets of the Company and its Subsidiaries as of December 31, 2003 and 2002, and the related consolidated statements of operations, stockholders' equity and cash flows for the years then ended (the "Audited Financial Statements"), and the unaudited consolidated balance sheet of the Company and its Subsidiaries as of April 30, 2004 and the related statement of income for the four month period ended April 30, 2004 (the "Interim Financial Statements"). The Audited Financial Statements and the Interim Financial Statements shall be collectively referred to herein as the "Financial Statements." The Financial Statements fairly present in all material respects the consolidated financial position of the Company and its Subsidiaries as of the dates thereof and the consolidated income of the Company and its Subsidiaries for the periods then ended and have been prepared by the Company as of their respective dates in accordance with GAAP during the periods involved (except as may be indicated therein or in the notes thereto, and provided that the Interim Financial Statements are subject to the absence of notes and to year-end adjustments).
- (b) Except (i) as set forth in the Company Disclosure Schedule, (ii) for liabilities incurred in the ordinary course of business consistent with past practice, or (iii) for liabilities contemplated herein or in connection herewith, the Company does not have any liabilities or obligations of any nature whatsoever (whether absolute, accrued, contingent or otherwise) which are not reflected on the Interim Financial Statements.
- (c) Schedule 3.05(c) contains the Company's accounts receivable report as of April 30, 2004, which report is true and accurate in all material respects and has been prepared in accordance with the Company's normal practice. The accounts receivable reflected in the April 30, 2004 report contained in Schedule 3.05(c) and all the accounts receivable arising after such date and prior to the Effective Time are valid and genuine and, subject to the allowance for bad debt set forth in the Interim Financial Statements, collectible, and arose from bona fide transactions in the ordinary course of the Company's business. Except as set forth in the Company Disclosure Schedule, no account receivable has been assigned or pledged to any other person and no defense or set off to any such account receivable has, to the knowledge of the

12

Company, been asserted by the account obligor. The allowance for bad debt for the Company's accounts receivable set forth on the Interim Financial Statements is in accordance with GAAP and in accordance with the historical accounting practices of the Company.

- (d) Except as set forth on <u>Schedule 3.05(d)</u>, since April 30, 2004, neither the Company nor its Subsidiaries has declared or paid any dividends, or made any other distribution on or in respect of, or directly or indirectly purchased, retired, redeemed or otherwise acquired any shares of the capital stock of the Company or issued or sold any such shares of capital stock.
- **3.06 Absence of Certain Changes or Events.** Since April 30, 2004, there has not been any Company Material Adverse Effect and, to the knowledge of the Company, no fact or condition specific to the Company exists which would reasonably be expected to cause such a Company Material Adverse Effect.
- 3.07 Legal Proceedings. The Company is not a party to any, and there are no pending or, to the knowledge of the Company, threatened, legal, administrative, arbitrable or other proceedings, claims, actions or governmental or regulatory investigations of any nature against or affecting the Company or any of its Subsidiaries or any property or asset of the Company or any of its Subsidiaries, before any court, arbitrator, administrative agency or Governmental Entity, domestic or foreign. To the knowledge of the Company, neither the Company nor any of its Subsidiaries nor any property or asset of the Company is subject to any order, writ, judgment, injunction, decree, determination or award which restricts its ability to conduct business in any area in which it presently does business.

3.08 Taxes and Tax Returns.

(a) For purposes of this Agreement, the terms "*Tax*" and "*Taxes*" shall mean any and all taxes, charges, fees, levies or other assessments, including, without limitation, all net income, gross income, gross receipts, premium, sales, use, ad valorem, value added, transfer, franchise,

profits, license, withholding, payroll, employment, excise, estimated, severance, stamp, occupation, property or other taxes, fees, assessments or charges of any kind whatsoever, together with any interest and any penalties (including penalties for failure to file in accordance with applicable information reporting requirements), and additions to tax by any authority, whether federal, state, local, domestic or foreign. The term "*Tax Return*" shall mean any report, return, form, declaration or other document or information required to be supplied to any authority in connection with Taxes.

(b) The Company and its Subsidiaries (collectively, the "*Taxpayer*") has filed all Tax Returns that were required to be filed. All such Tax Returns were when filed, and continue to be, correct and complete in all material respects, and all Taxes owed by the Taxpayer (whether or not shown on any Tax Return) for any taxable period or portion thereof ending on or before the Closing Date have been timely paid or adequate provision has been or will be made therefore prior to Closing. The Taxpayer currently is not the beneficiary of any extension of time within which to file any Tax Return. No claim has ever been made by an authority in a jurisdiction where the Taxpayer does not file Tax Returns that it is or may be subject to taxation by that jurisdiction. There are no liens with respect to Taxes on any of the assets or property of the Taxpayer.

13

- (c) The Taxpayer has withheld or collected and paid all Taxes required to have been withheld or collected and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, Stockholder, any other third party, or otherwise.
- (d) The Taxpayer has delivered to Parent correct and complete copies of all federal income Tax Returns, examination reports, and statements of deficiencies assessed against or agreed to by or on behalf of the Taxpayer since July 19, 2000 and prior to the date hereof.
- (e) No claim, assessment, deficiency, audit, investigation, or administrative proceeding with respect to Taxes or any Tax Return of the Taxpayer is pending or, to the knowledge of the Taxpayer, has been threatened.
- (f) The Taxpayer has not waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency.
- (g) The Taxpayer has not agreed to make, nor is it required to make, any adjustments under Section 481(a) of the Code by reason of a change in accounting method or otherwise.
- (h) The Taxpayer is not a party to any contract, arrangement or plan that has resulted or could result, separately or in the aggregate, in the payment of any "excess parachute payments" within the meaning of Section 280G of the Code.
- (i) The Taxpayer has not been a United States real property holding corporation within the meaning of Code Section 897(c)(2) during the applicable period specified in Code Section 897(c)(1)(A)(ii).
- (j) The Taxpayer is not a party to any agreement, whether written or unwritten, providing for the payment of Tax liabilities, payment for Tax losses, entitlements to refunds or similar Tax matters.
 - (k) No ruling with respect to Taxes relating to the Taxpayer has been requested by or on behalf of the Taxpayer.
- (l) The Taxpayer (i) has never been a member of an affiliated group (within the meaning of Section 1504 of the Code, or any similar group as defined for state, local or foreign tax purposes) filing a consolidated federal (or combined or unitary state, local or foreign) income Tax Return and (ii) does not have any liability for the taxes of any Person (other than the Taxpayer's) under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign Law), as a transferee or successor, by contract, or otherwise.
- (m) The unpaid Taxes of the Taxpayer (i) did not, as of the date of the Financial Statements, exceed the reserves for Tax liability (rather than any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the Financial Statements and (ii) do not exceed the reserve as adjusted for the passage of time through the Effective Date in accordance with the past custom and practice of the Taxpayer in filing its Tax Returns.

14

(n) Except as set forth on <u>Schedule 3.08(n)</u> hereto, the Taxpayer will not be required to include in any period ending after the Closing Date any income that accrued in a prior period but was not recognized in any prior period as a result of the installment method of accounting, the completed contract method of accounting, the long-term contract method of accounting or the cash method of accounting.

3.09 Employee Benefit Plans.

- (a) Schedule 3.09 hereto sets forth a true and complete list of all Plans maintained or contributed to by the Company or any of its Subsidiaries during the five (5) years preceding this Agreement. The term "Plans" for purposes of this ARTICLE III means all "employee benefit plans" (as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974 as amended ("ERISA") and each other plan, arrangement, agreement or policy relating to employment, bonus, deferred compensation, retirement, profit sharing, 401(k), stock option, stock purchase, employee stock ownership, stock appreciation rights, savings, severance, termination, collective bargaining, group insurance, fringe benefit, and other employee benefit, incentive and welfare plans, written or oral, and all trusts related thereto, that are maintained or contributed to, or that were maintained or contributed to at any time during the five (5) years preceding the date of this Agreement, by the Company or any of its Subsidiaries, or by any trade or business, whether or not incorporated (an "ERISA Affiliate"), all of which together with the Company would be deemed a "single employer" within the meaning of Section 4001 of ERISA.
- (b) The Company has heretofore made available to Parent true and complete copies of each of the Plans and all related documents, including but not limited to (i) all required Forms 5500 and all related schedules for such Plans (if applicable) for each of the last two (2) years, (ii) the actuarial report for such Plan (if applicable) for each of the last two (2) years, and (iii) the most recent determination letter from the IRS (if applicable) for such plan.

(c) Except as set forth in Schedule 3.09, (i) each of the Plans has been operated and administered in all material respects in accordance with applicable laws, including but not limited to ERISA and the Code, (ii) each of the Plans intended to be "qualified" within meaning of Section 401(a) of the Code has been maintained so as to qualify from the effective date of such Plan to the Effective Time, (iii) with respect to each Plan which is subject to Title IV of ERISA, the present value of "benefit liabilities" (within the meaning of Section 4001(a)(16) of ERISA) under such Plan, based upon the actuarial assumptions currently used by the Plan for IRS funding purposes did not, as of its latest valuation date, exceed the then current value of the assets of such Plan allocable to such accrued benefits, and there has been no "accumulated funding deficiency" (whether or not waived), (iv) no Plan provides benefits, including without limitation death, medical or other benefits (whether or not insured), with respect to current or former employees of the Company, any of its Subsidiaries or any ERISA Affiliate beyond their retirement or other termination of service, other than (A) coverage mandated by applicable law, (B) life insurance death benefits payable in the event of the death of a covered employee, (C) disability benefits payable to disabled former employees, (D) death benefits or retirement benefits under any "employee pension plan," as that term is defined in Section 3(2) of ERISA, (E) deferred compensation benefits accrued as liabilities on the books of the Company, any of its Subsidiaries or any ERISA Affiliate or (F) benefits the full cost of which

15

is borne by the current or former employee (or his beneficiary), (v) with respect to each Plan subject to Title IV of ERISA no liability under Title IV of ERISA has been incurred by the Company, any of its Subsidiaries or any ERISA Affiliate that has not been satisfied in full, no condition exists that presents a material risk to the Company, any of its Subsidiaries or any ERISA Affiliate of incurring a material liability to or on account of such Plan, and there has been no "reportable event" (within the meaning of Section 1013 of ERISA and the regulations thereunder), (vi) none of the Company, any of its Subsidiaries or any ERISA Affiliate has ever maintained or contributed to a "multiemployer pension plan," as such term is defined in Section 3(37) of ERISA, (vii) all contributions or other amounts payable by the Company, any of its Subsidiaries as of the Effective Time with respect to each Plan in respect of current or prior plan years have been paid or accrued in accordance with GAAP and Section 412 of the Code, (viii) none of the Company, any of its Subsidiaries or any ERISA Affiliate has engaged in a transaction in connection with which the Company, any of its Subsidiaries or any ERISA Affiliate has any material liability for either a civil penalty assessed pursuant to Section 409 or 502(i) of ERISA or a tax imposed pursuant to Section 4975 or 4976 of the Code, (ix) consummation of the transactions contemplated hereby will not cause any amounts payable under any of the Plans to fail to be deductible for federal income tax purposes under Sections 280G or 162(m) of the Code, and (x) there are no pending, or, to the knowledge of the Company, threatened or anticipated claims (other than routine claims for benefits) by, on behalf of or against any of the Plans or any trusts related thereto, and to the knowledge of the Company, no Plan is under governmental investigation or audit or under consideration with the Internal Revenue Service under the Employee Plans Compliance Resolution System.

- (d) With respect to any Plan that is a welfare plan (within the meaning of Section 3(l) of ERISA) (i) no such Plan is funded through a "welfare benefit fund," as such term is defined in Section 419(a) of the Code, and (ii) each such Plan complies in all material respects with the applicable requirements of Section 4980B(f) of the Code, Part 6 of Subtitle B of Title I of ERISA and any applicable state continuation coverage requirements ("COBRA").
- (e) Except as prohibited by law (including Section 411(d)(6) of the Code), each Plan may be amended, terminated, modified or otherwise revised by the Company, any of its Subsidiaries or its ERISA Affiliates as of the Effective Time to eliminate, without material effect, any and all future benefit accruals under any Plan (except claims incurred under any welfare plan).
- (f) Except as contemplated herein, neither the Company nor any of its Subsidiaries has entered into, adopted or amended in any respect any collective bargaining agreement or adopted or amended any bonus, profit sharing, compensation, stock option or other similar plan, agreement, trust, fund or arrangement for the benefit of employees (whether or not legally binding).
- (g) The consummation of the transaction contemplated herein will not (i) entitle any current or former employee of the Company or any Subsidiary to severance benefits or any other compensation (including, without limitation, golden parachute or bonus), or (ii) accelerate the time of payment or vesting of any such benefits, or increase the amount of compensation due any such employee or service provider, except in each case as expressly provided in this Agreement.

16

- (h) The Company and its Subsidiaries have no material unfunded liabilities for benefits or claims accrued pursuant to any Plan, which Plan is not intended to be qualified under Section 401(a) of the Code.
- 3.10 Compliance with Applicable Law; Certain Agreements; Licensing. Except as set forth in Schedule 3.10(i) hereto, to the knowledge of the Company, each of the Company and its Subsidiaries holds all licenses, franchises, permits and authorizations necessary for the lawful conduct of its business, except where the failure to hold such license, franchise, permit or authorization would not result in a Company Material Adverse Effect, and has complied with and is not in conflict with, or in default or violation of any (a) statute, code, ordinance, law, rule, regulation, order, writ, judgment, injunction or decree, published policies and guidelines of any Governmental Entity, applicable to the Company or any Subsidiary or by which any property or asset of the Company or any Subsidiary is bound or affected or (b) any note, bond, mortgage, indenture, deed of trust, lease, license, permit, franchise, material agreement or other material instrument or obligation to which the Company or any Subsidiary is a party or by which the Company or any Subsidiary or any property or asset of the Company is bound or affected; and, to the knowledge of the Company, neither the Company nor any Subsidiary has received notice of, any violations of any the above. Schedule 3.10(ii) hereto contains a list of all material federal and state licenses, franchises, permits and authorizations necessary for the lawful conduct of the Company's or any of is Subsidiaries' respective businesses.

3.11 Certain Contracts.

(a) Except as contemplated hereby or as set forth in Schedule 3.11(a)(i) hereto, neither the Company nor any of its Subsidiaries is a party to or bound by any contract, arrangement, commitment or understanding (whether written or oral): (i) with respect to the employment of any director, officer or employee, or with respect to the employment of any consultant which cannot be terminated without payment, (ii) which, upon the consummation of the transactions contemplated by this Agreement, will result in any payment (whether of severance pay or otherwise) becoming due from the Company or any of its Subsidiaries to any officer or employee thereof, (iii) which is a material contract (as defined in Item 601(b)(10) of Regulation S-K promulgated by the Securities and Exchange Commission) ("SEC") to be performed after the date of this Agreement, (iv) which is a material consulting or other agreement (including agreements entered into in the ordinary course and data processing, software programming and licensing contracts) not terminable on ninety (90)

days or less notice, (v) which restricts the conduct of any line of business by the Company or any of its Subsidiaries, (other than commercially reasonable nonsolicitation provisions in customer contracts), (vi) with or to a labor union or guild (including any collective bargaining agreement), or (vii) with respect to any stock option plan, stock appreciation rights plan, restricted stock plan or stock purchase plan any of the benefits of which will be increased, or the vesting of the benefits of which will be accelerated, by the occurrence of any of the transactions contemplated by this Agreement or the value of any of the benefits of which will be calculated on the basis of any of the transactions contemplated by this Agreement. Other than as set forth in the Company Disclosure Schedule or in this Agreement, no benefits under any of such plans will be increased, or the vesting of the benefits of which, will be accelerated by the occurrence of any of the transactions contemplated by this Agreement. The Company has previously delivered to Parent true and complete copies of all such employment, consulting and deferred compensation agreements which are in writing and

17

to which the Company or any of its Subsidiaries is a party. Each contract, arrangement, commitment or understanding of the type described in this Section is referred to herein as a "*Company Contract*".

- (b) Except as set forth in Schedule 3.11(b) hereto, (i) each Company Contract is legal, valid and binding upon the Company or any of its Subsidiaries, as the case may be, and in full force and effect, subject to the effect of any applicable bankruptcy, reorganization, insolvency (including, without limitation, all laws relating to fraudulent transfers), moratorium or similar laws affecting creditors' rights and remedies generally and subject, as to enforceability, to the effect of general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law), (ii) the Company has performed all material obligations required to be performed by it to date under each such Company Contract, (iii) to the knowledge of the Company, no party is in breach or default and (iv) no event or condition exists which constitutes or, after notice or lapse of time or both, would constitute, a material default on the part of the Company or any of its Subsidiaries, as the case may be, under any such Company Contract.
- **3.12 Agreements with Regulatory Agencies**. To the knowledge of the Company, neither the Company nor any of its Subsidiaries is subject to any cease-and-desist or other order issued in writing by, or is a party to any written agreement, consent agreement or memorandum of understanding, commitment letter, suspension order, or similar undertaking (each, a "*Regulatory Agreement*") with any regulatory agency or any other Governmental Entity, nor has the Company been notified in writing or, to the knowledge of the Company, otherwise by any regulatory agency or any other Governmental Entity that it is considering issuing or requesting any Regulatory Agreement.

3.13 Environmental Matters.

- (a) The Company and its Subsidiaries are, and have been, in material compliance with all applicable environmental laws and with all rules, regulations, standards and requirements of the United States Environmental Protection Agency (the "*EPA*") and of state and local agencies with jurisdiction over pollution or protection of the environment.
- (b) There is no suit, claim, action or proceeding pending or, to the knowledge of the Company, threatened, before any Governmental Entity or other forum in which the Company has been or, with respect to threatened proceedings, may be named as a defendant, responsible party or potentially responsible party in any way relating to any environmental law, rule, regulation, standard or requirement.

3.14 Properties.

- (a) Neither the Company nor any of its Subsidiaries owns or has any ownership interest in any real property.
- (b) Except as set forth in <u>Schedule 3.14</u> hereto, to the knowledge of the Company, all buildings and structures leased and used by the Company or any of its Subsidiaries conform in all material respects with all applicable ordinances, codes or regulations.

18

- (c) Schedule 3.14 contains a true, complete and correct list of all leases pursuant to which the Company or any of its Subsidiaries leases any real or personal property, either as lessee or as lessor which leases call for an annual payment in excess of \$10,000 (the "Company Leases"). Assuming due authorization of the other party or parties thereto, each of the Company Leases is valid and binding on the Company or any of its Subsidiaries and, to the knowledge of the Company, valid and binding on and enforceable against all other respective parties to such leases, in accordance with their respective terms, subject to the effect of any applicable bankruptcy, reorganization, insolvency (including, without limitation, all laws relating to fraudulent transfers), moratorium or similar laws affecting creditors' rights and remedies generally and subject, as to enforceability, to the effect of general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law). There are not under such Company Leases any existing material breaches, defaults or events of default by the Company or any of its Subsidiaries nor has the Company or any of its Subsidiaries received notice of, or made a claim with respect to, any breach or default by any other party to such Company Leases. Each of the Company and its Subsidiaries enjoys quiet and peaceful possession of all such leased properties occupied by it as lessee.
- 3.15 Insurance. The Company has made available to Parent true and complete copies of all policies of insurance of the Company and its Subsidiaries currently in effect, a list of which is attached as Schedule 3.15. All of the policies relating to insurance maintained by the Company with the respect to its properties and the conduct of its business (or any comparable policies entered into as a replacement thereof) are in full force and effect and the Company has not received any notice of cancellation with respect thereto. All life insurance policies on the lives of any of the current and former officers of the Company which are maintained by the Company or any of its Subsidiaries or which are otherwise included as assets on the books of the Company are, or will at the Effective Time be, owned by the Company or any of its Subsidiaries, free and clear of any claims thereon by the officers or members of their families, except with respect to the death benefits thereunder, as to which the Company agrees that there will not be an amendment prior to the Effective Time without the consent of Parent. The Company does not have any liability for unpaid premium or premium adjustments not properly reflected in the Company's Interim Financial Statements. All claims under any policy or bond have been duly and timely filed.
- **3.16 Labor Matters.** Neither the Company nor any of its Subsidiaries is a party to any collective bargaining or other labor union or guild contract nor has the Company or any of its Subsidiaries been approached by any collective bargaining or other labor union or guild seeking to enter into a contract with the Company or any of its Subsidiaries. There is no pending or, to the knowledge of the Company, threatened, labor dispute, strike or work stoppage against the Company or any of its Subsidiaries. Neither the Company nor any of its Subsidiaries nor any of its representatives or employees has

committed any unfair labor practices in connection with the operation of the business of the Company or any of its Subsidiaries, and there is no pending or, to the knowledge of the Company, threatened charge or complaint against the Company or any of its Subsidiaries by the National Labor Relations Board or any comparable state agency. To the knowledge of the Company, neither the Company nor any of its Subsidiaries has hired any illegal aliens as employees. To the knowledge of the Company, neither the Company nor any of its Subsidiaries has discriminated on the basis of race, age, sex or otherwise in its employment conditions or practices with respect to its employees. There are no race, age, sex or other

19

discrimination complaints pending or, to the knowledge of the Company, threatened against the Company or any of its Subsidiaries by any employee, former or current, before any domestic (federal, state or local) or foreign board, department, commission or agency nor, to the knowledge of the Company, does any basis therefor exist.

- 3.17 Intellectual Property. To the knowledge of the Company, the Company and its Subsidiaries own or possess a license or other right to use without payment of any amount all patents, copyrights, trade secrets, trade names, service marks, trademarks, domain names, know-how, software and other intellectual property necessary to the operation of its business as presently conducted (collectively, the "Intellectual Property Rights"). Schedule 3.17 sets forth a list of all patents, pending patent applications, registered copyrights, registered trademarks and service marks and applications for the registration of trademarks and service marks which are owned by the Company as well as all intellectual property license agreements (excluding off-the-shelf software programs licensed by the Company pursuant to "shrink wrap" licenses and licenses embodied in agreements entered into between the Company and its customers in connection with the services performed by the Company thereunder). Neither the Company nor any of its Subsidiaries has received any notice of conflict with the Intellectual Property Rights from any third party. The Company and its Subsidiaries are not in default under any contract, agreement, arrangement or commitment relating to any of the Intellectual Property Rights, except where the existence of a default would not have a Company Material Adverse Effect. The Intellectual Property Rights are valid and enforceable and, to the knowledge of the Company, do not infringe upon the rights of any third parties.
- **3.18 Broker's Fees.** Neither the Company nor any of its officers or directors, has employed any broker or finder or incurred any liability for any broker's fees, commissions or finder's fees in connection with any of the transactions contemplated by this Agreement.
- **3.19 Bank Accounts.** The Company has provided or made available to Parent complete and current summaries of information regarding all accounts, lock boxes and safe deposits maintained by the Company and its Subsidiaries at banks, trust companies, securities firms or other brokers or other financial institutions.
- **3.20 Disclosure.** No representation or warranty of the Company contained in this Agreement or any schedule to this Agreement contains any untrue statement of a material fact or omits to state a material fact necessary in order to make the statements herein or therein, in light of the circumstances in which they are made, not misleading.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

The Parent and Merger Sub hereby, jointly and severally, represent and warrant to the Company that the statements contained below are true and correct, except as set forth in the disclosure schedule (the "*Parent Disclosure Schedule*") delivered by the Parent and Merger Sub to Company, on the date hereof and as of the Effective Date. The disclosures in any section or subsection of the Parent Disclosure Schedule shall qualify other sections and subsections in this Article IV where it should be reasonably apparent such disclosure relates to other such sections and subsections. For purposes of this Agreement, "*Parent Material Adverse Effect*" shall mean

20

a material, adverse effect on the Parent, its business or financial condition, taken as a whole. When used herein, the term "to the knowledge of the Parent" shall mean the actual knowledge of the executive officers of the Parent.

4.01 Corporate Organization and Qualification. Each of Parent and Merger Sub is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. Each of Parent and Merger Sub has the corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted, and is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary. The Certificate of Incorporation and Bylaws of each of Parent and Merger Sub, copies of which have previously been delivered to the Company, are true and complete copies of such documents as in effect as of the date of this Agreement. Schedule 4.01 sets forth the name and address of all of the Parent's Subsidiaries. Neither Parent nor any of its Subsidiaries owns, controls or holds with the power to vote, directly or indirectly of record, beneficially or otherwise, any capital stock or any equity or ownership interest in any corporation, partnership, association, joint venture or other entity, except for less than five percent (5%) of any equity security registered under the Exchange Act.

4.02 Capitalization.

(a) The authorized capital stock of Parent consists of 40,000,000 shares of Parent Common Stock and 8,000,000 shares of preferred stock, par value \$0.001 per share ("Parent Preferred Stock"). As of the date hereof, [17,961,673] shares of Parent Common Stock and no shares of Parent Preferred Stock are issued and outstanding. All of the issued and outstanding shares of Parent Common Stock have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights. Except as set forth in Schedule 4.02 hereto, as of the date immediately prior to the date hereof, Parent does not have and is not bound by any outstanding subscriptions, options, warrants, calls, commitments or agreements of any character calling for the purchase or issuance of any shares of Parent Common Stock or any other equity security of Parent or any of its Subsidiaries or expressenting the right to purchase or otherwise receive any shares of Parent Common Stock or any other equity security of Parent or any of its Subsidiaries other than as provided for in this Agreement. There are no bonds, debentures, notes, shares of preferred stock or other indebtedness of Parent having the right to vote (or convertible into, or exchangeable for securities having the right to vote) on any matters on which the stockholders of Parent may vote. There are no agreements or understandings with respect to the voting of any shares of Parent Common Stock or any Subsidiary of Parent or which restrict the transfer of such shares to which Parent or any of its Subsidiaries is a party, other than applicable federal and state securities laws.

4.03 Authority; No Violations.

- Each of Parent and Merger Sub have full corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement by Parent and Merger Sub and the consummation by Parent and Merger Sub of the transactions contemplated hereby have been duly and validly authorized by all requisite corporate action on the part of each of Parent and Merger Sub. Except for the filing of the Certificate of Merger, no other corporate proceedings on the part of Parent or Merger Sub are necessary to approve this Agreement and to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by Parent and Merger Sub and, assuming the due authorization, execution and delivery by the Company, constitutes a valid and binding obligation of Parent and Merger Sub, enforceable against Parent and Merger Sub in accordance with its terms, subject to the effect of any applicable bankruptcy, reorganization, insolvency (including, without limitation, all laws relating to fraudulent transfers), moratorium or similar laws affecting creditors' rights and remedies generally and subject, as to enforceability, to the effect of general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- (b) Neither the execution and delivery of this Agreement by each of Parent and Merger Sub, nor the consummation by either Parent or Merger Sub, as the case may be, of the transactions contemplated hereby, nor compliance by either Parent or Merger Sub with any of the terms or provisions hereof, will (i) violate, conflict with or result in a breach of any provision of the Certificate of Incorporation or Bylaws of Parent, or Merger Sub, as the case may be, or (ii) violate any statute, code, ordinance, rule, regulations, judgment, order, writ, decree or injunction applicable to Parent or Merger Sub or any of their respective properties or assets, or (iii) violate, conflict with, result in a breach of any provisions of or the loss of any benefit under, constitute a default (or any event which, with notice or lapse of time, or both, would constitute a default) under, result in the termination of or a right of termination or cancellation under, accelerate the performance required by, or result in the creation of any lien, pledge, security interest, charge or other encumbrance upon any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, lease, material agreement or other instrument or obligation to which Parent or Merger Sub is a party, or by which they or any of their respective properties or assets may be bound or affected.
- **4.04 Consents and Approvals.** Except for (a) the filing of the Certificate of Merger with the Delaware Secretary pursuant to the DGCL to effect the Merger, (b) such filings under securities laws as may be necessary in connection with the offer and sale of the Parent Common Stock, (c) such filings as may be necessary as a result of any facts or circumstances related solely to the Company, and (d) the consents and approvals listed on <u>Schedule 4.04</u> hereto, no consents or approvals of or filings or registrations with any Governmental Entity or with any third party are necessary in connection with the execution and delivery by Parent and Merger Sub of this Agreement and the consummation by Parent and Merger Sub of the Merger and the other transactions contemplated hereby.
- **4.05 Broker's Fees.** Neither Parent nor Merger Sub, nor any of their respective officers or directors, has employed any broker or finder or incurred any liability for any broker's

22

fee, commission or finder's fee in connection with any of the transactions contemplated by this Agreement, except as set forth in Schedule 4.05 hereto.

- **4.06 No Prior Activities.** Merger Sub has not incurred, directly or indirectly, any liabilities or obligations, except those incurred in connection with its incorporation or with the negotiation of this Agreement and consummation of the transactions contemplated hereby. Merger Sub has not engaged, directly or indirectly, in any business or activity of any type or kind, or entered into any agreement or arrangement with any person or entity, or become subject to or bound by any obligation or undertaking, that is not contemplated by or in connection with this Agreement and the transactions contemplated hereby.
- **4.07 Labor Matters.** Neither Parent nor any of its Subsidiaries is a party to any collective bargaining or other labor union or guild contract nor, to Parent's knowledge, has Parent or any of its Subsidiaries been approached by any collective bargaining or other labor union or guild seeking to enter into a contract with Parent or any of its Subsidiaries. There is no pending, or to Parent's knowledge, threatened labor dispute, strike or work stoppage against Parent or any of its Subsidiaries which may interfere with the business activities of Parent or any of its Subsidiaries. None of Parent or any of its Subsidiaries or their respective representatives or employees has committed any unfair labor practices in connection with the operation of the business of Parent or any of its Subsidiaries, and there is no pending or, to Parent's knowledge, threatened charge or complaint against Parent or any of its Subsidiaries by the National Labor Relations Board or any comparable state agency.

4.08 Reports; Financial Statements.

Parent has filed all forms, reports, registration statements and documents required to be filed by it with the SEC since January 1, 1999 (such forms, reports, registration statements and documents, together with any amendments thereto, the "Parent SEC Filings"). As of their respective dates, the Parent SEC Filings (i) comply as to form in all material respects with the requirements of the Securities Act of 1933, as amended (the "Securities Act") and the Securities and Exchange Act of 1934, as amended (the "Exchange Act"), as the case may be, and (ii) do not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances under which they were made, not misleading. As of the date hereof and as of the Effective Date, and the representations and warranties of Parent contained in this Agreement or any schedule to this Agreement, taken as a whole, do not contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements herein, in light of the circumstances in which they are made, not misleading. No event or circumstance has occurred or information exists with respect to Parent or its business, properties, operations or financial conditions, which, under applicable law, rule or regulation, requires public disclosure or announcement by Parent but which has not been so publicly announced or disclosed (assuming for this purpose the preparation of an effective registration statement filed by Parent under the Securities Act into which Parent's reports filed under the Exchange Act are incorporated by reference).

accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes thereto), (ii) complied as of their respective dates in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, and (iii) fairly present the consolidated financial position of Parent and its Subsidiaries as of the dates thereof and the income, cash flows, and changes in stockholder's equity for the periods involved.

- **4.09 No Undisclosed Material Liabilities**. There are no liabilities, commitments or obligations of the Parent or any of its Subsidiaries of any kind whatsoever whether accrued, contingent, absolute, determined, determinable or otherwise, and there is no existing condition, situation or set of circumstances that would reasonably be likely to result in such a liability commitment or obligation, other than:
 - (a) liabilities, commitments or obligations disclosed or provided for in the Parent SEC Filings;
 - (b) liabilities, commitments or obligations incurred in the ordinary course of business consistent with past practice;
 - (c) liabilities, commitments or obligations under this Agreement; and
- (d) liabilities, commitments or obligations that individually or in the aggregate have not had and are not reasonably likely to have a Parent Material Adverse Effect.
- **4.10 Legal Proceedings.** The Parent is not a party to any, and there are no pending or to the knowledge of Parent threatened, legal, administrative, arbitrable or other proceedings, claims, actions or governmental or regulatory investigations of any nature against or affecting the Parent or any of its Subsidiaries or any property or asset of the Parent or any of its Subsidiaries, before any court, arbitrator, administrative agency or Governmental Entity, domestic or foreign which would have Parent Material Adverse Effect.
- **4.11 Continued Listing of Stock.** Parent's Common Stock is listed for trading on the Nasdaq SmallCap Market ("*Nasdaq*")and the Boston Stock Exchange and meets all maintenance criteria for continued listing thereon, respectively. The Parent has not received any notices or correspondences in writing from either the Nasdaq or the Boston Stock Exchange regarding any current investigation or inquiry or providing any notice of a current proceeding with respect to the delisting of the Parent's securities.
- **4.12 Valid Issuance of Parent Common Stock**. The shares of Parent Common Stock to be issued pursuant to this Agreement will, when issued, be duly authorized, validly issued, fully paid and non-assessable. Assuming each Company Preferred Stockholder executes a Stockholder Representation Letter in the form attached hereto as Exhibit E, delivers such Stockholder Representation Letter to Parent prior to the Effective Time, and does not amend or revoke such Stockholder Representation Letter prior to the Effective Time, and assuming the accuracy of the representations and warranties of the Company Preferred Stockholders in the Stockholder Representation Letters, the shares of Parent Common Stock to be issued pursuant to this Agreement will be validly issued pursuant to the "private placement" exemption from

24

registration provided by Section 4(2) of the Securities Act and/or Regulation D promulgated under the Securities Act.

ARTICLE V COVENANTS RELATING TO CONDUCT OF BUSINESS

- **5.01** Covenants of the Company. During the period from the date of this Agreement and continuing until the Effective Time, except as expressly contemplated or permitted by this Agreement or with the prior written consent of Parent, the Company shall (and shall cause its Subsidiaries to) carry on its business in the ordinary course consistent with past practice. The Company shall (and shall cause its Subsidiaries to) use all reasonable efforts to preserve its business organization, keep available the present services of its employees, continue to make regularly scheduled payments on all of its existing debt and preserve for itself and Parent the goodwill of the customers of the Company and its Subsidiaries and others with whom business relationships exist, including, but not limited to all Company Contracts. Without limiting the generality of the foregoing, and except as otherwise contemplated by this Agreement or consented to in writing by Parent, the Company shall not (and shall cause its Subsidiaries not to):
- (a) except as specifically provided in the Company Disclosure Schedule with regard to <u>Section 3.05(d)</u> hereof, declare or pay any dividends on, or make other distributions in respect of, any of its capital stock;
- (b) (i) split, combine or reclassify any shares of its capital stock; or issue or authorize or propose the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock except upon the exercise or fulfillment of rights or options issued or existing pursuant to employee benefit plans, programs or arrangements, all to the extent outstanding and in existence on the date of this Agreement, or (ii) repurchase, redeem or otherwise acquire, any shares of the capital stock of the Company, or any securities convertible into or exercisable for any shares of the capital stock of the Company;
- (c) issue, deliver or sell, or authorize or propose the issuance, delivery or sale of, any shares of its capital stock or any securities convertible into or exercisable for, or any rights, warrants or options to acquire, any such shares, or enter into any agreement with respect to any of the foregoing;
 - (d) amend its Certificate of Incorporation or Bylaws;
 - (e) make any capital expenditures in excess of \$10,000;

- (f) enter into any new line of business;

 (g) acquire or agree to acquire, by merging or consolidating with, or by purchasing a equity interest in or a portion of the assets of or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof, or otherwise acquire any assets other than in the ordinary course of business;

 (h) change its methods of accounting in effect at December 31, 2003, except as required by changes in GAAP or regulatory accounting principles as concurred to by the Company's independent auditors;

 (i) except as contemplated hereby or as set forth on Schedule 5.01(i) hereto, enter into, adopt, amend, renew or terminate any Plan or any agreement, arrangement, plan or policy between the Company or any of its Subsidiaries and one or more of its current or former directors, officers or
- (i) except as contemplated hereby or as set forth on Schedule 5.01(i) hereto, enter into, adopt, amend, renew or terminate any Plan or any agreement, arrangement, plan or policy between the Company or any of its Subsidiaries and one or more of its current or former directors, officers or employees, or increase in any manner compensation or fringe benefits of any director, officer or employee or pay any benefit not required by any plan or agreement as in effect as of the date hereof (including, without limitation, the granting of stock options, stock appreciation rights, restricted stock, restricted stock units or performance units or shares), excluding, however, any increase in compensation or benefits for any employees in the ordinary course of business and specifically approved in writing by Parent; or enter into, modify or renew any employment, severance or other agreement with any director, officer or employee of the Company or any of its Subsidiaries or establish, adopt, enter into, or amend any collective bargaining, bonus, profit sharing, thrift, compensation, stock option, restricted stock, pension, retirement, deferred compensation, employment, termination, severance or other plan, agreement, trust, fund, policy or arrangement providing for any benefit to any director, officer or employee (whether or not legally binding);
- (j) incur any indebtedness for borrowed money, assume, guarantee, endorse or otherwise as an accommodation become responsible for the obligations of any other individual, corporation or other entity except in the ordinary course of business consistent with past practices of the Company;
- (k) sell, lease, encumber, assign or otherwise dispose of, or agree to sell, lease, encumber, assign or otherwise dispose of, any of its assets, properties or other rights or agreements other than in the ordinary course of business;
 - (l) make any Tax election, or settle or compromise any federal, state, local or foreign Tax liability or file or amend any Tax Return;
- (m) pay, discharge or satisfy any claim, liability or obligation other than payments in the ordinary course of business and consistent with past practices of the Company, liabilities incurred in connection with the Merger and the transactions expressly contemplated hereby, or liabilities reflected or reserved against in the Financial Statements, or subsequently incurred in the ordinary course of business and consistent with past practices of the Company;
- (n) except as set forth on <u>Schedule 5.01(n)</u> hereto, enter into or renew, amend or terminate, or give notice of a proposed renewal, amendment or termination, or make any commitment with respect to, regardless of whether consistent with past practices, any lease, contract, agreement or commitment having a term of one (1) year or more from the time of execution or outside the ordinary course of business consistent with past practices;
 - (o) waive any material right, whether in equity or at law; or
 - (p) agree to do any of the foregoing.

26

- **5.02 Covenants of Parent and Merger Sub.** During the period from the date of this Agreement and continuing until the Effective Time, Parent and its Subsidiaries shall not:
 - (a) declare or pay any dividends on, or make other distributions in respect of, any of its capital stock;
- (b) (i) split, combine or reclassify any shares of its capital stock; or issue or authorize or propose the issuance of any other securities in respect of, in lieu of, or in substitution for shares of its capital stock except upon the exercise or fulfillment of rights or options issued or existing pursuant to employee benefit plans, programs or arrangements, all to the extent outstanding and in existence on the date of this Agreement, or (ii) repurchase, redeem or otherwise acquire, any shares of the capital stock of Parent or its Subsidiaries, or any securities convertible into or exercisable for any shares of the capital stock of Parent or its Subsidiaries, except, in all circumstances, if an appropriate adjustment is made to the Parent Stock Per Share Price to reflect the effect on the Parent Stock Per Share Price of any such event;
- (c) other than as set forth on <u>Schedule 4.02</u> hereof, issue, deliver or sell, or authorize or propose the issuance, delivery or sale of, any shares of its capital stock or any securities convertible into or exercisable for, or any rights, warrants or options to acquire, any such shares, or enter into any agreement with respect to any of the foregoing except as currently permitted pursuant to Parent's employee benefit plans;
 - (d) amend its Certificate of Incorporation or Bylaws;
- (e) change its methods of accounting in effect at December 31, 2003, except as required by changes in GAAP or regulatory accounting principles as concurred to by Parent and its Subsidiaries' independent auditors;
- (f) except under the Parent's credit facility or any equipment or capital lease facilities, incur any indebtedness for borrowed money, assume, guarantee, endorse or otherwise as an accommodation become responsible for the obligations of any other individual, corporation or other entity except in the ordinary course of business consistent with past practice of Parent and its Subsidiaries and except for indebtedness in an amount less than \$100,000; or
 - (g) agree to do any of the foregoing.

5.03 All Necessary Action. Each of the parties hereto shall use all reasonable efforts to take, or cause to be taken, all action and to do, or cause to be done, all things necessary, proper or advisable to consummate the transaction contemplated hereby as soon as practicable. No party shall intentionally perform any act which, if performed, or omit to perform any act which, if omitted to be performed, would prevent or excuse the performance of this Agreement by any party hereto or which would result in any representation or warranty herein contained of such party being untrue in any material respect as if originally made on and as of the Closing Date.

5.04 Notification. Each party shall promptly give the other party written notice of the existence or occurrence of any condition which would make any representation or warranty

27

herein contained of either party untrue or which might reasonably be expected to prevent the consummation of the transactions contemplated hereby.

ARTICLE VI ADDITIONAL AGREEMENTS

Regulatory Matters. The parties hereto shall cooperate with each other and use all reasonable efforts promptly to prepare and file all necessary documentation, to effect all applications, notices, petitions and filings, and to obtain as promptly as practicable all permits, consents, approvals and authorizations of all third parties and Governmental Entities which are necessary or advisable to consummate the transactions contemplated by this Agreement (including without limitation the Merger). The Company and Parent shall have the right to review in advance, and to the extent practicable each will consult with the other on, in each case subject to applicable laws relating to the exchange of information, all the information relating to the Company, Parent or Merger Sub, as the case may be, which appear in any filing made with or written materials submitted to, any third party or any Governmental Entity in connection with the transactions contemplated by this Agreement. In exercising the foregoing right, each of the parties hereto shall act reasonably and as promptly as practicable. The parties hereto agree that they will consult with each other with respect to the obtaining of all permits, consents, approvals and authorizations of all third parties and Governmental Entities necessary or advisable to consummate the transactions contemplated by this Agreement and each party will keep the other apprised of the status of matters relating to completion of the transactions contemplated herein. Parent (or Merger Sub as the case may be) and the Company shall promptly furnish each other with copies of written communications received by Parent, Merger Sub or the Company, as the case may be, from or delivered by any of the foregoing to, any Governmental Entity in respect of the transactions contemplated hereby.

6.02 Securities Matters.

- (a) During the two (2) year period following the Closing Date, Parent shall (i) use its best efforts to make current public information available in accordance with Rule 144(c) under the Securities Act and to maintain the continued listing of its shares of Common Stock for trading on the Nasdaq and the Boston Stock Exchange and (ii) furnish to any Stockholder upon written request, (x) a written statement as to its compliance with the requirements of Rule 144(c) and the reporting requirements of the Securities Act and the Exchange Act and (y) a copy of the most recent annual or quarterly report of Parent.
- (b) Parent shall file, within sixty (60) days after the Closing Date, a registration statement ("*Registration Statement*") on Form S-3, or other appropriate registration form, with the SEC under the Securities Act with respect to the offer and sale by the Stockholders pursuant to Rule 415 promulgated under the Securities Act of (i) 25% of the shares of Parent Common Stock to be issued to Continuing Employee Stockholders pursuant to this Agreement (which shares shall not include any Escrowed Shares) and (ii) 100% of the shares of Parent Common Stock to be issued to other Company Stockholders pursuant to this Agreement (collectively, the "*Registrable Securities*") and will use reasonable, prompt and diligent efforts to cause such Registration Statement to become effective as soon as practicable thereafter.

28

Parent shall use its reasonable, prompt and diligent efforts to cause the Registrable Securities to be listed on Nasdaq and the Boston Stock Exchange.

- (c) Notwithstanding any other provision of this Section 6.02, if the Parent shall furnish to the Stockholder Representative a certificate signed by the president or chief executive officer of the Parent stating that in the good faith judgment of the board of directors of the Parent it would be seriously detrimental to the Parent and its stockholders (including the Stockholders) for such Registration Statement to be filed or such registration to be effected at such time, the Parent shall have the right to defer the filing of the registration statement for so long as reasonably necessary, but no later than 120 days after the Closing Date.
- (d) Parent shall prepare and file with the SEC such amendments and supplements to such registration statement and the prospectus used in connection with the Registration Statement as may be necessary to comply with the provisions of the Securities Act with respect to the disposition of all Registrable Securities and to keep such registration statement effective until the earlier of such time as all Stockholders have completed the distribution described in the Registration Statement or the date on which all the Registrable Securities may be immediately sold without registration, and without restriction as to the number of securities to be sold, pursuant to Rule 144 under the Securities Act.
- (e) Parent shall, if required under applicable law at the time, use its best efforts to register and qualify the Registrable Securities under such other securities or blue sky laws of such jurisdictions as shall be reasonably requested by the Holders; provided that Parent shall not be required in connection with such registration and qualification or as a condition to such registration and qualification (i) to qualify to do business or to file a general consent to service of process in any such states or jurisdictions or (ii) to subject itself to taxation in any jurisdiction.
- (f) Parent shall notify each Stockholder covered by the Registration Statement at any time when a prospectus relating to the Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the prospectus included in the Registration Statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated in such prospectus or necessary to make the statements in such prospectus not misleading in the light of the circumstances then existing.
- (g) Parent shall furnish to each Stockholder covered by the Registration Statement such number of conformed copies of the Registration Statement and of each amendment and supplement thereto (in each case including all exhibits and documents incorporated by reference), such

number of copies of the prospectus contained in such registration statement (including each preliminary prospectus and any prospectus supplement) and any other prospectus filed under Rule 424 promulgated under the Securities Act relating to such Stockholder's shares included in the Registration Statement.

(h) Parent shall notify each Stockholder covered by the Registration Statement (i) when such Registration Statement or any prospectus used in connection therewith, or any amendment or supplement thereto, has been filed and, with respect to such Registration

29

Statement or any post-effective amendment thereto, when the same has become effective, (ii) of any written request by the SEC for amendments or supplements to such Registration Statement or prospectus or for supplemental information, (iii) of the notification to Parent by the SEC of its initiation of any proceeding with respect to the issuance by the SEC of, or of the issuance by the SEC of, any stop order suspending the effectiveness of such Registration Statement; and (iv) of the receipt by Parent of any notification with respect to the suspension of the qualification of the Registrable Securities for sale under the applicable securities or "blue-sky" laws of any jurisdiction.

- (i) In the event of the issuance of any stop order suspending the effectiveness of the Registration Statement, or of any order suspending or preventing the use of any related prospectus or suspending the qualification of the Registrable Securities for sale in any jurisdiction, Parent shall use all reasonable efforts promptly to obtain the withdrawal of such order.
- (j) All expenses incurred in effecting the registration under Registration Statement shall be borne by Parent. All underwriting discounts, selling commissions, and stock transfer taxes relating to the Registrable Securities shall be borne by the Stockholders pro rata on the basis of the number of shares of Registrable Securities registered on their behalf.
- (k) Parent may require any Stockholder to, and each such Stockholder, shall, furnish Parent with such information regarding such Stockholder and the distribution of the Registrable Securities as Parent may from time to time reasonably request in writing and to otherwise cooperate in connection with such registration. At any time during the effectiveness of the Registration Statement, if such Stockholder becomes aware of any change materially affecting the accuracy of the information contained in such Registration Statement or the prospectus (as then amended or supplemented) relating to such Stockholder, including but not limited to the sale or disposition of all Registrable Securities owned by each such Stockholder, he or it will promptly notify Parent of such change.
- (l) Upon receipt of any notice from Parent of the happening of any event as a result of which any prospectus included in the Registration Statement, as then in effect, includes an untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading, each Stockholder will forthwith discontinue such Stockholder's disposition of Registrable Securities pursuant to the Registration Statement until such Stockholder receives copies of a supplemented or amended prospectus from Parent and, if so directed by Parent, shall deliver to Parent (at Parent's expense) all copies, other than permanent file copies, then in such Stockholder's possession of the prospectus relating to such Registration Statement current at the time of receipt of such notice.
- (m) Parent shall, to the full extent permitted by law, indemnify and hold harmless each Stockholder included in the Registration Statement against any expenses, claims, losses, damages or liabilities to which such Stockholder may become subject under the Securities Act or otherwise, insofar as such expenses, claims, losses, damages or liabilities or actions in respect thereof arise out of or are based upon any untrue statement of any material fact contained in the Registration Statement, final prospectus, preliminary prospectus, or prospectus supplement

30

contained therein or filed with the SEC, or any amendment or supplement thereto, or any omission to state therein a material fact required to be stated therein or. necessary to make the statements therein (in the case of a prospectus, in the light of the circumstances under which they were made) not misleading; provided, that Parent shall not be liable in any such case to the extent that any such loss (or actions in respect thereof) arises out of or is based upon an untrue statement or omission made in any such Registration Statement, final prospectus, amendment or supplement in reliance upon and in conformity with information furnished in writing to Parent by such Stockholder and stated to be specifically for use therein.

- Each Stockholder shall, to the full extent permitted by law, indemnify and hold harmless Parent, its directors, officers, employees, agents and each other person, if any, who controls Parent within the meaning of the Securities Act, against any expenses, claims, losses, damages or liabilities to which Parent or any such director, officer, employee, agent or controlling person may become subject under the Securities Act or otherwise, insofar as such expenses, claims, losses, damages or liabilities arise out of or are based upon any untrue statement of any material fact contained in the Registration Statement, final prospectus or prospectus supplement contained therein or filed with the SEC, or any amendment or supplement thereto, or any omission to state therein a material fact required to be stated therein or necessary to make the statements therein (in the case of a prospectus, in the light of the circumstances under which they were made) not misleading, if such untrue statement or omission was made in reliance upon and in conformity with written information furnished to Parent by such Stockholder specifically stating that it is for use in the preparation of such Registration Statement, final prospectus, amendment or supplement; provided, however, that the obligation to provide indemnification pursuant to this Section 6.02(n), shall be several among such indemnifying parties on the basis of the number of shares of Parent Common Stock of each such indemnifying party included in the Registration Statement, and shall not exceed the value as of the date hereof of the shares of Parent Common Stock received by such Stockholder pursuant to this Agreement. The foregoing indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of Parent or any such director, officer, employee, agent or controlling person and shall survive the transfer of such securities by such Stockholder. Each Stockholder shall also indemnify each other Stockholder who participates in the offering or sale u
- (o) Promptly after receipt by any party of notice of the commencement of any action or proceeding involving a claim referred to in Section 6.02(m) or 6.02(n), such party shall, if a claim in respect thereof is to be made against another party pursuant to such paragraphs, give written notice to the latter of the commencement of such action, provided that any failure of any person to give notice as provided therein shall not relieve any other person of its obligations under Section 6.02(m) or 6.02(n), as the case may be, except to the extent that such other person is actually prejudiced by such failure. In case any such action is brought, the party obligated to indemnify pursuant to Section 6.02(m) or 6.02(n), as the case may be, shall be entitled to participate in and, unless, in the reasonable judgment of counsel to any indemnified party, a conflict of interest between such indemnified party and any indemnifying party

31

indemnified party, and after notice from the indemnifying party to such indemnified party of its election so to assume the defense thereof, the indemnifying party shall not be liable to such indemnified party for any legal or other expenses subsequently incurred by the latter in connection with the defense thereof other than reasonable costs of investigation; provided that the indemnified party may participate in such defense at the indemnified party's expense. Without the consent of the indemnified party, no indemnifying party shall consent to entry of any judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to each indemnified party of a release from all liability in respect to such claim or litigation. No indemnifying party shall be subject to any liability for any settlement made without its consent, which consent shall not be unreasonably withheld.

- (p) If the indemnity and reimbursement obligation provided for in Section 6.02(m) or 6.02(n) is unavailable or insufficient to hold harmless a party entitled to indemnification hereunder in respect of any expenses, claims, losses, damages or liabilities (or actions with respect thereto) referred to therein, the party obligated to indemnify hereunder shall contribute to the amount paid or payable by the indemnified party as a result of such expenses, claims, losses, damages or liabilities (or actions) in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and the indemnified party on the other hand in connection with statements or omissions which resulted in such expenses, claims, losses, damages or liabilities as well as any other relevant equitable considerations. Relative fault shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the indemnifying party or the indemnified party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such untrue statement or omission. The parties hereto agree that it would not be just and equitable if contributions pursuant to this paragraph were-to be determined by pro rata allocation or by any other method of allocation which does not take account of the equitable considerations referred to in the first sentence of this paragraph. No person guilty of fraudulent misrepresentation within the meaning of the Securities Act shall be entitled to contribution from any person not guilty of such fraudulent misrepresentation.
 - (q) Notwithstanding the foregoing, each Stockholder in the Stockholder Representation Letters, shall covenant and agree as follows:
- (i) with respect to each Continuing Employee Stockholder, except for Registrable Securities, that during the thirty-six (36)-month period commencing with the Closing, he will not (A) offer, pledge, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant to purchase, lend or otherwise transfer or dispose of, directly or indirectly, any shares of Parent Common Stock or any securities convertible into or exercisable or exchangeable for Parent Common Stock, or (B) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of Parent Common Stock; provided, however, that Stockholder may engage in any action described in (A) or (B) above only to the extent that no more than an aggregate of twelve and one-half percent (12.5%) of the shares of Parent Common Stock received by Stockholder pursuant to this Agreement has been the subject of actions described in (A) or (B) above during the period between the first anniversary of the Closing and the second anniversary of the Closing, and no more than an aggregate of twenty-five

32

percent (25%) of the shares of Parent Common Stock received by Stockholder pursuant to this Agreement has been the subject of actions described in (A) or (B) above during the period between the first anniversary of the Closing and the third anniversary of the Closing; or

(ii) with respect to all other Stockholders, that (A) during the one-year period following the Closing Date, he will not offer, pledge, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant to purchase, lend or otherwise transfer or dispose of, directly or indirectly, any shares of Parent Common Stock held in escrow pursuant to the Escrow Agreement, and (B) during the ten (10)-month period commencing with the effective date of the Registration Statement filed in accordance with Section 6.02, he will not (x) offer, pledge, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant to purchase, lend or otherwise transfer or dispose of, directly or indirectly, any shares of Parent Common Stock or any securities convertible into or exercisable or exchangeable for Parent Common Stock, or (y) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of Parent Common Stock; provided, however, that Stockholder may engage in any action described in (x) or (y) above with respect to Registrable Securities other than those Registrable Securities held in escrow under the Escrow Agreement and only to the extent that no more than ten percent (10%) of the Registrable Securities registered pursuant to such Registration Statement, less the Registrable Securities held in escrow pursuant to the Escrow Agreement, and attributable to the Stockholder is the subject of actions described in (x) or (y) above in each monthly period following the effective date of the Registration Statement.

The restrictions on transferability of Parent Common Stock set forth in this <u>Section 6.02</u> shall survive any change of control of the Parent. Each share of Parent Common Stock issued pursuant to this Agreement shall bear a restrictive legend stating the above restrictions.

6.03 Stockholder Approval; Exemption from Registration.

- (a) Immediately upon signing this Agreement, and in accordance with applicable law, the Company's Certificate of Incorporation and Bylaws, the Company shall solicit written consents from its Stockholders to obtain their approval and adoption of this Agreement, the Merger and the other transactions contemplated hereby. The Company shall ensure that all written consents are solicited and obtained from the Company stockholders in compliance with applicable law, the Company's Certificate of Incorporation and Bylaws. The Company agrees to use all reasonable efforts to take all action necessary or advisable to secure the necessary votes required by applicable law and the Company's Certificate of Incorporation and Bylaws to effect the Merger.
- (b) Each of the parties hereto acknowledge that the shares of Parent Common Stock to be issued to the Company Preferred Stockholders pursuant to this Agreement are intended to be issued pursuant to the "private placement" exemption from registration under Section 4(2) of the Securities Act and/or Regulation D promulgated under the Securities Act and agree to fully cooperate with Parent in its efforts to ensure that such shares of Parent Commons Stock may be issued pursuant to such private placement exemption.

- (c) Immediately upon signing this Agreement, the Company will distribute an information statement (together with any amendments thereof or supplements thereto, the "Information Statement") in a form previously approved by Parent, describing the Parent, the Company and the transactions contemplated hereby and otherwise in accordance with applicable law, and the Company will send to each Company Stockholder the Information Statement for the purpose of considering, approving and adopting this Agreement, and the other transactions contemplated hereby. None of the information included in the Information Statement and any other document prepared to comply with Federal or state securities laws, shall, at the time it is first mailed to the Company Stockholders or at the Effective Time, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. No amendment or supplement of the Information Statement will be made by the Company without the approval of Parent, which shall not be unreasonably withheld.
- **6.04** Access to Information; Confidentiality. (a) Subject to Section 6.04(b), the Company shall afford to Parent, and shall cause its independent accountants to afford to Parent and Parent's accountants, counsel and other representatives, reasonable access during normal business hours during the period prior to the Closing to all of the Company's assets, properties, books, Company Contracts and records, and the Company shall permit Parent and its representatives to make abstracts from and copies of such books and records. During such period, the Company shall use its reasonable best efforts to furnish promptly to Parent all other information concerning the business, properties and personnel of the Company as Parent may reasonably request.
- (b) No party (or its representatives, agents, counsel, accountants or investment bankers) hereto shall disclose to any third party, other than either party's representatives, agents, counsel, accountants or investment bankers any confidential or proprietary information about the business, assets or operations of the other parties to this Agreement or the transactions contemplated hereby, except as contemplated hereby and as may be required by applicable law. The parties hereto agree that the remedy at law for any breach of the requirements of this subsection will be inadequate and that any breach would cause such immediate and permanent damage as would be impossible to ascertain, and, therefore, the parties hereto agree and consent that in the event of any breach of this subsection, in addition to any and all other legal and equitable remedies available for such breach, including a recovery of damages, the non-breaching parties shall be entitled to obtain preliminary or permanent injunctive relief without the necessity of proving actual damage by reason of such breach and, to the extent permissible under applicable law, a temporary restraining order may be granted immediately on commencement of such action.
- **6.05 No Solicitation of Transactions.** Neither the Company, nor any of its Subsidiaries, nor their respective directors, officers, employees, stockholders, representatives, agents and advisors nor other persons controlled by the Company shall solicit or hold discussions or negotiations with, or assist or provide any information to, any person, entity or group (other than Parent, Merger Sub and their affiliates and representatives) concerning (i) any merger, consolidation, business combination, share exchange, or other similar transaction involving the Company; (ii) any sale, lease, exchange, mortgage, pledge, license transfer or other disposition of any shares of Company Common Stock or significant assets of the Company; or (iii) the

34

issuance of any new shares of capital stock of the Company or any options, warrants or other rights to acquire shares of capital stock of the Company. The Company will promptly communicate to Parent, Merger Sub and their affiliates and representatives the terms of any proposal, discussion, negotiation or inquiry relating to a merger or disposition of a significant portion of its capital stock or assets or similar transaction involving the Company and the identity of the party making such proposal or inquiry, which it may receive with respect to any such transaction.

- **6.06 Legal Conditions to Merger.** Each of Parent, Merger Sub and the Company shall use all reasonable efforts (a) to take, or cause to be taken, all actions necessary, proper or advisable to comply promptly with all legal requirements which may be imposed on such party with respect to the Merger and, subject to the conditions set forth in <u>ARTICLE VII</u> hereof, to consummate the transactions contemplated by this Agreement and (b) to obtain (and to cooperate with the other party to obtain) any consent, authorization, order or approval of or any exemption by, any Governmental Entity and any other third party which is required to be obtained by Parent, Merger Sub or the Company in connection with the Merger and the other transactions contemplated by this Agreement.
- **6.07 Disclosure Supplements.** Prior to the Effective Time, each party will supplement or amend the applicable Disclosure Schedule delivered in connection with the execution of this Agreement to reflect any matter which, if existing, occurring or known at the date of this Agreement, would have been required to be set forth or described in such Disclosure Schedule or which is necessary to correct any information in such Disclosure Schedule which has been rendered inaccurate thereby. No supplement or amendment to such Disclosure Schedule shall have any effect for the purposes of determining satisfaction of the conditions set forth in Section 7.02(b) hereof or the compliance by the Company with the covenants set forth in Section 5.01 hereof (unless Parent consents in writing to such satisfaction of conditions or compliance or elects to waive such matter by closing the transactions contemplated hereby) or for the purposes of determining satisfaction of the conditions set forth in Section 7.03(a) hereof or the compliance by Parent with the covenants set forth in Section 5.02 hereof (unless the Company consents to such satisfaction of conditions or elects to waive such matter by closing the transactions contemplated hereby).

6.08 Tax Matters.

Audits. Each party hereto agrees to notify the other party or parties promptly of any audits, investigations, or examinations of or having a potential impact on the Taxes of the Company or Stockholders, by any federal, state or local taxing authority with regard to any taxable periods or portions thereof ending up to and including the Closing Date, if the audit, investigation, or examination could affect the Taxes of the Company or any parties' indemnification obligations hereunder. In the event of such an audit, investigation, or examination, Parent or Merger Sub shall control such proceedings and each such party agrees to (i) allow the other party and its attorneys and accountants to participate in said audit, investigation or examination in respect to items which may affect the Company or Stockholders, at such other party's expense, and (ii) provide the other party and its agents, accountants, and attorneys, with reasonable access to such books and records as they reasonably deem necessary to determine the validity of the items under audit, investigation, or examination by said federal,

35

state or local taxing authority. The parties agree to either (x) obtain the written consent of the other party prior to agreeing to any audit adjustment or settlement which would adversely affect such other party or its affiliates or (y) waive their right to seek reimbursement from such other party or its affiliates with respect to such audit adjustment or settlement. Notwithstanding anything to the contrary herein, a party hereto other than the Parent or Merger Sub can control such proceedings concerning an audit, investigation, or examination if and only if such party expressly agrees to fully and totally indemnify and hold

harmless the Parent and the Merger Sub for any tax liability attributable to periods prior to the Closing Date (or any tax liability in excess of the Estimated Payroll Tax Obligation in respect of past due payroll tax obligations relating to periods prior to the Closing Date).

- (b) **Inspection of Tax Records.** Following the Closing Date, Parent, affiliates of Parent, the Company, and Surviving Corporation shall, upon reasonable request, afford to the Stockholders and the Stockholder Representative reasonable access during normal business hours to the books, records and other data of or relating to the Company and Surviving Corporation (and permit the Stockholders and their Stockholder Representative to make copies thereof at their own expense) with respect to taxable periods or portions thereof ending prior to the Closing Date, to the extent that such access may be reasonably required by the Stockholders for the Company or Surviving Corporation to prepare their individual federal, state and local tax returns with respect to all taxable periods ending on or including the Closing Date, and to defend an audit or other investigation or examination of any of their personal federal, state or local Tax Returns for all taxable periods or portions thereof ending on or including the Closing Date.
- (c) **Retention of Tax Records.** The Company, Surviving Corporation, Parent and Parent's affiliates agree to retain all books and records with respect to Tax matters pertinent to the Company until the expiration of three (3) months after the relevant statute of limitations of the respective taxable periods, and to abide by all record retention agreements entered into with any taxing authority.

(d) **Preparation of Tax Returns.**

- (i) The Company shall file at Company's expense, on or prior to the due date thereof (taking into account applicable extensions), all Tax Returns required to be filed by the Company on or before the Closing Date, with a copy thereof provided to Parent promptly after such filing, and shall timely pay any Taxes due with respect to such Tax Return.
- (ii) Parent, Surviving Corporation or their affiliates will file (or cause to be filed) all Tax Returns of the Company or Surviving Corporation due after the Closing Date. With respect to any such Tax Return for a taxable period that ends on or prior to the Closing Date, any Tax due with respect to such Tax Return shall constitute a Parent Indemnified Tax.
- (iii) From and after the date hereof, the Stockholders shall not, and shall not permit any of their respective affiliates to, amend any Tax Return previously filed which includes information relating to the Company, unless prior written notice thereof has been delivered to Parent and such amended Tax Return will not result in any tax liability to the Company.

36

- (e) **Tax Treatment.** The parties agree that any amounts paid or payable pursuant to <u>Section 6.08(d)</u> shall be treated by the parties on their Tax Returns as an adjustment to the Merger Consideration.
- **6.09 Employment Agreements.** Simultaneously with the execution of this Agreement, Parent and each of the Continuing Employee Stockholders listed on <u>Schedule 6.09</u> have entered into employment agreements in substantially the form attached as <u>Exhibit G</u> hereto, with such changes as may be mutually agreed to by Parent and such Stockholder.
- **6.10 Company Employees.** Schedule 6.10 contains a list of all employees of the Company (including the Continuing Stockholder Employees) to whom the Parent will offer employment with the Surviving Corporation or Parent as of and following the Closing (the "Continuing Employees"). It shall be a condition to the employment of such Continuing Employees with the Surviving Corporation or Parent that the execute a Confidentiality and Intellectual Property Assignment Agreement in the form attached hereto as Exhibit H, unless another form of such agreement is attached to an Employment Agreement entered into between Parent and the Continuing Employee. On or before the Closing Date and in accordance with applicable law, the Company shall terminate all of the employees of the Company other than Continuing Employees and other than those employees noted on Schedule 6.10 as "Transition Employees." The Company may, with the prior approval of Parent, provide for severance payments for such employees.

6.11 Plans and Stock Option Grants.

- (a) From and after the Effective Time, (i) all Transition Employees shall continue in their existing benefit plans until their employment with the Surviving Corporation is terminated, and (ii) all Continuing Employees shall continue in their existing benefit plans until such time as, in Parent's sole discretion, an orderly transition can be accomplished to employee benefit plans and programs maintained by Parent for its and its affiliate's employees in the United States, except that the Company's Stock Plan shall be terminated as of the business day immediately preceding the Closing Date. Parent shall take as promptly as reasonably practical after the Closing Date, and with respect to the health and welfare benefits programs no later than 90 days after the Closing Date, and to the extent permitted by Parent's benefit programs, such reasonable actions as are necessary to allow eligible Continuing Employees to participate in the health, welfare and other benefits programs of Parent or alternative benefits programs that, in the aggregate, are substantially equivalent to those applicable to employees of Parent in similar functions and positions on similar terms. Pending such action, Parent shall maintain the effectiveness of the Company's benefit plans, except as otherwise provided herein.
- (b) On or before the Closing, Parent shall have reserved an aggregate of 525,000 shares of Parent Common Stock under Parent's Stock Option Plan (the "Continuing Employee Option Pool") for the grant to Continuing Employees in such amounts as determined by Parent and the Company and as set forth on Schedule 6.11 hereto. After the Effective Time, Parent shall take all necessary corporate action to grant options consisting of the entire Continuing Employee Option Pool to the Continuing Employees, at the first meeting of Parent's Board of Directors following the Closing under the terms and conditions of Parent's Stock Option Plan and the form stock option agreement (the "Parent Stock Option Agreement") set forth in Exhibit

37

<u>I</u> hereto. The exercise price per share of Parent Common Stock purchasable under each option granted from the Continuing Employee Pool shall be equal to the fair market value per share of Parent Common Stock on the option grant date.

6.12 Non-Compete Agreements. As additional consideration for Parent, and as a material inducement for Parent and Merger Sub to enter into this Agreement and to consummate the Merger, each person listed on <u>Schedule 6.12</u> shall enter into a Non-Compete Agreement with Parent, in the form

attached hereto as Exhibit J (the "Non-Compete Agreement"), on or before the Closing Date, pursuant to which each such person shall agree to certain noncompetition, nonsolicitation and no-hire provisions as mutually agreed to between Parent and such Stockholder or other person.

6.13 Publicity. Parent, Merger Sub, the Company and the Stockholder Representative agree that, except as otherwise required by law, they (a) will make no public comment concerning or announcement regarding the Merger and (b) institute procedures to restrict knowledge of the proposed transaction to those who need to know. In addition, the parties acknowledge that Parent, as a publicly held company, is subject to certain disclosure requirements under federal securities laws. Accordingly, Parent reserves the right to disclose the Merger, including financial information regarding the Company and the status of negotiations, at any time it decides that such disclosure is appropriate under the securities laws or the rules of any stock exchange, provided, however, that Parent shall provide the Company and its counsel a reasonable time to review and comment upon such disclosure.

Except as otherwise required by law or the rules of the Nasdaq or the Boston Stock Exchange and notwithstanding anything in this Agreement to the contrary, so long as this Agreement is in effect, none of Parent, Merger Sub or the Company shall, or shall permit any of their Subsidiaries, if applicable, to issue or cause the publication of any press release or other public announcement with respect to, or otherwise make any public statement concerning, the transactions contemplated by this Agreement without the consent of the other party.

- **6.14 Indemnification.** Parent agrees that it will, after the Effective Time, provide to those individuals who have served as directors or officers of the Company indemnification equivalent to that provided by the Certificate of Incorporation and Bylaws of the Company with respect to matters occurring prior to the Effective Time, for a period of six (6) years from the Effective Time (or, in the case of matters occurring prior to the Effective Time which, have not been resolved prior to the sixth (6th) anniversary, until such matters are finally resolved). To the extent permitted by law, Parent will advance expenses in connection with the foregoing indemnification. In the event Parent or any of its successors or assigns (a) consolidates with or merges into any other person and Parent shall not be the continuing or surviving corporation or entity of such consolidation or merger or (b) transfers all or substantially all of its properties and assets or any person, then and in each such case, proper provision shall be made so that the successors and assigns of Parent shall assume the obligations set forth in this Section 6.14.
- **6.15 Termination of Company Warrants, Stock Agreement and Profit Sharing Plan.** Prior to the Closing, the Company shall take all actions necessary to cause each Company Warrant to be validly exercised or terminated pursuant to its terms, to cause the termination of the Stock Agreement dated as of July 25, 2002 between the Company and L. M. Earle, and to

38

cause the termination of the profit sharing plan of the Company reflected in <u>Schedule 3.05</u> hereto, and the Company shall provide Parent satisfactory documentation of any such exercise or termination.

6.16 Obligations of Surviving Corporation. Parent agrees that it will, promptly after the Effective Time, (a) pay in full no later than five business days after the Closing Date all obligations to the Internal Revenue Service as necessary to satisfy the Estimated Payroll Tax Obligations, (b) pay in full all obligations owing KeyBank National Association under the Company's loan #0100427938/20001, loan #0100417186/10001 to Donna Galeti, loan #0100415800/00001 to Mark A. Polly, and loan #0100415966/00001 to James J. Miller, (c) provide a standby letter of credit in substitution of the Company's Standby Letter of Credit #S030367300, and (d) file with the Internal Revenue Service under the Delinquent Filer Voluntary Compliance Program all Forms 5500 for the Plans that have not previously been filed as required and pay all obligations arising in connection therewith.

ARTICLE VII CONDITIONS PRECEDENT

- **7.01 Conditions to Each Party's Obligation to Effect the Merger.** The respective obligation of each party to effect the Merger shall be subject to the satisfaction at or prior to the Closing of the following conditions:
- (a) <u>Regulatory Approvals</u>. All necessary approvals, authorizations and consents of all Governmental Entities required to consummate the transactions contemplated hereby shall have been obtained and shall remain in full force and effect and all statutory waiting periods in respect thereof shall have expired or been terminated (all such approvals and the expiration of all such waiting periods being referred to herein as the "*Requisite Regulatory Approvals*").
- (b) No Injunctions or Restraints; Illegality. No order, injunction or decree issued by any court or agency of competent jurisdiction or other legal restraint or prohibition (an "Injunction") preventing the consummation of the Merger or any of the other transactions contemplated by this Agreement shall be in effect and no proceeding initiated by any Governmental Entity seeking an injunction shall be pending. No statute, rule, regulation, order, injunction or decree shall have been enacted, entered, promulgated or enforced by any Governmental Entity which prohibits, restricts or makes illegal consummation of the Merger, or any of the other transactions contemplated by this Agreement.
- **7.02 Conditions to Obligations of Parent and Merger Sub.** The obligation of Parent and Merger Sub to effect the Merger is also subject to the satisfaction or waiver by Parent or Merger Sub, at or prior to the Effective Time, of the following conditions:
- (a) Stockholder Approval. This Agreement shall have been approved and adopted by the requisite vote of the Stockholders in accordance with the DGCL and the Company's Certificate of Incorporation and Bylaws and by each Company Preferred Stockholder.

39

(b) Representations and Warranties. The representations and warranties of the Company set forth in this Agreement shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date (except to the extent such representations and warranties speak as of an earlier date), except to the extent that such representations and warranties are qualified by the term "material" or contain terms such as "Company Material Adverse Effect" in which case such representations and warranties shall be true and correct in all respects as of the date of this Agreement and at and as of the Closing Date (except to the extent such representations and warranties speak as of an earlier date). Parent shall receive at Closing a certificate signed on behalf of the Company by an authorized officer to the foregoing effect.

- (c) <u>Performance of Obligations of the Company</u>. The Company shall have performed all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and Parent shall receive at Closing a certificate signed on behalf of the Company by an authorized officer to such effect.
- (d) <u>Consents Under Agreements</u>. The consent, approval, waiver or amendment of each person (other than the Governmental Entities) set forth on Schedule 3.04 hereto shall have been obtained and shall be reasonably satisfactory to Parent.
- (e) <u>Stockholder Representation Letters</u>. Each of the Company Preferred Stockholders shall have executed and delivered to Parent a Stockholder Representation Letter.
- (f) <u>Stock Restriction Agreements</u>. Each of the Continuing Employee Stockholders shall have executed Stock Restriction Agreements with Parent.
- (g) <u>Non-Compete Agreements</u>. Each of the persons listed on <u>Schedule 6.12</u> shall have executed and delivered to Parent a Non-Compete Agreement.
- (h) <u>FIRPTA Compliance</u>. The Company shall, prior to the Closing Date, provide Parent with a properly executed Foreign Investment in Real Property Tax Act of 1980 ("*FIRPTA*") Notification Letter, in form and substance satisfactory to Parent, which states that shares of capital stock of the Company do not constitute "United States real property interests" under Section 897(c) of the Code, for purposes of satisfying Parent's obligations under Treasury Regulation Section 1.1445-2(c)(3). In addition, simultaneously with delivery of such Notification Letter, the Company shall have provided to Parent, as agent for the Company, a form of notice to the Internal Revenue Service in accordance with the requirements of Treasury Regulation Section 1.897-2(h)(2) along with written authorization for Parent to deliver such notice form to the Internal Revenue Service on behalf of the Company upon the consummation of the Merger.
- (i) <u>Resignations</u>. The Company shall have received written letters of resignation from each of the current members of the Board of Directors of the Company, and each officer of the Company shall resign from such office, effective at the Effective Time.
- (j) <u>Termination of the Company's Agreements</u>. Parent shall have been furnished evidence satisfactory to it that all rights granted by the Company to its Stockholders and in effect prior to the Closing, including, but not limited to, rights of co-sale, voting,

40

registration, first refusal, first offer, preemptive, board observation or information or operational covenants, shall have been terminated or shall terminate as of the Effective Time.

- (k) <u>Termination of the Stock Plan</u>. Parent shall have been furnished evidence satisfactory to it that the Company has effected the termination of the Stock Plan and has satisfied all requirements under the Stock Plan or otherwise to effect such termination, including, but not limited to, the giving of any notice and obtaining any consents, amendments or waivers required by the Stock Plan or any agreement relating to the Company Options.
- (l) <u>Termination of Company Warrants, Stock Agreement and Profit Sharing Plan</u>. Parent shall have been furnished evidence satisfactory to it that all Company Warrants have been exercised or terminated or shall terminate as of the Effective Time that all rights granted by the Company under the Stock Agreement dated as of July 25, 2002 between the Company and L. M. Earle have been terminated or shall terminate as of the Effective Time, and that all rights of employees of the Company under the profit sharing plan of the Company have been terminated or shall terminate as of the Effective Time.
- (m) Opinion of Counsel for Company. The Parent and Merger Sub shall have received an Opinion of Counsel of the Company in substantially the form attached as Exhibit K.
- (n) <u>Escrow Agreement</u>. The Parent, Merger Sub, Stockholder Representative and the Escrow Agent shall each have executed and delivered the Escrow Agreement.
- (o) <u>No Material Adverse Changes</u>. During the period between the execution of this Agreement and the Closing Date, there shall not have been any Company Material Adverse Effect and no fact or condition specific to the Company shall exist which has had or would reasonably be expected to cause such a Company Material Adverse Effect after the Closing.
- **7.03 Conditions to Obligations of the Company.** The obligations of the Company to effect the Merger is also subject to the satisfaction, or waiver by the Company, at or prior to the Closing of the following conditions:
- (a) Representations and Warranties. The representations and warranties of Parent and Merger Sub set forth in this Agreement shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date (except to the extent such representations and warranties speak as of an earlier date), except to the extent that such representations and warranties are qualified by the term "material" or contain terms such as "Parent Material Adverse Effect" in which case such representations and warranties shall be true and correct in all respects as of the date of this Agreement and at and as of the Closing Date (except to the extent such representations and warranties speak as of an earlier date). The Company shall receive at Closing a certificate signed on behalf of Parent by an authorized officer to the foregoing effect.
- (b) <u>Performance of Obligations of Parent and Merger Sub</u>. Parent and Merger Sub shall have each performed all obligations required to be performed by them under this Agreement at or prior to the Closing Date, and the Company shall receive at Closing a certificate

- (c) <u>Opinion of Counsel for the Parent and Merger Sub</u>. The Company shall have received an opinion of counsel of Parent and Merger Sub in form and substance reasonably acceptable to the parties.
- (d) <u>Escrow Agreement</u>. The Company, Parent, Merger Sub, Stockholder Representative and the Escrow Agent shall each have executed and delivered the Escrow Agreement.
- (e) <u>No Material Adverse Changes</u>. During the period between the execution of this Agreement and the Closing Date, there shall not have been any Parent Material Adverse Effect and no fact or condition specific to the Parent shall exist which has had or would reasonably be expected to cause such a Parent Material Adverse Effect after the Closing.

ARTICLE VIII TERMINATION AND AMENDMENT

- **8.01 Termination.** This Agreement may be terminated and the Merger abandoned at any time prior to the Effective Time:
- (a) by mutual consent of Parent and the Company in a written instrument, if the Board of Directors of each so determines by a vote of a majority of the members of its entire Board;
- (b) by either Parent or the Company (provided, however, that the right to terminate this Agreement under this clause shall not be available to any party who's breach or failure to fulfill any of its obligations under this Agreement has been the cause of or resulted in the failure of the Closing to occur) if there shall have been any material breach of any of the covenants or agreements set forth in this Agreement on the part of the other party, or any of the representations and warranties of such party shall cease to be materially true and correct, such that the provisions of Sections 7.02(b) and 7.02(c) or Sections 7.03(a) and 7.03(b), as the case may be, would not be satisfied and such breach has not been cured within ten (10) days after notice thereof to the breaching party; provided, however, that no cure period shall be required for a breach which by its nature cannot be cured; or
- (c) by either Parent or the Company if the Closing shall not have occurred by July 31, 2004; provided, however, such date may be increased by an additional thirty (30) days at the request of the Parent if the Closing is delayed solely because any Requisite Regulatory Approval has not been obtained and Parent is diligently undertaking such efforts required to obtain the same; provided, further, that the right to terminate this Agreement under this Section 8.01(c) shall not be available to any party whose actions or failure to act has been a primary cause of, or resulted in, the failure of the Merger to occur on or before such date and such action or failure to act constitutes a breach of this Agreement.
- **8.02 Effect of Termination.** In the event of termination of this Agreement by either Parent or the Company as provided in Section 8.01, this Agreement shall forthwith become void

42

and have no effect, except that <u>Section 6.04(b)</u> shall survive any termination of this Agreement for a period of two (2) years following such termination, and there shall be no further obligation on the part of Parent, Merger Sub, the Company, or their respective officers or directors or the Stockholders except for the obligations under such provisions. Notwithstanding anything to the contrary contained in this Agreement, no party shall be relieved or released from any liabilities or damages arising out of its intentional breach of any provision of this Agreement.

- **8.03 Expenses.** Regardless of whether the transactions contemplated by this Agreement close, each party shall bear its own costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby.
 - **8.04 Amendment.** This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.
- **8.05 Extension; Waiver.** Any agreement on the part of a party hereto to (a) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (b) waive any inaccuracies in the representations and warranties contained herein or in any document delivered pursuant hereto or (c) waive compliance with any of the agreements or conditions contained herein shall be valid only if set forth in a written instrument signed on behalf of such party, but such extension or waiver shall nor operate as a waiver of, or estoppel with respect to, any subsequent or other failure.

ARTICLE IX INDEMNIFICATION

- **9.01 Agreement to Indemnify.** Following the Closing and subject to the limitations set forth herein,
- (a) the Company Preferred Stockholders shall severally indemnify and agree to defend and hold harmless Parent and the Surviving Corporation (and their respective affiliates, officers, directors, employees, representatives and agents) ("*Purchaser Indemnities*" and, singularly, a "*Purchaser Indemnitee*") against and in respect of any and all Damages, by reason of or otherwise arising out of:
 - (i) Parent Indemnified Taxes;
- (ii) any liability for past-due payroll tax obligations of the Company relating to periods prior to the Closing Date, to the extent the Surviving Corporation's liability for all such past-due payroll obligations exceeds the Estimated Payroll Tax Obligation;
- (iii) any claim for software license fees or penalties arising as a result of the audit of the Company's software use occurring on or about the date of this Agreement;
 - (iv) any liability of the Surviving Corporation related to the failure to timely file any required Forms 5500 for the Plans;

- (v) the Adjustment of Merger Consideration pursuant to Section 2.04 to the extent the Additional Merger Consideration Adjustment is greater than the Cash Holdback Amount;

 (vi) any claim by an employee or former employee of the Company, or any other person or entity, based upon (A) such
- (vi) any claim by an employee or former employee of the Company, or any other person or entity, based upon (A) such employee's or former employee's employment with the Company prior to the Effective Time, (B) the termination of employment of current or former employees of the Company pursuant to this Agreement or otherwise prior to the Effective Time, or (C) any severance arrangements or payments or any benefit, salary, bonus, commission or other compensation payments made or required to be made in connection with such terminations prior to the Effective Time:
- (vii) any claim by a Stockholder or former stockholder of the Company, or any other person or entity, against the Company or any of its affiliates, officers, directors, employees or agents, based upon (A) any rights of a stockholder (other than the right of Company Preferred Stockholders to receive Merger Consideration pursuant to this Agreement), including appraisal rights under the applicable provisions of the DGCL, any option, preemptive rights or rights to notice or to vote, or (B) any claim of false or misleading information relating to the Company or the transactions contemplated hereby included in the Information Statement; or
 - (viii) a breach by the Company of a representation, warranty or covenant contained in this Agreement.
- "Damages" shall include, reasonable attorneys' fees and disbursements, reasonable accountants' fees and disbursements, costs of litigation and other expenses incurred by them (or their respective affiliates, officers, directors or employees) in the defense of any claim asserted against them (or their respective affiliates, officers, directors, employees or agents) and any amounts paid in settlement or compromise of any claim asserted against them to the extent that the claim asserted is or would have been subject to the indemnification provisions hereof, subject to the limitations on indemnification set forth in Sections 9.02 and 9.03. "Damages" shall not include any amount for which reimbursement is received by Parent, the Surviving Corporation or the Company, as the case may be, pursuant to insurance policies or third-party payments by virtue of indemnification or subrogation received by such party which the Parent, the Company and the Surviving Corporation shall use their best efforts to pursue, and shall be determined net of any tax benefit actually realized by the Indemnified Party as a result of the claim. "Parent Indemnified Taxes" means any and all Taxes together with any costs, expenses, losses or damages (including court and administrative costs and reasonable legal fees and expenses incurred in investigating and preparing for any audit, litigation or other proceeding) arising out of or incident to the determination, assessment or collection of such Taxes (i) imposed on the Company or any of its Subsidiaries for which it may otherwise be liable for any taxable period or portion thereof ending on or prior to the Effective Time, including as a result of the transactions contemplated by the Agreement (subject to the availability of the Company's net operating losses for such period), (ii) resulting from the breach of the representations and warranties set forth in Section 3.08 (determined without regard to any materiality or knowledge qualifiers or any scheduled items that may be contained therein) or co

44

Company pursuant to Treasury Regulation §1.1502-6 or any analogous or similar state, local or foreign law, as a successor or transferee, by contract or otherwise; but only to the extent such Taxes were not included in Closing Date Debt. Parent Indemnified Taxes shall also include any cost or expense of preparing federal and state income Tax Returns filed after the Closing Date with respect to taxable periods ending on or before the Closing Date.

- (b) Parent shall indemnify and agrees to defend and hold harmless the Stockholders (and their respective affiliates, representatives and agents) against and in respect of any and all Damages by reason of or otherwise arising out of
 - (i) a breach by Parent or Merger Sub of any representation, warranty or covenant contained in this Agreement; or
- (ii) any claim of false or misleading information relating to Parent included in the Information Statement or Parent SEC Filings.
- 9.02 Survival of Indemnity. The representations and warranties and indemnification obligations of the Company and each Indemnifying Party pursuant to Section 9.01 shall survive the Closing for a period of twelve (12) months, except for Damages arising out of a breach of any of the representations or warranties in either Section 3.08 or Section 3.13, which shall survive for a period equal to the applicable statute of limitations on any claim relating thereto, and any claims for indemnification in accordance with this ARTICLE IX with respect to any representation or warranty must be made (and will be null and void unless made) prior to the end of the applicable survival period. Upon expiration of such periods, no Indemnifying Party shall have any liability for Damages under such indemnification obligations unless it has received written notice from an Indemnified Party claiming indemnification prior to the expiration of the applicable period as required.

9.03 Additional Provisions.

- (a) <u>Limitations on Indemnified Amounts of the Company Preferred Stockholders</u>. The Company Preferred Stockholders shall have the obligation to indemnify parties under this <u>ARTICLE IX</u> with respect to all matters listed in <u>Section 9.01(a)</u> or as otherwise provided herein, except that the Company Preferred Stockholders shall have the obligation to indemnify parties with respect to matters arising under <u>Section 9.01(a)(viii)</u> only to the extent the Company Preferred Stockholders' aggregate indemnity obligations under <u>Section 9.01(a)(viii)</u> exceed \$50,000.00. Notwithstanding the preceding sentence, in no event shall any Company Preferred Stockholder's aggregate indemnity obligations exceed an amount equal to 50% of the aggregate Merger Consideration received by such Company Preferred Stockholder. The liability of the Company Preferred Stockholders for indemnification under this <u>ARTICLE IX</u> by reason of or arising out of any breach by the Company of any representation or warranty shall not be modified, waived or diminished by any examination or investigation conducted by Parent of the books, records or operations of the Company.
- (b) <u>Limitations on Indemnified Amounts of Parent</u>. Parent shall have the obligation to indemnify the Company under this <u>ARTICLE IX</u> with respect to all matters listed in <u>Section 9.01(b)</u>, except that Parent shall have the obligation to indemnify the Company under

obligation of the Preferred Stockholders as provided in <u>Section 9.03(a)</u>. The liability of Parent for indemnification under this <u>ARTICLE IX</u> by reason of or arising out of any breach by Parent or Merger Sub of any representation or warranty shall not be modified, waived or diminished by any examination or investigation conducted by the Company of the books, records or operations of Parent and Merger Sub.

- (c) <u>Satisfaction of Indemnification Obligations</u>. Parent and Merger Sub agree that all Damages shall be satisfied as follows:
- (i) First with the Escrowed Shares, until the aggregate amount of unresolved indemnification claims made for the benefit of the Purchaser Indemnities exceeds the value of the Escrowed Shares: and
- (ii) Second, the Parent Indemnified Parties shall be free to pursue Damages directly against the Company Series A Preferred Stockholders subject to the maximum potential indemnification obligation of the Company Preferred Stockholders as provided in <u>Section 9.03(a)</u>.

Notwithstanding the preceding sentence, (x) the Stockholders may at their option through a notice by the Stockholder Representative to Parent within ten (10) days of the final determination of any Damages, satisfy the payment of any Damages through the payment of cash in lieu of the Escrowed Shares or with shares of Parent Common Stock issued as Merger Consideration in lieu of the payment of cash, and (y) Parent may offset any Damages for items listed in Sections 9.01(a)(i)-9.01(a)(vii) against any and all amounts payable to the Company Series A Preferred Stockholders in accordance with Section 2.04(d). Any shares of Parent Common Stock used to satisfy such Damages shall be valued at the Parent Stock Per Share Price.

- (d) <u>No Limitation in Event of Fraud.</u> Notwithstanding any other provision hereof, nothing in this <u>ARTICLE IX</u> (including the provisions of paragraphs (a), (b) or (c) of this <u>Section 9.03</u>) or otherwise shall limit, in any manner, any remedy at law or equity, to which any party may be entitled as a result of fraud by any Indemnifying Party or its employees, officers or directors or a violation of the federal securities laws.
- (e) <u>Exclusivity of Remedy; Survival of Covenants</u>. Following the Closing, except in respect of claims based upon fraud or violation of the federal securities laws, the indemnification accorded by this Section shall be the sole and exclusive remedy of the parties indemnified under this <u>ARTICLE IX</u> in respect of any misrepresentation or inaccuracy in, or breach of, any representation or warranty or any breach or failure in performance of any covenant or agreement made in this Agreement or in any document or certificate delivered pursuant hereto other than the Employment Agreements. Notwithstanding the foregoing, in the event of any breach or failure in performance after the Closing of any covenant or agreement, a non-breaching party shall also be entitled to seek specific performance, injunctive or other equitable relief. The covenants of any party shall terminate according to the terms of such covenant and the expiration of the applicable statutes of limitations.

46

(f) <u>Subrogation</u>. Upon making any payment to an Indemnified Party for any indemnification claim pursuant to this <u>ARTICLE IX</u>, an Indemnifying Party shall be subrogated, to the extent of such payment, to any rights that the Indemnified Party may have against any other persons with respect to the subject matter underlying such indemnification claim and the Indemnified Party shall take such actions as the Indemnifying Party may reasonably require to perfect such subrogation or to pursue such rights against such other persons as the Indemnified Party may have.

9.04 Claim Notice; Definitions; Third Party Claim Procedures.

- (a) <u>Claim Notice</u>. An Indemnified Party shall give each Indemnifying Party from whom indemnification is sought prompt written notice (a "Claim Notice") of any claim, demand, action, suit, proceeding or discovery of fact upon which the Indemnified Party intends to base the claim for indemnification under this <u>ARTICLE IX</u>, which shall contain (i) a description and the amount of any Damages incurred or reasonably expected to be incurred by the Indemnified Party, (ii) a statement that the Indemnified Party is entitled to indemnification under this <u>ARTICLE IX</u> for such Damages, and (iii) a demand for payment, provided, however, that no failure to give such Claim Notice shall excuse any Indemnifying Party from any obligation hereunder except to the extent the Indemnifying Party is materially and actually prejudiced by such failure. Parent, Merger Sub, the Company and the Stockholder's Representative agree that the procedures set forth in the Escrow Agreement with respect to Claim Notices and responses thereto shall govern all claims made against the Escrowed Shares.
- (b) <u>Definitions</u>. The term "*Indemnified Party*" shall mean a party (or its successor) who is entitled to indemnification from a party hereto pursuant to this <u>ARTICLE IX</u>; the term "*Indemnifying Party*" shall mean a party (or its successor) hereto who is required to provide indemnification under this <u>ARTICLE IX</u> to another party; and the term "*Third Party Claim*" shall mean any claim, action, suit, proceeding, investigation or like matter which is asserted or threatened by a party other than the parties hereto, their successors and permitted assigns, against any Indemnified Party or to which any Indemnified Party is subject.
- (c) Procedure. The Indemnified Party may, upon reasonable notice, tender the exclusive defense of a Third Party Claim (subject to the provisions of this Section 9.04(c)) to the Indemnifying Party. If (i) the defense of a Third Party Claim is so tendered and within thirty (30) days thereafter such tender is accepted without qualification (or reservation of rights) by the Indemnifying Party; or (ii) within thirty (30) days after the date on which written notice of a Third Party Claim has been given pursuant to this Section 9.04(c), the Indemnifying Party shall acknowledge in writing to the Indemnified Party and without qualification (or reservation of rights) its indemnification obligations as provided in this ARTICLE IX; then, except as hereinafter provided, the Indemnified Party shall not, and the Indemnifying Party shall, have the right to contest, defend, litigate or settle such Third Party Claim. The Indemnifying Party provided that the Indemnified Party shall be entitled to reimbursement therefor if the Indemnifying Party shall lose its right to contest, defend, litigate and settle the Third Party Claim as herein provided. The Indemnifying Party shall lose its right to defend and settle the Third Party Claim if it shall fail to diligently contest, defend, litigate and settle the Third Party Claim as provided herein. So long as the

47

Indemnifying Party has not lost its right, defend, litigate and settle and/or obligation to contest, defend, litigate and settle as herein provided, the Indemnifying Party shall have the exclusive right to contest, defend and litigate the Third Party Claim and shall have the right, upon receiving the prior written approval of the Indemnified Party (which shall not be unreasonably withheld unless such settlement does not fulfill the conditions set forth in the following sentence and which shall be deemed automatically given if a response has not been received within the thirty (30) day period following receipt of the proposed settlement by the Indemnified Party), to settle any such matter, either before or after the initiation of litigation, at such time and upon such terms as it deems fair and

reasonable. Notwithstanding anything to the contrary herein contained, in connection with any settlement negotiated by an Indemnifying Party, no Indemnified Party or Indemnifying Party (as the case may be) that is not controlling the defense and or settlement of the Third Party Claim (the "Non-Control *Party*") shall be required by an Indemnifying Party or Indemnified Party controlling the litigation to (and no such party shall) (x) enter into any settlement that does not include as an unconditional term thereof the delivery by the claimant or plaintiff to the Non-Control Party of a release from all liability in respect of such claim or litigation, (y) enter into any settlement that attributes by its terms liability to the Non-Control Party or which may otherwise have an adverse effect on the Indemnified Party's business, or (z) consent to the entry of any judgment that does not include as a term thereof a full dismissal of the litigation or proceeding with prejudice. All expenses (including attorneys' fees) incurred by the Indemnifying Party in connection with the foregoing shall be paid by the Indemnifying Party. No failure by an Indemnifying Party to acknowledge in writing its indemnification obligations under this ARTICLE IX shall relieve it of such obligations to the extent they exist. If an Indemnified Party is entitled to indemnification against a Third Party Claim, and the Indemnifying Party fails to accept a tender of, or assume, the defense of a Third Party Claim pursuant to this ARTICLE IX, or if, in accordance with the foregoing, the Indemnifying Party does not have the right or shall lose its right to contest, defend, litigate and settle such a Third Party Claim, the Indemnified Party shall have the right, without prejudice to its right of indemnification hereunder, in its discretion exercised in good faith and upon the advice of counsel, to contest, defend and litigate such Third Party Claim, and may settle such Third Party Claim, either before or after the initiation of litigation, at such time and upon such terms as the Indemnified Party deems fair and reasonable, provided that at least twenty (20) days prior to any such settlement, written notice of its intention to settle is given to the Indemnifying Party. If, pursuant to this Section 9.04(c), the Indemnified Party so contests, defends, litigates or settles a Third Party Claim, for which it is entitled to indemnification hereunder as provided herein, the Indemnified Party shall be reimbursed by the Indemnifying Party for the reasonable attorneys' fees and other expenses of defending, contesting, litigating and/or settling the Third Party Claim which are incurred from time to time, forthwith following the presentation to the Indemnifying Party of itemized bills for said attorneys' fees and other expenses. The Indemnified Party or the Indemnifying Party, as the case may be, shall furnish such information in reasonable detail as it may have with respect to a Third Party Claim (including copies of any summons, complaint or other pleading which may have been served on such party and any written claim, demand, invoice, billing or other document evidencing or asserting the same) to the other party if such other party is assuming defense of such claim, and make available all records and other similar materials which are reasonably required in the defense of such Third Party Claim and shall otherwise cooperate with and assist the defending party in the defense of such Third Party Claim.

48

ARTICLE X STOCKHOLDERS REPRESENTATIVE

10.01 Appointment of Stockholders Representative. The initial Stockholders Representative shall be Robert Honner (the "Stockholders Representative"). The Stockholders Representative shall be the attorney-in-fact and agent of Stockholders with respect to the matters set forth in this Agreement. Notwithstanding anything to the contrary set forth in this Agreement, the Stockholder Representative shall have no duties or responsibilities except those expressly set forth herein, and no implied covenants, functions, responsibilities, duties, obligations or liabilities on behalf of any Stockholder shall otherwise exist against the Stockholder Representative. The foregoing power is irrevocable and coupled with an interest, and shall not be affected by the death, incapacity, illness, dissolution or other inability to act of any of the Stockholders.

- **10.02 Authority.** Each Company Preferred Stockholder, pursuant to the Stockholder Representation Letter, shall grant the Stockholder Representative full power and authority:
- (a) to execute and deliver, on behalf of such Stockholder, and to accept delivery of, on behalf of such Stockholder, such documents as may be deemed by the Stockholder Representative, in his sole discretion, to be appropriate to consummate this Agreement;
- (b) to (i) dispute or refrain from disputing, on behalf of such Stockholder, any claim made by Parent or the Surviving Corporation under this Agreement; (ii) negotiate and compromise, on behalf of such Stockholder, any dispute that may arise under, and to exercise or refrain from exercising any remedies available under this Agreement, and (iii) execute, on behalf of such Stockholder, any settlement agreement, release or other document with respect to such dispute or remedy;
- (c) to give or agree to, on behalf of such Stockholder, any and all consents, waivers, amendments or modifications, deemed by the Stockholder Representative, in his sole discretion, to be necessary or appropriate, under this Agreement, and, in each case, to execute and deliver any documents that may be necessary or appropriate in connection therewith;
- (d) to enforce, on behalf of such Stockholder, any claim against Parent, Merger Sub or the Surviving Corporation arising under this Agreement; and
- (e) to give such instructions and to take such action or refrain from taking such action, on behalf of such Stockholders, as the Stockholder Representative deems, in their sole discretion, necessary or appropriate to carry out the provisions of this Agreement.
- **10.03 Reliance.** Each Company Preferred Stockholder, pursuant to the Stockholder Representation Letter, shall agree that: (a) in all matters in which action by the Stockholder Representative is required or permitted, the Stockholder Representative is authorized to act on behalf of such Stockholder, notwithstanding any dispute or disagreement among Stockholders or between any Stockholder and the Stockholders Representative, and Parent, Merger Sub and the Surviving Corporation shall be entitled to rely on any and all action taken by the Stockholder

49

Representative, under this Agreement without any liability to, or obligation to inquire of, any of the Stockholders notwithstanding any knowledge on the part of the Parent of any such dispute or disagreement; (b) the power and authority of the Stockholder Representative, as described in this Agreement, shall be effective until all rights and obligations of Stockholders under this Agreement have terminated, expired or been fully performed; and (c) if the Stockholder Representative resigns or otherwise ceases to function in his or her capacity as such for any reason whatsoever, a majority of the Stockholders shall have the right, exercisable upon written notice delivered to Purchaser to appoint another individual to serve as a new Stockholder Representative to fill the vacancy caused by the circumstance described above.

10.04 Indemnification of Parent, Merger Sub and Their Affiliates. Company Preferred Stockholders, pursuant to the Stockholder Representation Letter, shall jointly and severally indemnify the Purchaser Indemnities against, and agree to hold the Purchaser Indemnities harmless from, any and all Damages incurred or suffered by any Purchaser Indemnitee arising out of, with respect to or incident to the operation of, or any breach of any covenant or agreement pursuant to, this <u>ARTICLE X</u> by a Stockholder or a Stockholder Representative, or the designation, appointment and actions of the Stockholder Representative pursuant to the provisions hereof, including with respect to (a) actions taken by the Stockholder Representative, and (b) reliance in good faith by any Purchaser Indemnitee on, and actions in good faith taken by any Purchaser Indemnitee in response to or in reliance on, the instructions of, notice given by or any other action taken by the Stockholder Representative.

10.05 Indemnification of Stockholder Representative. Each Company Preferred Stockholder, pursuant to the Stockholder Representation Letter, shall severally indemnify and hold any Person serving as the Stockholder Representative harmless from and against any Damages (except as result from such Person's bad faith, gross negligence or willful misconduct) that such Person may suffer or incur in connection with any action taken by such Person as the Stockholder Representative. Each Stockholder shall bear its pro-rata portion of such Damages. No Person serving as Stockholder Representative shall be liable to any Stockholder with respect to any action or omission taken or omitted to be taken by the Stockholder Representative pursuant to this ARTICLE X, except for such Person's gross negligence or willful misconduct. No Stockholder Representative shall be responsible in any manner whatsoever for any failure or inability of Parent or Merger Sub, or of anyone else, to honor any of the provisions of this Agreement. The Stockholder Representative shall be fully protected by Stockholders in acting on and relying upon any written notice, direction, request, waiver, notice, consent, receipt or other paper or document which they in good faith believe to be genuine and to have been signed or presented by the proper party or parties. The Stockholder Representative shall not be liable to the Stockholders for any error of judgment, or any act done or step taken or omitted by any of them in good faith or for any mistake in fact or law, or for anything which any of them may do or refrain from doing in connection herewith, except for their own bad faith, willful misconduct or gross negligence. The Stockholder Representative may seek the advice of legal counsel, engage experts or otherwise incur reasonable expenses in the event of any dispute or question as to the construction of any of the provisions of this Agreement or their duties hereunder, and they shall incur no liability to Stockholders with respect to any action taken, omitte

50

expenses, and, in addition to any and all other remedies available, the Stockholder Representative shall have the right to set-off against any amounts due to the Stockholders.

ARTICLE XI GENERAL PROVISIONS

11.01 Notices. All notices and other communications hereunder shall be in writing and shall be deemed given when delivered personally or telecopied (with confirmation from recipient) provided that a copy of all telecopies is sent by one of the other delivery methods set forth in this <u>Section 11.01</u> within one (1) day of being telecopied, three (3) days after mailed by registered or certified mail (return receipt requested) or on the day delivered by an express courier (with confirmation from recipient) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice):

(a) if to Parent or Merger Sub, to:

Perficient, Inc. 1120 S. Capital of Texas Highway Building 3, Suite 220 Austin, Texas 78746

Attn: John T. McDonald, Chief Executive Officer

Phone: (512) 531-6000 Facsimile: (512) 531-6011

with a copy to:

Vinson & Elkins L.L.P.
The Terrace 7
2801 Via Fortuna, Suite 100
Austin, Texas 78746
Attn: J. Nixon Fox III, Esq.
Phone: (512) 542-8427
Facsimile: (512) 236-3216

(b) if to the Company, to:

Meritage Technologies, Inc. 1400 Commerce Center Drive Suite D

Franklin, Ohio 45005 Attn: Chris Gianattasio

Phone: (937) 743-1870 ext. 104 Facsimile: (937) 743-1874 Greenhill and Co., Inc. 300 Park Avenue New York, New York 10022

Attention: John Liu

(c) if the Stockholder Representative, to:

Robert Honner 3160 Terra Drive Columbus, Ohio 43228 Phone: (614) 487-1956

with a copy to:

Greenhill and Co., Inc. 300 Park Avenue New York, New York 10022

Attention: John Liu

- **11.02 Interpretation.** When a reference is made in this Agreement to Sections, Exhibits or Schedules, such reference shall be to a Section of or Exhibit or Schedule to this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "*include*," "*includes*" or "*including*" are used in this Agreement, they shall be deemed to be followed by the words "*without limitation*."
- **11.03 Counterparts.** This Agreement may be executed in counterparts, all of which shall be considered one and the same agreement and shall become effective when counterparts have been signed by each of the parties and delivered to the other parties, it being understood that all parties need not sign the same counterpart.
- **11.04 Entire Agreement.** This Agreement (including the documents and the instruments referred to herein) constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof.
- **11.05 Governing Law.** This Agreement shall be governed and construed in accordance with the laws of the State of Delaware, without regard to any applicable conflicts of law principles thereof.
- **11.06 Enforcement of Agreement.** The parties hereto agree that irreparable damage would occur in the event that the provisions contained in Sections 6.05 or 6.12 of this Agreement were not performed in accordance with its specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of Sections 6.05 or 6.12 of this Agreement and to enforce specifically the terms and

52

provisions thereof in any court of the United States or any court located in the city of Chicago in the State of Illinois, this being in addition to any other remedy to which they are entitled at law or in equity.

- **11.07 Severability.** Any term or provision of this Agreement which is invalid or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Agreement or affecting the validity or enforceability of any of the terms or provisions of this Agreement in any other jurisdiction. If any provision of this Agreement is deemed to be so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.
- **11.08 Assignment.** Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto (whether by operation of law or otherwise) without the prior written consent of the other parties. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of and be enforceable by the parties and their respective successors and assigns. Except as otherwise expressly provided herein, this Agreement (including the documents and instruments referred to herein) is not intended to confer upon any person other than the parties hereto any rights or remedies hereunder.
- **11.09 Amendment.** This Agreement may be amended with respect to any of the terms contained herein only by written agreement, signed by each of the parties hereto.

[Signature Page Follows]

53

IN WITNESS WHEREOF, Parent, Merger Sub and the Company have caused this Agreement to be executed by their respective officers thereunto duly authorized as of the date first above written.

PERFICIENT, INC.

By:

Name: Jeffrey Davis Title: President

By: Name: Michael Hill Title: Vice President MERITAGE TECHNOLOGIES, INC. By: Name: Christopher Gianattasio Title: Chief Executive Officer STOCKHOLDER REPRESENTATIVE Robert Honner

SIGNATURE PAGE TO AGREEMENT AND PLAN OF MERGER

PERFICIENT MERITAGE, INC.

SECURITIES PURCHASE AGREEMENT

This SECURITIES PURCHASE AGREEMENT (this "<u>Agreement</u>"), dated as of June 16, 2004 is made by and among Perficient, Inc., a Delaware corporation, with headquarters located at 1120 South Capital of Texas Highway, Suite 220, Building 3, Austin, Texas 78746 (the "<u>Company</u>"), and the investors named on the signature pages hereto, together with their permitted transferees (the "<u>Investors</u>").

RECITALS:

- A. The Company and the Investors are executing and delivering this Agreement in reliance upon the exemptions from securities registration afforded by Section 4(2) of the Securities Act and Rule 506 under Regulation D.
- B. The Investors desire, upon the terms and conditions stated in this Agreement, to purchase an aggregate of 800,000 shares of Common Stock of the Company (the "Common Shares") at \$3.09 per share, for an aggregate purchase price of \$2,472,000.00, and to receive, in consideration for such purchase, Warrants to acquire additional shares of Common Stock.
- C. Contemporaneously with the execution and delivery of this Agreement, the parties hereto are executing and delivering a Registration Rights Agreement under which the Company has agreed to provide certain registration rights under the Securities Act, the rules and regulations promulgated thereunder and applicable state securities laws.
 - D. The capitalized terms used herein and not otherwise defined have the meanings given them in Article X hereof.

In consideration of the premises and the mutual covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Company and the Investors hereby agree as follows:

ARTICLE I PURCHASE AND SALE OF COMMON STOCK

- 1.1 <u>Purchase and Sale of Common Stock.</u> At the Closing, subject to the terms of this Agreement and the satisfaction or waiver of the conditions set forth in Articles VI and VII hereof, the Company will issue and sell to each Investor, and each Investor will (on a several and not a joint basis) purchase from the Company, such number of Common Shares set forth opposite such Investor's name on <u>Exhibit A</u> hereto under the heading "Common Shares."
- 1.2 <u>Payment</u>. Each Investor will pay the purchase amount for the Common Shares as is set forth opposite such Investor's name on <u>Exhibit A</u> hereto under the heading "Purchase"

Price," by wire transfer of immediately available funds in accordance with the Company's written wire instructions, simultaneously with delivery by the Company to each Investor of (i) one or more stock certificates, free and clear of all restrictive and other legends (except as expressly provided herein), representing the Common Shares so purchased by such Investor and (ii) Warrants in an amount determined in accordance with Section 1.4 hereof, and the Company will deliver such stock certificates and Warrants against delivery of the purchase price as described above.

- 1.3 <u>Closing Date</u>. Subject to the satisfaction or waiver of the conditions set forth in Articles VI and VII hereof, the Closing will take place at 1:00 p.m., Central Time, on June 16, 2004, or at another date or time agreed upon by the parties to this Agreement (the "<u>Closing Date</u>"). The Closing will be held at the offices of Faegre & Benson LLP, or at such other place as the parties agree.
- 1.4 <u>Warrants</u>. In consideration of the purchase of the Common Shares by the Investors, at the Closing the Company will issue warrants to each Investor, substantially in the form of <u>Exhibit B</u> hereto (the "<u>Warrants</u>"), giving each Investor the right, for a period of two years after the Closing Date, to acquire 200 shares of Common Stock for each 1,000 shares of Common Stock purchased by such Investor on the Closing Date at a purchase price of \$4.64 per share.

ARTICLE II REPRESENTATIONS AND WARRANTIES OF INVESTORS

Each Investor represents and warrants to the Company, severally and solely with respect to itself and its purchase hereunder and not with respect to any other Investor, that:

- 2.1 <u>Investment Purpose</u>. The Investor is purchasing the Securities for its own account and not with a present view toward the public sale or distribution thereof, except pursuant to sales registered or exempted from registration under the Securities Act; <u>provided</u>, <u>however</u>, that by making the representations herein, such Investor does not agree to hold any of the Securities for any minimum or other specific term and reserves the right to dispose of the Securities at any time in accordance with or pursuant to a registration statement or an exemption from the registration requirements of the Securities Act.
- 2.2 Accredited Investor Status. The Investor is an "accredited investor" as defined in Rule 501(a) of Regulation D. The Investor has experience as an investor in securities representing an investment decision like that involved in the purchase of the Securities and acknowledges that it has the knowledge, sophistication and experience in financial and business matters as to be capable of evaluating the merits and risks of an investment in the Securities and has the ability to bear the economic risks of an investment in the Securities.
- 2.3 <u>Reliance on Exemptions</u>. The Investor understands that the Securities are being offered and sold to it in reliance upon specific exemptions from the registration requirements of United States federal and state securities laws and that the Company is relying upon the truth and accuracy of, and the Investor's compliance with, the representations, warranties, agreements,

acknowledgments and understandings of the Investor set forth herein in order to determine the availability of such exemptions and the eligibility of the Investor to acquire the Securities.

- 2.4 <u>Information</u>. The Investor and its advisors, if any, have reviewed the SEC Documents and been furnished with all materials relating to the business, finances and operations of the Company, and materials relating to the offer and sale of the Securities, that have been requested by the Investor or its advisors, if any. The Investor and its advisors, if any, have been afforded the opportunity to ask questions of the Company. Neither such inquiries nor any other due diligence investigation conducted by Investor or any of its advisors or representatives modify, amend or affect the Investor's right to rely on the Company's representations and warranties contained in Article III below. The Investor acknowledges and understands that its investment in the Securities involves a significant degree of risk, including the risks reflected in the SEC Documents, and that the market price of the Common Stock has been and continues to be volatile and that no representation is being made as to the future value of the Common Stock.
- 2.5 <u>Governmental Review</u>. The Investor understands that no United States federal or state agency or any other government or governmental agency has passed upon or made any recommendation or endorsement of the Securities or an investment therein.
 - 2.6 <u>Transfer or Resale</u>. The Investor understands that:
- (a) except as provided in the Registration Rights Agreement, the Securities have not been and are not being registered under the Securities Act or any applicable state securities laws and, consequently, the Investor may have to bear the risk of owning the Securities for an indefinite period of time because the Securities may not be transferred unless (i) the resale of the Securities is registered pursuant to an effective registration statement under the Securities Act; (ii) the Investor has delivered to the Company an opinion of counsel (in form, substance and scope customary for opinions of counsel in comparable transactions) to the effect that the Securities to be sold or transferred may be sold or transferred pursuant to an exemption from such registration; (iii) the Securities are sold or transferred to an affiliate (as defined in Rule 144) of the Investor:
- (b) any sale of the Securities made in reliance on Rule 144 may be made only in accordance with the terms of Rule 144 and, if Rule 144 is not applicable, any resale of the Securities under circumstances in which the seller (or the person through whom the sale is made) may be deemed to be an underwriter (as that term is defined in the Securities Act) may require compliance with another exemption under the Securities Act or the rules and regulations of the SEC thereunder;
- (c) except as set forth in the Registration Rights Agreement, neither the Company nor any other person is under any obligation to register the Securities under the Securities Act or any state securities laws or to comply with the terms and conditions of any exemption thereunder; and
- (d) notwithstanding anything in this Agreement or the Warrants to the contrary, the Company agrees to reregister any Common Shares or Warrant issued to an Investor

3

hereunder in the name of any partner or member of such Investor, and any such partner or member shall be deemed to be an Investor for all purposes of this Agreement and the Registration Rights Agreement, provided that any such partner or member agrees in writing to be subject to the terms of this Agreement and the Registration Rights Agreement to the same extent as of such partner or member were an original Investor hereunder and thereunder.

2.7 <u>Legends</u>. The Investor understands that until (a) the Common Shares and the Warrants may be sold by the Investor under Rule 144(k) or (b) such time as the resale of the Securities has been registered under the Securities Act as contemplated by the Registration Rights Agreement, the certificates representing the Securities will bear a restrictive legend in substantially the following form (and a stop-transfer order may be placed against transfer of the certificates for such Securities):

THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR THE SECURITIES LAWS OF ANY STATE OF THE UNITED STATES (COLLECTIVELY, THE "ACTS"). THE SECURITIES MAY NOT BE SOLD, DISTRIBUTED, OFFERED, PLEDGED, ENCUMBERED, ASSIGNED OR OTHERWISE TRANSFERRED IN THE ABSENCE OF THE FOLLOWING: (1) AN EFFECTIVE REGISTRATION STATEMENT FOR THE SECURITIES UNDER THE ACTS COVERING THE TRANSACTION, (2) THE COMPANY RECEIVES AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE ACTS, OR (3) THE COMPANY OTHERWISE SATISFIES ITSELF THAT REGISTRATION IS NOT REQUIRED UNDER THE ACTS. NOTWITHSTANDING THE FOREGOING, THE SECURITIES MAY BE PLEDGED IN CONNECTION WITH A BONA FIDE MARGIN ACCOUNT OR OTHER LOAN OR FINANCING ARRANGEMENT SECURED BY THE SECURITIES.

The legend set forth above will be removed and the Company will issue a certificate without the legend to the holder of any certificate upon which it is stamped, in accordance with the terms of Article V hereof.

- 2.8 <u>Authorization; Enforcement.</u> This Agreement and the Registration Rights Agreement have been duly and validly authorized, executed and delivered on behalf of the Investor and are valid and binding agreements of the Investor, enforceable in accordance with their terms, subject to the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the rights of creditors generally and the application of general principles of equity.
 - 2.9 <u>Residency</u>. The Investor is a resident of the jurisdiction set forth immediately below such Investor's name on the signature pages hereto.

4

2.11 No Hedging. The Investor has not established any hedge or other position in the Common Stock that is outstanding on the Closing Date which is designed to or could reasonably be expected to lead to or result in any sale of the Common Shares. For purposes hereof, a "hedge or other position" would include effecting any short sale or having in effect any short position or any purchase, sale or grant of any put option, call option or prepaid forward contract with respect to the Common Stock of the Company or with respect to any security (other than a broad-based market basket or index) that includes, relates to or derives any significant part of its value from the Common Stock.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company, and each of its Subsidiaries, as applicable, represents and warrants to the Investors that:

- 3.1 <u>Organization and Qualification</u>. The Company and its Subsidiaries are duly incorporated, validly existing and in good standing under the laws of the jurisdictions in which they are incorporated, with full power and authority (corporate and other) to own, lease, use and operate their properties, if any, and to carry on their businesses as and where now owned, leased, used, operated and conducted. The Company is duly qualified to do business and is in good standing in every jurisdiction in which the nature of the business conducted by it makes such qualification necessary, except where the failure to be so qualified or in good standing would not have a Material Adverse Effect.
- 3.2 <u>Authorization; Enforcement.</u> (a) The Company has all requisite corporate power and authority to enter into and to perform its obligations under this Agreement, the Registration Rights Agreement and the Warrants, to consummate the transactions contemplated hereby and thereby and to issue the Securities in accordance with the terms hereof and thereof; (b) the execution, delivery and performance of this Agreement, the Registration Rights Agreement and the Warrants by the Company and the consummation by it of the transactions contemplated hereby and thereby (including without limitation the issuance of the Common Shares, the Warrants and the Warrant Shares and reservation for issuance of the Warrant Shares) have been duly authorized by the Company's Board of Directors and no further consent or authorization of the Company, its Board or Directors, or its stockholders is required; (c) this Agreement, the Registration Rights Agreement and the Warrants have been duly executed by the Company; and (d) each of this Agreement, the Registration Rights Agreement and the Warrants constitutes a legal, valid and binding obligation of the Company enforceable against the Company in accordance with its respective terms, except as may be limited by any applicable bankruptcy, insolvency, reorganization, or moratorium or similar laws affecting the rights of creditors generally and the application of general principles of equity.

5

- Capitalization. As of June 14, 2004, the authorized capital stock of the Company consisted of (i) 40,000,000 shares of Common Stock, par value \$.001 per share, of which 17,161,673 shares were issued and outstanding; 7,306,652 shares were reserved for issuance under the Option Plan, of which 5,655,563 shares are subject to outstanding option grants; 318,239 shares were reserved for issuance upon exercise of outstanding warrants as set forth in Schedule 3.3; and no shares were reserved for issuance pursuant to any other securities (except as set forth under this Section 3.3); and (ii) 8,000,000 shares of preferred stock, par value \$.001 per share, of which (a) 2,200,000 shares are designated as Series A Convertible Preferred Stock, of which no shares were issued and outstanding; and (b) 2,777,500 shares are designated as Series B Convertible Preferred Stock, of which no shares were issued and outstanding. All of such outstanding shares of capital stock have been, or upon issuance will be, duly authorized, validly issued, fully paid and nonassessable. No shares of capital stock of the Company, including the Common Shares and the Warrant Shares, are subject to preemptive rights or any other similar rights of the other stockholders of the Company or any liens or encumbrances imposed through the actions or failure to act of the Company. Except as set forth above and in Schedule 3.3 and except for the transactions contemplated hereby, (i) there are no outstanding options, warrants, scrip, rights to subscribe for, puts, calls, rights of first refusal, agreements, understandings, claims or other commitments or rights of any character whatsoever relating to, or securities or rights convertible into, exercisable for, or exchangeable for any shares of capital stock of the Company, or arrangements by which the Company is or may become bound to issue additional shares of capital stock of the Company; (ii) there are no agreements or arrangements (other than the Registration Rights Agreement) under which the Company is obligated to register the sale of any of its securities under the Securities Act; and (iii) there are no anti-dilution or price adjustment provisions contained in any security issued by the Company (or in any agreement providing rights to security holders) that will be triggered by the issuance of the Securities. The Company has furnished to each Investor true and correct copies of the terms of all securities convertible into or exercisable for Common Stock of the Company and the material rights of the holders thereof in respect thereto. The Company's Certificate of Incorporation and the Company's By-laws, each as in effect on the date hereof, filed as exhibits to the Company's SEC Documents, are true and correct copies of each such document.
- 3.4 <u>Issuance of Securities</u>. The Common Shares and the Warrants have been duly authorized and, upon issuance in accordance with the terms of this Agreement, will be validly issued, fully paid and non-assessable, free from all taxes, liens, claims, encumbrances and charges with respect to the issuance thereof, will not be subject to preemptive rights or other similar rights of stockholders of the Company, and will not impose personal liability on the holders thereof. The Warrant Shares have been duly authorized and reserved for issuance, and, upon exercise of the Warrants in accordance with the terms thereof, will be validly issued, fully paid and non-assessable, and free from all taxes, liens, claims and encumbrances and charges with respect to the issuance thereof and will not be subject to preemptive rights or other similar rights of stockholders of the Company and will not impose personal liability upon the holder thereof.
- 3.5 <u>Outstanding Debt.</u> The Company has no Indebtedness for Borrowed Money (as hereinafter defined) except as otherwise set forth in <u>Schedule 3.5.</u> The Company is not in default in the payment of the principal of or interest or premium on any such Indebtedness for Borrowed

6

Money, and no event has occurred or is continuing under the provisions of any instrument, document or agreement evidencing or relating to any such Indebtedness for Borrowed Money which with the lapse of time or the giving of notice, or both, would constitute an event of default thereunder.

3.6 <u>No Conflicts; No Violation</u>.

(a) The execution, delivery and performance of this Agreement, the Registration Rights Agreement and the Warrants by the Company, and the consummation by the Company of the transactions contemplated hereby and thereby (including, without limitation, the issuance of the Common Shares, the Warrants and the Warrant Shares and reservation for issuance of the Warrant Shares) do not and will not (i) conflict with or result in a violation of

any provision of the Certificate of Incorporation or By-laws, (ii) violate or conflict with, or result in a breach of any provision of, or constitute a default (or an event which with notice or lapse of time or both could become a default) under, or give to others any rights of termination, amendment (including without limitation, the triggering of any anti-dilution provision), acceleration or cancellation of, any agreement, indenture, patent, patent license, or instrument to which the Company is a party, or (iii) result in a violation of any law, rule, regulation, order, judgment or decree (including U.S. federal and state securities laws and regulations and regulations of any self-regulatory organizations to which the Company or its securities are subject) applicable to the Company or by which any property or asset of the Company is bound or affected (except for such conflicts, breaches, defaults, terminations, amendments, accelerations, cancellations and violations as would not, individually or in the aggregate, have a Material Adverse Effect).

- (b) The Company is not in violation of its Certificate of Incorporation, By-laws or other organizational documents and the Company is not in default (and no event has occurred which with notice or lapse of time or both could put the Company in default) under, and the Company has not taken any action or failed to take any action that (and no event has occurred which, without notice or lapse of time or both) would give to others any rights of termination, amendment, acceleration or cancellation of, any agreement, indenture or instrument to which the Company is a party or by which any property or assets of the Company is bound or affected, except for possible defaults as would not, individually or in the aggregate, have a Material Adverse Effect.
- (c) The Company is not conducting, and, so long as any Investor owns any of the Securities, will not conduct, its business in violation of any law, ordinance or regulation of any governmental entity, the failure to comply with which would, individually or in the aggregate, have a Material Adverse Effect.
- (d) Except as specifically contemplated by this Agreement and as required under the Securities Act and any applicable state securities laws or any listing agreement with any securities exchange or automated quotation system, the Company is not required to obtain any consent, authorization or order of, or make any filing or registration with, any court or governmental agency or any regulatory or self regulatory agency in order for it to execute, deliver or perform any of its obligations under this Agreement, the Warrants or the Registration Rights

7

Agreement, in each case in accordance with the terms hereof or thereof, or to issue and sell the Common Shares and the Warrants in accordance with the terms hereof and to issue the Warrant Shares upon exercise of the Warrants. Except as set forth in <u>Schedule 3.6</u>, all consents, authorizations, orders, filings and registrations which the Company is required to obtain pursuant to the preceding sentence have been obtained or effected on or prior to the date hereof. The Company is not in violation of the continued listing requirements of Nasdaq and does not reasonably anticipate that the Common Stock will be delisted by Nasdaq in the foreseeable future. The Company is unaware of any facts or circumstances which might give rise to the foregoing.

SEC Documents, Financial Statements; Sarbanes-Oxley. Since January 1, 2001, the Company has timely filed all reports, schedules, forms. statements and other documents required to be filed by it with the SEC pursuant to the reporting requirements of the Exchange Act (all of the foregoing filed prior to the date hereof and all exhibits included therein and financial statements and schedules thereto and documents (other than exhibits) incorporated by reference therein, being hereinafter referred to herein as the "SEC Documents"). The Company has delivered to each Investor, or each Investor has had access to, true and complete copies of the SEC Documents, except for such exhibits and incorporated documents. As of their respective dates, the SEC Documents complied in all respects with the requirements of the Exchange Act or the Securities Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to the SEC Documents, and none of the SEC Documents, at the time they were filed with the SEC, contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. As of their respective dates, the financial statements of the Company included in the SEC Documents complied as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto. Such financial statements have been prepared in accordance with U.S. generally accepted accounting principles, consistently applied, during the periods involved (except (i) as may be otherwise indicated in such financial statements or the notes thereto, or (ii) in the case of unaudited interim statements, to the extent they may not include footnotes or may be condensed or summary statements) and fairly present in all material respects the financial position of the Company as of the dates thereof and the results of its operations and cash flows for the periods then ended (subject, in the case of unaudited statements, to normal year-end audit adjustments). Except as set forth in the financial statements included in the SEC Documents, the Company has no liabilities, contingent or otherwise, other than liabilities incurred in the ordinary course of business subsequent to March 31, 2004, and liabilities of the type not required under generally accepted accounting principles to be reflected in such financial statements. Such liabilities incurred subsequent to March 31, 2004 are not, in the aggregate, material to the financial condition or operating results of the Company. The Company is in substantial compliance with the applicable provisions of the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), and the rules and regulations promulgated thereunder, that are effective, and intends to comply substantially with other applicable provisions of the Sarbanes-Oxley Act, and the rules and regulations promulgated thereunder, upon the effectiveness of such provisions and has no reason to believe that it will not be so compliant upon such effectiveness. Without limiting the generality of the foregoing, the Chief Executive

8

Officer and the Chief Financial Officer of the Company have signed, and the Company has furnished to the SEC, all certifications required by Section 302 and Section 906 of the Sarbanes-Oxley Act; such certifications contain no qualifications or exceptions to the matters certified therein and have not been modified or withdrawn; and neither the Company nor any of its officers has received notice from any governmental entity questioning or challenging the accuracy, completeness, form or manner of filing or submission of such certifications. The Company is in compliance with all applicable provisions of the listing requirements of the Nasdaq.

- 3.8 <u>Absence of Certain Changes</u>. Except as disclosed in the SEC Documents, since March 31, 2004, there has been no material adverse change in the assets, liabilities, business, properties, operations, financial condition, prospects or results of operations of the Company or, to the knowledge of the Company, any development that could reasonably be expected to have such a material adverse change.
- 3.9 <u>Absence of Litigation</u>. There is no action, suit, claim, proceeding, inquiry or investigation before or by any court, public board, government agency, self-regulatory organization or body pending or, to the knowledge of the Company, threatened against or affecting the Company or any of its officers or directors acting as such that could, individually or in the aggregate, have a Material Adverse Effect. The Company is not aware of any facts or circumstances which would reasonably be expected to give rise to any such action or proceeding.

- 3.10 <u>Intellectual Property Rights</u>. The Company owns or possesses the licenses or rights to use all patents, patent applications, patent rights, inventions, know-how, trade secrets, trademarks, trademark applications, service marks, service names, trade names and copyrights necessary to enable it to conduct its business as now operated (and, to the best of the Company's knowledge, as presently contemplated to be operated in the future) (the "<u>Intellectual Property</u>"). There is no claim or action or proceeding pending or, to the Company's knowledge, threatened that challenges the right of the Company with respect to any Intellectual Property. To the Company's knowledge after due inquiry, the Company's current and intended products, services and processes do not infringe on any Intellectual Property or other rights held by any person. The Company is unaware of any facts or circumstances which might give rise to any of the foregoing. The Company has taken reasonable security measures to protect the secrecy, confidentiality and value of its Intellectual Property.
- 3.11 No Materially Adverse Contracts, Etc. The Company is not subject to any charter, corporate or other legal restriction, or any judgment, decree, order, rule or regulation which in the reasonable judgment of the Company's officers has or is expected in the future, individually or in the aggregate, to have a Material Adverse Effect. The Company is not a party to any contract or agreement which in the reasonable judgment of the Company's officers has or is expected to have a Material Adverse Effect.
- 3.12 <u>Tax Status</u>. Except as set forth on <u>Schedule 3.12</u>, the Company has made or filed all federal, state and foreign income and all other tax returns, reports and declarations required by any jurisdiction to which it is subject (unless and only to the extent that the Company has set aside on its books provisions reasonably adequate for the payment of all unpaid and unreported

9

taxes) and has paid all taxes and other governmental assessments and charges that are material in amount, shown or determined to be due on such returns, reports and declarations, except those being contested in good faith, and has set aside on its books provisions reasonably adequate for the payment of all taxes for periods subsequent to the periods to which such returns, reports or declarations apply. There are no unpaid taxes in any material amount claimed to be due by the taxing authority of any jurisdiction, and the Company knows of no basis for any such claim. The Company has not executed a waiver with respect to the statute of limitations relating to the assessment or collection of any foreign, federal, state or local tax. None of the Company's tax returns is presently being audited by any taxing authority.

- 3.13 Certain Transactions. Except as disclosed in the SEC Documents and except for arm's-length transactions pursuant to which the Company makes payments in the ordinary course of business upon terms no less favorable than the Company could obtain from third parties, and other than the grant of stock options, employment agreements or the ownership of other securities and rights disclosed in filings under the Exchange Act, none of the officers, directors, or employees of the Company is presently a party to any transaction with the Company (other than for services as employees, officers and directors), including any contract, agreement or other arrangement providing for the furnishing of services to or by, providing for rental of real or personal property to or from, or otherwise requiring payments to or from any officer, director or employee or, to the knowledge of the Company, any corporation, partnership, trust or other entity in which any officer, director, or employee has a substantial interest or is an officer, director, trustee or partner.
- 3.14 <u>Environmental Laws</u>. The Company (i) is in compliance with all applicable foreign federal, state and local laws and regulations relating to the protection of human health and safety, the environment or hazardous or toxic substances or wastes, pollutants or contaminants ("<u>Environmental Laws</u>"), (ii) has received all permits, licenses or other approvals required of them under applicable Environmental Laws to conduct its business and (iii) is in compliance with all terms and conditions of any such permit, license or approval where, in each of the three foregoing clauses, the failure to so comply would have, individually or in the aggregate, a Material Adverse Effect.
- 3.15 <u>Disclosure</u>. No information relating to or concerning the Company set forth in this Agreement or provided to the Investors pursuant to Section 2.4 hereof or otherwise provided in connection with the transactions contemplated hereby, including without limitation any oral or written statements made or given by the officers of the Company, or any of the Company's agents, to any Investor, or any Investor's agent, taken as a whole, contained any untrue statement of a material fact nor omitted to state any material fact necessary in order to make the statements made herein or therein, in light of the circumstances under which they were made, not misleading. No event or circumstance has occurred or information exists with respect to the Company or its business, properties, operations or financial conditions, which, under applicable law, rule or regulation, requires public disclosure or announcement by the Company but which has not been so publicly announced or disclosed (assuming for this purpose that the Company's reports filed under the Exchange Act are being incorporated into an effective registration statement filed by the Company under the Securities Act).

- 3.16 Acknowledgment Regarding Investors' Purchase of Securities. The Company acknowledges and agrees that each Investor is acting solely in the capacity of an arm's-length purchaser with respect to this Agreement and the transactions contemplated hereby. The Company further acknowledges that no Investor is acting as a financial advisor or fiduciary of the Company (or in any similar capacity) with respect to this Agreement and the transactions contemplated hereby and any statement made by any Investor or any of their respective representatives or agents in connection with this Agreement and the transactions contemplated hereby is not advice or a recommendation and is merely incidental to such Investor's purchase of the Securities and has not been relied on by the Company in any way. The Company further represents to each Investor that the Company's decision to enter into this Agreement has been based solely on an independent evaluation by the Company and its representatives.
- 3.17 No Integrated Offering. Neither the Company, nor any of its affiliates, nor any person acting on its or their behalf, has directly or indirectly made any offers or sales in any security or solicited any offers to buy any security under circumstances that would require registration under the Securities Act of the issuance of the Securities to the Investors will not be integrated with any other issuance of the Company's securities (past, current or future) for purposes of the Securities Act or any applicable rules of Nasdaq.
- 3.18 <u>No Brokers</u>. The Company has taken no action which would give rise to any claim by any person for brokerage commissions, finder's fees or similar payments relating to this Agreement or the transactions contemplated hereby.
- 3.19 <u>Permits; Compliance.</u> The Company is in possession of all franchises, grants, authorizations, licenses, permits, easements, variances, exemptions, consents, certificates, approvals and orders necessary to own, lease and operate its properties and to carry on its business as it is now being conducted, except those the failure of which to possess would not, individually or in the aggregate, have a Material Adverse Effect (collectively, the "<u>Company Permits</u>"), and there is no action pending or, to the knowledge of the Company, threatened regarding suspension or cancellation of any of the Company Permits. The Company is not in conflict with, or in default or violation of, any of the Company Permits, except for any such conflicts, defaults or

violations which, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect. Since March 31, 2004, the Company has not received any notification with respect to possible conflicts, defaults or violations of applicable laws, except for notices relating to possible conflicts, defaults or violations, which conflicts, defaults or violations would not have a Material Adverse Effect.

- 3.20 <u>Title to Property</u>. The Company has good and marketable title in fee simple to all real property and good and marketable title to all personal property owned by it which is material to the business of the Company. Any real property and facilities held under lease by the Company are held by it under valid and enforceable leases with such exceptions as would not have a Material Adverse Effect.
- 3.21 <u>Insurance</u>. The Company is insured by insurers of recognized financial responsibility against such losses and risks and in such amounts as is prudent and customary in

11

the businesses in which the Company is engaged. The Company has no reason to believe that it will not be able to renew its existing insurance coverage as and when such coverage expires or to obtain similar coverage from similar insurers as may be necessary to continue its business at a cost that would not have a Material Adverse Effect.

- 3.22 <u>Internal Accounting Controls.</u> The Company maintains a system of internal accounting controls sufficient, in the judgment of the Company's board of directors, to provide reasonable assurance that (a) transactions are executed in accordance with management's general or specific authorizations, (b) transactions are recorded as necessary to permit preparation of financial statements in conformity with generally accepted accounting principles and to maintain asset accountability, (c) access to assets is permitted only in accordance with management's general or specific authorization, (d) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences, and (e) financial reporting and the preparation of financial statements for external purposes in accordance with U.S. generally accepted accounting principles are reliable. The Chief Executive Officer and Chief Financial Officer of the Company have made all certifications required by the Sarbanes-Oxley Act and the rules and regulations promulgated thereunder, and the statements contained in any such certification are complete and correct; the Company maintains "disclosure controls and procedures" (as currently defined in Rule 13a-15(e) under the Exchange Act) in compliance with the Exchange
- 3.23 <u>Employment Matters</u>. The Company is in compliance with all federal, state, local and foreign laws and regulations respecting employment and employment practices, terms and conditions of employment and wages and hours except where failure to be in compliance would not have a Material Adverse Effect. There are no pending investigations involving the Company by the U.S. Department of Labor or any other governmental agency responsible for the enforcement of such federal, state, local or foreign laws and regulations. There is no unfair labor practice charge or complaint against the Company pending before the National Labor Relations Board or any strike, picketing, boycott, dispute, slowdown or stoppage pending or, threatened against or involving the Company. No representation question exists respecting the employees of the Company, and no collective bargaining agreement or modification thereof is currently being negotiated by the Company. No grievance or arbitration proceeding is currently pending under any expired or existing collective bargaining agreements of the Company. No material labor dispute with the employees of the Company exists or, to the knowledge of the Company, is imminent.
- 3.24 <u>Investment Company Status</u>. The Company is not and upon consummation of the sale of the Securities will not be an "investment company," a company controlled by an "investment company" or an "affiliated person" of, or "promoter" or "principal underwriter" for, an "investment company" as such terms are defined in the Investment Company Act of 1940, as amended.
- 3.25 <u>No General Solicitation</u>. Neither the Company nor any distributor participating on the Company's behalf in the transactions contemplated hereby (if any) nor any person acting for the Company, or any such distributor, has conducted any "general solicitation," as such term is defined in Regulation D, with respect to any of the Securities being offered hereby.

12

3.26 <u>Application of Takeover Protections</u>. The Company and its board of directors have taken all necessary action, if any, in order to render inapplicable any control share acquisition, business combination or other similar anti-takeover provision under the laws of the state of its incorporation which is or could become applicable to the Investors as a result of the transactions contemplated by this Agreement, including, without limitation, the Company's issuance of the Securities and the Investors' ownership of the Securities.

ARTICLE IV COVENANTS

- 4.1 <u>Best Efforts</u>. Each party will use its best efforts to satisfy in a timely fashion each of the conditions to be satisfied by it under Articles VI and VII of this Agreement.
- 4.2 Form D; Blue Sky Laws. The Company will file a Notice of Sale of Securities on Form D with respect to the Securities, as required under Regulation D, and will provide a copy thereof to each Investor promptly after such filing. The Company will, on or before the Closing Date, take such action as it reasonably determines to be necessary to qualify the Securities for sale to the Investors under this Agreement under applicable securities (or "blue sky") laws of the states of the United States (or to obtain an exemption from such qualification), and will provide evidence of any such action so taken to the Investors on or prior to the date of the Closing. The Company will file with the SEC a Current Report on Form 8-K disclosing this Agreement and the transactions contemplated hereby within five business days after the Closing Date and will make any required notice filings with state securities law authorities on a timely basis.
- 4.3 <u>Reporting Status; Eligibility to Use Form S-3.</u> The Company's Common Stock is registered under Section 12 of the Exchange Act. Throughout the Registration Period (as defined in the Registration Rights Agreement), the Company will timely file all reports, schedules, forms, statements and other documents required to be filed by it with the SEC under the reporting requirements of the Exchange Act, and the Company will not terminate its status as an issuer required to file reports under the Exchange Act even if the Exchange Act or the rules and regulations thereunder would permit such

termination. The Company currently meets, and will take all reasonably necessary action to continue to meet, the "registrant eligibility" requirements set forth in the general instructions to Form S-3.

- 4.4 <u>Use of Proceeds</u>. The Company will use the proceeds from the sale of the Securities for working capital and merger and acquisition purposes.
- 4.5 Expenses. Upon the consummation of the sale of the Common Shares and the Warrants anticipated by this Agreement, or upon failure by the parties to consummate such sales, the Company will pay the reasonable out-of-pocket expenses incurred by the Investors in connection with the transactions herein contemplated, including without limitation consulting fees incurred, research reports purchased, travel and lodging expenses incurred during the due diligence process, and the fees and out-of-pocket expenses of Faegre & Benson LLP for their services as special counsel to the Investors in connection with the transactions herein contemplated, up to an aggregate amount of \$20,000. The Company will also pay (a) all reasonable fees and expenses incurred by the Investors with respect to any amendments or

13

waivers requested by the Company (whether or not the same become effective) under or in respect of this Agreement or the agreements contemplated hereby; (b) all reasonable fees and expenses incurred by the Investors with respect to the enforcement of the rights granted under this Agreement or the agreements contemplated hereby; and (c) all reasonable fees and expenses incurred by the Investors in connection with the registration of the Registrable Securities (as defined in the Registration Rights Agreement) pursuant to Article V of the Registration Rights Agreement; provided, however, that the Company shall be obligated to pay for the reasonable fees and out-of-pocket expenses of only one legal counsel for the Investors for purposes of this clause (c) and the Company's total payment obligation under this clause (c) shall not exceed \$7,500, it being understood that the amounts in clauses (a), (b) and (c) are not included within the \$20,000 amount specified in the previous sentence, but are separate and in addition to such amount.

- 4.6 <u>Financial Statements</u>. The financial statements of the Company will be prepared in accordance with United States generally accepted accounting principles, consistently applied, and will fairly present in all material respects the consolidated financial position of the Company and results of its operations and cash flows for the periods then ended (subject, in the case of unaudited statements, to normal year-end audit adjustments).
- 4.7 <u>Listing</u>. On or before the 10th business day after the date of this Agreement, the Company will secure the listing of the Common Shares and the Warrant Shares upon each national securities exchange or automated quotation system, including without limitation the Nasdaq, if any, upon which shares of Common Stock are then listed (subject to official notice of issuance) and, so long as any Investor owns any of the Securities, will maintain such listing of all such Common Shares and Warrant Shares. The Company will use its best efforts to obtain and, so long as any Investor owns any of the Securities, maintain the listing and trading of its Common Stock on Nasdaq, the American Stock Exchange or the New York Stock Exchange and will comply in all respects with the Company's reporting, filing and other obligations under the bylaws or rules of the National Association of Securities Dealers, Inc. and such exchanges or automated quotation system, as applicable. The Company will promptly provide to each Investor copies of any notices it receives regarding the continued eligibility of the Common Stock for listing on Nasdaq or other principal exchange or quotation system on which the Common Stock is listed or traded.
- 4.8 <u>Solvency; Corporate Existence; Compliance with Law.</u> The Company (both before and after giving effect to the transactions contemplated by this Agreement) is solvent (i.e., its assets have a fair market value in excess of the amount required to pay its probable liabilities on its existing debts as they become absolute and matured) and currently the Company has no information that would lead it to reasonably conclude that the Company would not have, nor does it intend to take any action that would impair its ability to pay its debts from time to time incurred in connection therewith as such debts mature. The Company will maintain and cause each Subsidiary to maintain its corporate existence in good standing. The Company will conduct its business in compliance with all applicable laws, rules and regulations of the jurisdictions in which it is conducting business, including, without limitation, all applicable local, state and federal environmental laws and regulations, where the failure to comply with such would have a Material Adverse Effect.

- 4.9 <u>Insurance</u>. The Company will maintain liability, casualty and other insurance (subject to customary deductions and retentions) with responsible insurance companies against such risk of the types and in the amounts customarily maintained by companies of comparable size and in lines of business of the Company.
- 4.10 <u>No Integration</u>. The Company will not make any offers or sales of any security (other than the Securities) under circumstances that would cause the offering of Securities to be integrated with any other offering of securities by the Company (i) for the purpose of any stockholders approval provision applicable to the Company or its securities or (ii) for purposes of any registration requirement under the Securities Act.
- 4.11 Reservation of Shares. The Company will at all times have authorized, and reserved for the purpose of issuance, a sufficient number of shares of Common Stock to provide for the full exercise of the Warrants and the issuance of the Warrant Shares in connection therewith (based upon the exercise price of the Warrants in effect from time to time). The Company will not reduce the number of shares of Common Stock reserved for issuance upon exercise of the Warrants, without the consent of all the Investors. If at any time the number of shares of Common Stock authorized and reserved for issuance is below the Warrant Shares issued and issuable upon exercise of the Warrants (based on the exercise price of the Warrants then in effect), the Company will promptly take all corporate action necessary to authorize and reserve a sufficient number of shares, including, without limitation, calling a special meeting of stockholders to authorize additional shares to meet the Company's obligations under this Section 4.11, in the case of an insufficient number of authorized shares, and using its best efforts to obtain stockholder approval of an increase in such authorized number of shares.
- 4.12 <u>Sales by Investors</u>. Each Investor will sell any Securities sold by it in compliance with applicable prospectus delivery requirements, if any, or otherwise in compliance with the requirements for an exemption from registration under the Securities Act and the rules and regulations promulgated thereunder. No Investor will make any sale, transfer or other disposition of the Securities in violation of federal or state securities laws.

4.13 Additional Equity Capital; Right of First Refusal.

- (a) Subject to the exceptions described below, during the period beginning on the Closing Date and ending 12 months following the Closing Date, the Company will not effect any equity offering, including any debt financing with an equity component and any offering involving warrants or other securities convertible into or exchangeable for Common Stock ("Future Offering") unless it first delivers to each Investor, at least 20 days prior to the closing of such Future Offering, written notice describing the proposed Future Offering, including the terms and conditions thereof and proposed definitive documentation to be entered into in connection therewith, and providing each Investor an option during the 10-day period following delivery of such notice to purchase its pro rata share of the securities being offered in the Future Offering ("Additional Securities") on the same terms as contemplated by such Future Offering (the limitations referred to in this Section 4.13(a) is referred to as the "Capital Raising Limitation"). In the event that the terms and conditions of a proposed Future Offering are amended in any material respect after delivery of the notice to the Investors concerning the proposed Future Offering, the Company will deliver a new notice to the Investors describing the amended terms and conditions of the proposed Future Offering and each Investor thereafter will have an option, during the 10-day period following delivery of such new notice, to purchase its pro rata share of the Additional Securities on such amended terms. The foregoing sentence applies to successive amendments to the terms and conditions of any proposed Future Offering.
- (b) The Capital Raising Limitation does not apply to any transaction involving (i) issuances of securities in a firm commitment underwritten public offering (excluding a continuous offering pursuant to Rule 415 under the Securities Act), or (ii) issuances of securities as consideration for a merger, acquisition, consolidation or purchase of assets, or in connection with any strategic investments, joint venture or similar commercial relationship (the primary purpose of which is not to raise equity capital). The Capital Raising Limitation also does not apply to (x) the issuance of securities upon exercise or conversion of the Company's options, warrants or other convertible securities outstanding as of the date hereof or upon exercise of the Warrants, or (y) the grant of additional options, or the issuance of additional securities pursuant to such additional options, under the Option Plan.
- (c) An Investor's "pro rata share" is the number of shares of Additional Securities (rounded to the nearest whole share or dollar amount) as is equal to the product of (a)(i) the number of shares of Common Stock issuable upon the exercise of the Warrants held by such Investor immediately prior to the issuance of the Additional Securities being offered plus any shares of Common Stock held by such Investor, divided by (ii) the total number of shares of Common Stock issuable upon conversion of all outstanding options, warrants and other convertible securities by the Company immediately prior to the issuance of the Additional Securities being offered plus all issued and outstanding shares of Common Stock, multiplied by (b) the entire offering of Additional Securities.
- 4.14 <u>Pledge of Securities</u>. The Company acknowledges and agrees that the Securities may be pledged by an Investor in connection with a bona fide margin agreement or other loan or financing arrangement that is secured by the Securities. The pledge of Securities shall not be deemed to be a transfer, sale or assignment of the Securities hereunder, and no Investor effecting

16

a pledge of Securities shall be required to provide the Company with any notice thereof or otherwise make any delivery to the Company pursuant to this Agreement; provided that an Investor and its pledgee shall be required to comply with the provisions of Section 2.6 hereof in order to effect a sale, transfer or assignment of Securities to such pledgee. The Company hereby agrees to execute and deliver such documentation as a pledgee of the Securities may reasonably request in connection with a pledge of the Securities to such pledgee by an Investor.

- 4.15 Disclosure of Transactions and Other Material Information. The Company shall not, and shall cause each Subsidiary and each of its respective officers, directors, employees and agents, not to, provide any Investor with any material nonpublic information regarding the Company or any Subsidiary from and after the Closing Date without the express written consent of such Investor. In the event of a breach of the foregoing covenant by the Company, any Subsidiary, or its each of respective officers, directors, employees and agents, in addition to any other remedy provided herein or in the Transaction Documents, such Investor shall have the right to demand that the Company make a public disclosure, and if the Company fails to do so within five business days, the Investor may make a public disclosure, in the form of a press release, public advertisement or otherwise, of such material nonpublic information without the prior approval by the Company, each Subsidiary, or each of its respective officers, directors, employees or agents. In such event, such Investor shall provide a copy of such public disclosure to the Company at or prior to the dissemination of such disclosure to the public. No Investor shall have any liability to the Company, any Subsidiary, or any of its or their respective officers, directors, employees, stockholders or agents for any such disclosure unless such Investor acts with negligence or willful misconduct. Subject to the foregoing, neither the Company nor any Investor shall issue any press releases or any other public statements with respect to the transactions contemplated hereby without the prior approval of the other party; which approval shall not be unreasonably withheld or delayed; provided, however, that the Company shall be entitled, without the prior approval of any Investor, to make any press release or other public disclosure with respect to such transactions (i) in a Current Report on Form 8-K in compliance with the requirements of the Exchange Act, and (ii) as may otherwise be required by applicable law and regulations, including the applicable rules and regulations of the Nasdaq (provided that in the case of clause (i) each Investor shall be provided a copy of any proposed press release to be issued by the Company at least one day prior to its release).
- 4.16 <u>Company Business Review</u>. During the period beginning on the Closing Date and ending 12 months following the Closing Date, each Investor shall be entitled to participate in a quarterly teleconference meeting with the Company's senior management team, if so requested by any Investor, the purpose of which shall be to provide an in-depth review to the Investor of the Company's growth outlook and acquisition plans. Each Investor acknowledges that the Company shall have no obligation to disclose any material non-public information to the Investor during such quarterly teleconference meeting unless a binding non-disclosure agreement between the Investor and the Company is in effect at the time of the disclosures and the Investor provides express written consent to the Company to disclose material non-public information to the Investor.

17

ARTICLE V TRANSFER AGENT INSTRUCTIONS; REMOVAL OF LEGENDS

5.1 <u>Issuance of Certificates</u>. The Company shall instruct its transfer agent to issue certificates, registered in the name of each Investor or its nominee, for Common Shares and Warrant Shares in such amounts as specified from time to time by each Investor to the Company issuable upon exercise of the Warrants in accordance with their terms. All such certificates will bear the restrictive legend described in Section 2.7, except as otherwise specified in this Article V. In addition, the Company will issue irrevocable Transfer Agent Instructions to the transfer agent relating to the Common Shares and the Warrant

Shares in the form of Exhibit D hereto. The Company will not give to its transfer agent any instruction other than as described in this Article V and stop transfer instructions to give effect to Section 2.7 hereof (prior to registration of the Common Shares and the Warrant Shares under the Securities Act). Nothing in this Section 5.1 will affect in any way the Investors' obligations and agreement set forth in Section 2.7 hereof to comply with all applicable prospectus delivery requirements, if any, upon resale of the Common Shares and the Warrant Shares.

- 5.2 <u>Unrestricted Securities</u>. If, unless otherwise required by applicable state securities laws, (a) the resale of the Securities represented by a certificate has been registered under an effective registration statement filed under the Securities Act, (b) a holder of Securities provides the Company and the Transfer Agent with an opinion of counsel, in form, substance and scope customary for opinions of counsel in comparable transactions, to the effect that a public sale or transfer of such Securities may be made without registration under the Securities Act and such sale either has occurred or may occur without restriction on the manner of such sale or transfer, (c) such holder provides the Company and the Transfer Agent with reasonable assurances that such Securities can be sold under Rule 144, or (d) the Securities represented by a certificate can be sold without restriction as to the number of securities sold under Rule 144(k), the Company will permit the transfer of the Common Shares or the Warrant Shares, and the Transfer Agent will issue one or more certificates, free from any restrictive legend, in such name and in such denominations as specified by such holder. Notwithstanding anything herein to the contrary, the Securities may be pledged as collateral in connection with a bona fide margin account or other lending arrangement; provided that such pledge will not alter the provisions of this Article V with respect to the removal of restrictive legends.
- 5.3 <u>Enforcement of Provision</u>. The Company acknowledges that a breach by it of its obligations hereunder will cause irreparable harm to each Investor by vitiating the intent and purpose of the transaction contemplated hereby. Accordingly, the Company acknowledges that the remedy at law for a breach of its obligations under this Article V will be inadequate and agrees, in the event of a breach or threatened breach by the Company of the provisions of this Section, that each Investor will be entitled, in addition to all other available remedies, to an injunction restraining any breach and requiring immediate transfer, without the necessity of showing economic loss and without any bond or other security being required.

18

ARTICLE VI CONDITIONS TO THE COMPANY'S OBLIGATION TO SELL

The obligation of the Company to issue and sell the Common Shares to each Investor at the Closing is subject to the satisfaction by such Investor, on or before the Closing Date, of each of the following conditions. These conditions are for the Company's sole benefit and may be waived by the Company at any time in its sole discretion:

- 6.1 The Investor will have executed this Agreement and the Registration Rights Agreement and will have delivered those agreements to the Company.
 - 6.2 The Investor will have delivered the purchase price for the Common Shares to the Company in accordance with this Agreement.
- 6.3 The representations and warranties of the Investor must be true and correct in all material respects as of the Closing Date as though made at that time (except for representations and warranties that speak as of a specific date, which representations and warranties must be correct as of such date), and the Investor will have performed and complied in all material respects with the covenants and conditions required by this Agreement to be performed or complied with by the Investor at or prior to the Closing.
- No litigation, statute, rule, regulation, executive order, decree, ruling or injunction will have been enacted, entered, promulgated or endorsed by or in any court or governmental authority of competent jurisdiction or any self-regulatory organization having authority over the matters contemplated hereby which prohibits the consummation of any of the transactions contemplated by this Agreement.

ARTICLE VII CONDITIONS TO THE INVESTOR'S OBLIGATION TO PURCHASE

The obligation of each Investor hereunder to purchase the Common Shares from the Company at the Closing is subject to the satisfaction, on or before the Closing Date, of each of the following conditions. These conditions are for each Investor's respective benefit and may be waived by any Investor at any time in its sole discretion:

- 7.1 The Company will have executed this Agreement and the Registration Rights Agreement and each such Agreement will have been delivered the Investor.
- 7.2 The Company will have delivered to the Investors duly executed certificates representing the Common Shares and duly executed Warrants in the amounts specified in Sections 1.1 and 1.4 hereof.
- 7.3 The representations and warranties of the Company must be true and correct in all material respects as of the Closing as though made at that time (except for representations and warranties that speak as of a specific date, which representations and warranties must be true and correct as of such date) and the Company must have performed and complied in all material respects with the covenants and conditions required by this Agreement to be performed or

19

complied with by the Company at or prior to the Closing. The Investor must have received a certificate or certificates dated as of the Closing Date and executed by the Chief Executive Officer or the Chief Financial Officer of the Company certifying as to the matters in contained in this Section 7.3 and as to such other matters as may be reasonably requested by such Investor, including, but not limited to, the Company's Certificate of Incorporation, By-laws, Board of Directors' resolutions relating to the transactions contemplated hereby and the incumbency and signatures of each of the officers of the Company who may execute on behalf of the Company any document delivered at the Closing.

- 7.4 No litigation, statute, rule, regulation, executive order, decree, ruling or injunction will have been enacted, entered, promulgated or endorsed by or in any court or governmental authority of competent jurisdiction or any self-regulatory organization having authority over the matters contemplated hereby which prohibits the consummation of any of the transactions contemplated by this Agreement.
- 7.5 Trading and listing of the Common Stock on the Nasdaq must not have been suspended by the SEC or the Nasdaq, nor shall Nasdaq have notified the Company of any failure of the Company to meet any of the continued listing standards.
- 7.6 The Investors will have received an opinion of the Company's counsel, dated as of the Closing Date, in form, scope and substance reasonably satisfactory to the Investors and in substantially the form attached hereto as <u>Exhibit E</u>.
- 7.7 The Irrevocable Transfer Agent Instructions, in form and substance substantially like the form attached hereto as <u>Exhibit D</u> will have been delivered to the Company's transfer agent and acknowledged in writing by such transfer agent.

ARTICLE VIII

[This Article has been left blank intentionally]

ARTICLE IX INDEMNIFICATION

9.1 <u>Indemnification by Company.</u> In consideration of each Investor's execution and delivery of this Agreement and its acquisition of the Securities hereunder, and in addition to all of the Company's other obligations under this Agreement, the Registration Rights Agreement, and the Warrants, the Company will defend, protect, indemnify and hold harmless each Investor and each other holder of the Securities and all of their stockholders, officers, directors, employees, advisors and direct or indirect investors and any of the foregoing person's agents or other representatives (including, without limitation, those retained in connection with the transactions contemplated by this Agreement) (collectively, the "<u>Indemnitees</u>") from and against any and all actions, causes of action, suits, claims, losses, costs, penalties, fees, liabilities and damages, and expenses in connection therewith (regardless of whether any such Indemnitee is a

20

party to the action for which indemnification hereunder is sought), and including reasonable attorneys' fees and disbursements (the "Indemnified Liabilities"), incurred or suffered by an Indemnitee as a result of, or arising out of, or relating to (a) any breach of any representation or warranty made by the Company herein or in any other certificate, instrument or document contemplated hereby or thereby, (b) any breach of any covenant, agreement or obligation of the Company contained herein or in any other certificate, instrument or document contemplated hereby or thereby, or (c) any cause of action, suit or claim brought or made against such Indemnitee and arising out of or resulting from the execution, delivery, performance, breach or enforcement of this Agreement, the Registration Rights Agreement or the Warrants by the Company; provided, however, that, with respect to this clause (c), the Company shall not be liable to the extent such Indemnified Liabilities are finally determined by a court of competent jurisdiction to have resulted primarily and directly from the Investors' gross negligence or willful misconduct. To the extent that the foregoing undertaking by the Company is unenforceable for any reason, the Company will make the maximum contribution to the payment and satisfaction of each of the Indemnified Liabilities that is permissible under applicable law.

ARTICLE X DEFINITIONS

- 10.1 "Additional Shares" has the meaning set forth in Section 4.13.
- 10.2 "Capital Raising Limitation" has the meaning set forth in Section 4.13.
- 10.3 "Closing" means the closing of the purchase and sale of the Common Shares.
- 10.4 "Closing Date" has the meaning set forth in Section 1.3.
- 10.5 "Common Shares" has the meaning set forth in the recitals.
- 10.6 "Common Stock" means the common stock, par value \$.001 per share, of the Company.
- 10.7 "Company" means Perficient, Inc.
- 10.8 "Company Permit" has the meaning set forth in Section 3.19.
- 10.9 "Environmental Laws" has the meaning set forth in Section 3.14.
- 10.10 "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- 10.11 "Future Offering" has the meaning set forth in Section 4.13.
- 10.12 "Indebtedness for Borrowed Money" shall include only (i) indebtedness of the Company and its Subsidiaries incurred as the result of a direct borrowing of money, and (ii) guarantees by the Company and its Subsidiaries of indebtedness of third parties, and shall not include any other indebtedness including, but not limited to, indebtedness incurred with respect to trade accounts, equipment leases, and intercompany loans.

- 10.14 "Indemnitees" has the meaning set forth in Article IX.
- 10.15 "Intellectual Property" has the meaning set forth in Section 3.10.
- 10.16 "Investors" means the investors whose names are set forth on the signature pages of this Agreement, and their permitted transferees.
- "Material Adverse Effect" means a material adverse effect on (a) the assets, liabilities, business, properties, operations, financial condition, prospects or results of operations of the Company and its Subsidiaries, taken as a whole, or (b) the ability of the Company to perform its obligations pursuant to the transactions contemplated by this Agreement or under the agreements or instruments to be entered into or filed in connection herewith.
 - 10.18 "Nasdaq" means the Nasdaq SmallCap Market.
 - 10.19 "Option Plan" means the Company's 1999 Stock Option/Stock Issuance Plan, including all amendments thereto.
- 10.20 "Registration Rights Agreement" means the Registration Rights Agreement, dated as of the date of this Agreement and among the parties to this Agreement, in the form attached hereto as Exhibit C.
 - 10.21 "Regulation D" means Regulation D as promulgated by the SEC under the Securities Act.
 - 10.22 "Rule 144" and "Rule 144(k)" mean Rule 144 and Rule 144(k), respectively, promulgated under the Securities Act, or any successor rule.
 - 10.23 "Sarbanes-Oxley Act" has the meaning set forth in Section 3.7.
 - 10.24 "SEC" means the United States Securities and Exchange Commission.
 - 10.25 "SEC Documents" has the meaning set forth in Section 3.7.
 - 10.26 "Securities" means the Common Shares, Warrants and Warrant Shares.
 - 10.27 "Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations thereunder, or any similar successor statute.
- 10.28 "Subsidiaries" means any entity in which the Company, directly or indirectly, owns capital stock or holds an equity or similar interest, as set forth on Schedule 10.28.
 - 10.29 "Transaction Documents" shall have the meaning set forth in Section 11.16.
 - 10.30 "Warrant Shares" means the shares of Common Stock issuable upon the exercise (or otherwise) of the Warrants.

22

10.31 "Warrants" means the stock purchase warrants of the Company, in the form attached hereto as <u>Exhibit B</u>, to acquire shares of Common Stock.

ARTICLE XI GOVERNING LAW; MISCELLANEOUS

- 11.1 <u>Governing Law; Jurisdiction</u>. This Agreement will be governed by and interpreted in accordance with the laws of the State of Delaware without regard to the principles of conflict of laws. The parties hereto hereby submit to the jurisdiction of the United States federal and state courts located in the State of Delaware with respect to any dispute arising under this Agreement, the agreements entered into in connection herewith or the transactions contemplated hereby or thereby.
- 11.2 <u>Counterparts; Signatures by Facsimile</u>. This Agreement may be executed in two or more counterparts, all of which are considered one and the same agreement and will become effective when counterparts have been signed by each party and delivered to the other parties. This Agreement, once executed by a party, may be delivered to the other parties hereto by facsimile transmission of a copy of this Agreement bearing the signature of the party so delivering this Agreement.
- 11.3 <u>Headings</u>. The headings of this Agreement are for convenience of reference only, are not part of this Agreement and do not affect its interpretation.
- 11.4 <u>Severability</u>. If any provision of this Agreement is invalid or unenforceable under any applicable statute or rule of law, then such provision will be deemed modified in order to conform with such statute or rule of law. Any provision hereof that may prove invalid or unenforceable under any law will not affect the validity or enforceability of any other provision hereof.
- 11.5 <u>Entire Agreement</u>. This Agreement, the Registration Rights Agreement and the Warrants (including all schedules and exhibits thereto) constitute the entire agreement among the parties hereto with respect to the subject matter hereof and thereof. There are no restrictions, promises, warranties or undertakings, other than those set forth or referred to herein or therein. This Agreement supersedes all prior agreements and understandings among the parties hereto with respect to the subject matter hereof.
- 11.6 <u>Consents; Waivers and Amendments</u>. Except as otherwise specifically provided herein, in each case in which approval of the Investors is required by the terms of this Agreement, such requirement shall be satisfied only upon receipt of the written consent of each Investor.
- 11.7 <u>Changes, Waivers, etc.</u> Neither this Agreement nor any provision hereof may be changed, waived, discharged or terminated orally, but only by a statement in writing signed by the party against which enforcement of the change, waiver, discharge or termination is sought.

11.8 <u>Notices</u>. Any notices required or permitted to be given under the terms of this Agreement must be sent by certified or registered mail (return receipt requested) or delivered personally or by courier (including a recognized overnight delivery service) or by facsimile and will be effective five days after being placed in the mail, if mailed by regular U.S. mail, or upon receipt, if delivered personally, by courier (including a recognized overnight delivery service) or by facsimile, in each case addressed to a party. The addresses for such communications are:

If to the Company: Perficient, Inc.

1120 South Capital of Texas Highway

Suite 220, Building 3 Austin, TX 78746 Attn: John T. McDonald Facsimile: (512) 531-6011

With a copy to: Vinson & Elkins L.L.P.

2801 Via Fortuna, Suite 100 Austin, Texas 78746 Attn: J. Nixon Fox III Facsimile: (512) 236-3216

If to an Investor: To the address set forth immediately below such Investor's name on the signature pages hereto.

With a copy to: Faegre & Benson LLP

2200 Wells Fargo Center 90 South Seventh Street Minneapolis, Minnesota 55402

Attn: Mark A. Sides Facsimile: (612) 766-1600

Each party will provide written notice to the other parties of any change in its address.

Successors and Assigns. This Agreement is binding upon and inures to the benefit of the parties and their successors and assigns. The Company will not assign this Agreement or any rights or obligations hereunder without the prior written consent of the Investors, and no Investor may assign this Agreement or any rights or obligations hereunder without the prior written consent of the Company. Notwithstanding the foregoing, an Investor may assign all or part of its rights and obligations hereunder to any of its "affiliates," as that term is defined under the Securities Act, without the consent of the Company so long as the affiliate is an accredited investor (within the meaning of Regulation D under the Securities Act) and agrees in writing to be bound by this Agreement. This provision does not limit the Investor's right to transfer the Securities pursuant to the terms of this Agreement or to assign the Investor's rights hereunder to any such transferee pursuant to the terms of this Agreement.

- 11.10 <u>Third Party Beneficiaries</u>. This Agreement is intended for the benefit of the parties hereto and their respective permitted successors and assigns, and is not for the benefit of, nor may any provision hereof be enforced by, any other person.
- 11.11 <u>Survival</u>. All representations and warranties contained herein shall survive the execution and delivery of this Agreement, any investigation at any time made by the Investors or on their behalf, and the sale and purchase of the Common Shares and the Warrants and payment therefor. All statements contained in any certificate, instrument or other writing delivered by or on behalf of the Company pursuant hereto or in connection with or contemplation of the transactions herein contemplated (other than legal opinions) shall constitute representations and warranties by the Company hereunder.
- 11.12 <u>Publicity.</u> The Company and each Investor have the right to review, a reasonable period of time before issuance thereof, any press releases, or relevant portions of any SEC or Nasdaq filings or any other public statements, with respect to the transactions contemplated hereby. However, the Company may make any press release or SEC or Nasdaq filings with respect to such transactions as are required by applicable law and regulations (including NASD requirements) without the prior approval of the Investors (although the Company will make reasonable efforts to consult with the Investors in connection with any such press release prior to its release and filing).
- 11.13 <u>Further Assurances</u>. Each party will do and perform, or cause to be done and performed, all such further acts and things, and will execute and deliver all other agreements, certificates, instruments and documents, as another party may reasonably request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.
- 11.14 <u>No Strict Construction</u>. The language used in this Agreement is deemed to be the language chosen by the parties to express their mutual intent, and no rules of strict construction will be applied against any party.
- 11.15 <u>Equitable Relief.</u> The Company recognizes that, if it fails to perform or discharge any of its obligations under this Agreement, any remedy at law may prove to be inadequate relief to the Investors. The Company therefore agrees that the Investors are entitled to temporary and permanent injunctive relief in any such case without the necessity of proving actual damages.
- 11.16 <u>Independent Nature of Investors' Obligations and Rights</u>. The obligations of each Investor under any of the Warrant, Registration Rights Agreement or this Agreement (collectively, the "<u>Transaction Documents</u>") are several and not joint with the obligations of any other Investor, and no Investor shall be responsible in any way for the performance of the obligations of any other Investor under any Transaction Document. Nothing contained herein or in any Transaction Document, and no action taken by any Investor pursuant thereto, shall be deemed to constitute the Investors as a partnership, an association, a

	nvestors are in any way acting in concert or as a group with respect to such t. Each Investor shall be entitled to independently protect and enforce its rights,
	25
Investor to be joined as an additional party in any proceeding for such pur	at of the other Transaction Documents, and it shall not be necessary for any other rpose. Each Investor has been represented by its own separate legal counsel in their elected to provide all Investors with the same terms and Transaction Documents for ested to do so by the Investors.
	26
IN WITNESS WHEREOF, the undersigned Investors and the Cowritten.	ompany have caused this Agreement to be duly executed as of the date first above
	COMPANY:
	PERFICIENT, INC.
[Signatures o	By: Name: John T. McDonald Title: Chief Executive Officer and Chairman of the Board continued on next page]
	INVESTOR:
	Tate Capital Partners Fund, LLC By Tate Capital Partners, LLC Its Managing Member By Tate Capital Management, LLC Its Managing Member By:
	Kurtis J. Winters Managing Member
Aggregate Purchase Amount: \$339,900 (110,000 Shares)	
ADDRESS:	
Tate Capital Partners Fund, LLC 3600 Minnesota Drive, Suite 525 Minneapolis, MN 55435 Attn: Kurtis J. Winters	
(any notice hereunder to this Investor shall include a copy to):	
Faegre & Benson LLP 2200 Wells Fargo Center	

[Signatures continued on next page]

90 South Seventh Street Minneapolis, Minnesota 55402

Attn: Mark A. Sides

INVESTOR:

Pandora Select Partners, LP By Pandora Select Advisors, LLC Its Managing Member By AJR Financial, LLC Its Managing Member

By:	
	Name: Andrew J. Redleaf
	Its: Managing Member

Aggregate Purchase Amount: \$1,977,600 (640,000 Shares)

ADDRESS:

Pandora Select Partners, LP 3033 Excelsior Blvd., Ste 300 Minneapolis MN 55416 Attn: Andrew J. Redleaf

(any notice hereunder to this Investor shall include a copy to):

Faegre & Benson LLP 2200 Wells Fargo Center 90 South Seventh Street Minneapolis, MN 55402 Attn: Marklan M. Strefling

[Signatures continued on next page]

INVESTOR:

Sigma Opportunity Fund, LLC

By: Name: Its:

Aggregate Purchase Amount: \$154,500.00 (50,000 Shares)

ADDRESS:

Sigma Opportunity Fund, LLC 800 Third Avenue, Suite 1701 New York, New York 10022

Attn: Thom Waye

(any notice hereunder to this Investor shall include a copy to):

Moomjian & Waite, LLP 100 Jericho Quadrangle Jericho, New York 11753 Attn: Gary T. Moomjian

Exhibit A

Schedule of Investors

Investor	Common Shares	Purchase Price	
Tate Capital Partners Fund, LLC	110,000	\$	339,900.00
Pandora Select Partners, LP	640,000	\$	1,977,600.00
Sigma Opportunity Fund, LLC	50,000	\$	154,500.00
Total:	800,000	\$	2,472,000.00

Exhibit B
Form of Warrant
Exhibit C
Registration Rights Agreement
Exhibit D
Transfer Agent Instructions

Exhibit E

Form of Legal Opinion

LOAN MODIFICATION AGREEMENT

This Loan Modification Agreement (the "Amendment") is entered into as of June 8, 2004, by and between Perficient, Inc. ("Perficient") and Perficient Canada, Corp. (Canada, and collectively with Perficient, the "Borrower") and Silicon Valley Bank ("Bank"). Capitalized terms used but not otherwise defined herein shall have the same meanings as set forth in the Loan Agreement (defined below).

1. <u>DESCRIPTION OF EXISTING OBLIGATIONS</u>: Among other Obligations which may be owing by Borrower to Bank, Borrower is indebted to Bank pursuant to, among other documents, a Loan and Security Agreement, dated December 5, 2003, as may be amended from time to time, (the "Loan Agreement"). The Loan Agreement provides for, among other things, a revolving line of credit in the original principal amount of Six Million Dollars (\$6,000,000)(the "Revolving Line").

Hereinafter, all indebtedness owing by Borrower to Bank shall be referred to as the "Obligations."

DESCRIPTION OF COLLATERAL. Repayment of the Obligations is secured by the Collateral as described in the Loan Agreement.

Hereinafter, the Loan Agreement, together with all other documents securing repayment of the Obligations shall be referred to as the "Security Documents". Hereinafter, the Security Documents, together with all other documents evidencing or securing the Obligations shall be referred to as the "Existing Loan Documents".

- 3. <u>DESCRIPTION OF CHANGE IN TERMS</u>. Effective as of the date of this Amendment, the Loan Agreement is hereby amended as follows:
- A. The term "Borrower" as used in the Loan Agreement is hereby amended to include Perficient Genisys, Inc. (f/k/a Genisys Consulting, Inc.) ("Genisys").
 - B. The following provision is hereby incorporated into the Loan Agreement as Section 2.1.3:

2.1.3 Acquisition Term Loan Facility.

- (a) Subject to the terms and conditions of this Agreement, Bank agrees to lend to Borrower, from time to time prior to the Term Loan applicable Commitment Termination Date, advances (each a "Term Loan Advance" and collectively the "Term Loan Advances") in an aggregate amount not to exceed the Committed Term Loan Line. When repaid, the Term Loan Advances may not be reborrowed. The proceeds of the Term Loan Advances will be used solely to fund Permitted Acquisitions. Notwithstanding the foregoing, Bank shall make the initial Equipment Advance to Borrower to reimburse Borrower for its purchase of Eligible Equipment up to 120 days prior to the Effective Date. Each Term Loan Advance shall be considered a promissory note evidencing the amounts due hereunder for all purposes. Bank's obligation to lend hereunder shall terminate on the earlier of (i) the occurrence and continuance of an Event of Default, or (ii) the Term Loan applicable Commitment Termination Date.
- (b) To obtain a Term Loan Advance, Borrower will deliver to Bank (i) a completed supplement in substantially the form attached as Exhibit F ("Loan Supplement") signed by a Responsible Officer or his or her designee, (ii) a certificate signed by a Responsible Officer or his or her designee affirming that following such requested Term Loan Advance and following such Permitted Acquisition, based on pro-forma projections reviewed by a Responsible Officer, Borrower will remain in compliance with all representations, warranties, covenants and agreements contained in this Agreement, and (iii) such additional information as Bank may request at least five (5)

1

Business Days before the proposed funding date (the "Funding Date"). On each Funding Date, Bank will specify in the Loan Supplement for each Term Loan Advance, the Basic Rate and the Payment Dates. If Borrower satisfies the conditions of each Term Loan Advance specified herein, Bank will disburse such Term Loan Advance by internal transfer to Borrower's deposit account with Bank on the Funding Date. Term Loan Advances for any Permitted Acquisition may not exceed in the aggregate thirty percent (30%) of the Total Acquisition Cost for such Permitted Acquisition.

- (c) Bank's obligation to lend the undisbursed portion of the Committed Term Loan Line will terminate if, in Bank's sole discretion, there has been a material adverse change in the general affairs, management, results of operation, condition (financial or otherwise) or the prospects of Borrower, whether or not arising from transactions in the ordinary course of business, or there has been any material adverse deviation by Borrower from the most recent business plan of Borrower presented to and accepted by Bank prior to the execution of this Agreement.
- C. The following provision is hereby incorporated into the Loan Agreement as Section 2.3.3:

2.3.3 <u>Term Loan Payments</u>.

(a) Principal and Interest Payments On Payment Dates. Borrower will repay the Term Loan Advances on the terms provided herein and in the Loan Supplement. Borrower will make payments monthly of principal in advance and/or accrued interest, as set forth herein and in the Loan Supplement, for each Term Loan Advance (collectively, "Scheduled Payments"), commencing on the calendar date thirty (30) days following the Funding Date with respect to such Term Loan Advance and continuing thereafter during the applicable Repayment Period on the same calendar date of each calendar month (each a "Payment Date"). Borrower shall make payments of accrued but unpaid interest only on the first, second and third Payment Dates for each Term Loan Advance. Beginning on the fourth Payment Date for each Term Loan Advance and continuing thereafter during the applicable Repayment Period, Borrower will pay thirty-six (36) equal installments of principal, plus all accrued but unpaid interest for such Term Loan Advance (collectively with the interest-only payments required in the preceding sentence, each a "Term Loan Payment"). Borrower's final Term Loan Payment for each Term Loan Advance shall be due and payable on the Maturity Date for such Term Loan Advance and shall include all outstanding principal and all accrued but

unpaid interest for such Term Loan Advance. Payments received after 12:00 noon Pacific time are considered received at the opening of business on the next Business Day. A Term Loan Advance may only be prepaid in accordance with Sections 2.3.3(d).

- (b) <u>Interest Rate</u>. Borrower will pay interest on the Payment Dates (as described above) at the per annum rate of interest equal to the Basic Rate determined by Bank as of the Funding Date for each Term Loan Advance in accordance with the definition of the Basic Rate. Any amounts outstanding during the continuance of an Event of Default shall bear interest at a per annum rate equal to the Default Interest Rate, as defined in Section 2.3.1(c).
- (c) <u>Mandatory Prepayment Upon an Acceleration</u>. If the Term Loan Advances are accelerated following the occurrence of an Event of Default or otherwise, then Borrower will immediately pay to Bank (i) all outstanding principal and all accrued but unpaid interest, including interest accruing at the Default Interest Rate, with respect to each Term Loan Advance to the date of such prepayment, and (ii) all other sums, if any, that shall have become due and payable with respect to this Agreement.

2

- (d) Permitted Prepayment of Loans. Borrower shall have the option to prepay all, but not less than all, of any Term Loan Advance advanced by Bank under this Agreement, provided Borrower (i) provides written notice to Bank of its election to prepay such Term Loan Advance at least fifteen (15) days prior to such prepayment, and (ii) pays, on the date of the prepayment (A) all outstanding principal and all accrued but unpaid interest with respect to such Term Loan Advance; and (B) all other sums, if any, that shall have become due and payable hereunder with respect to this Agreement.
- D. The following provision is hereby incorporated into the Loan Agreement as Section 2.4(d):
 - (d) <u>Term Loan Facility Fee</u>. A fully earned, non-refundable facility fee for the Committed Term Loan Line in the amount of \$10,000 (the "Term Loan Fee"). Borrower shall pay Bank \$5,000 of the Term Loan Fee on the date this Agreement is amended and modified to include the Committed Term Loan Line. Borrower shall pay Bank the remaining \$5,000 of the Term Loan Fee on the Funding Date for the first advance under the Committed Term Loan Line.
- E. The following provision is hereby incorporated into the Loan Agreement as Section 6.7(c):
 - (c) <u>Debt Service Coverage</u>. At any time Borrower has any outstanding obligation under the Committed Term Loan Line, a ratio of earnings after tax plus interest, depreciation and amortization for the specified period on an annualized basis to current maturities of long term debt and capitalized leases, plus interest expense annualized of at least 1.50 to 1.00. The debt service coverage ratio shall be computed on a trailing three month basis. The Debt Service Coverage ratio shall exclude any maturities on the Revolving Line
- F. The following provisions are hereby incorporated into the end of Section 6 as Sections 6.10, 6.11 and 6.12:
 - **6.10. IBM Services Agreement.** Borrower will immediately notify Bank regarding any material change in the status or terms of that certain Agreement dated August 17, 2000, and as amended on August 28, 2003 by and between Borrower and IBM.
 - **6.11 Permitted Acquisitions.** Immediately following any Permitted Acquisition, but in no event later than fifteen (15) days following such Permitted Acquisition, Borrower shall deliver to Bank
 - (a) the original stock certificates or ownership certificates for all of the capital stock purchased by Borrower in such Permitted Acquisition, and an executed, but blank and undated, stock power certificate in the form attached hereto as Exhibit G for each certificate of capital stock; and
 - (b) a joinder agreement in the form attached hereto as Exhibit H, whereby the newly acquired Person shall agree to become a Borrower under this Agreement and the Loan Documents, and shall grant to Bank a continuing security interest in the Collateral.
 - **6.12 Permitted Acquisitions Financial Covenant.** As soon as available, but no later than thirty (30) days after the last day of the month during which the Acquisition Covenant Date occurs, Borrower shall deliver to Bank evidence satisfactory to Bank that the Person acquired in such Permitted Acquisition shall have a minimum Net Income as of the Acquisition Covenant Date of no less than one dollar (\$1.00), after giving effect to Pro Forma Adjustments.

3

G. The following definitions are hereby incorporated into Section 13.1 of the Loan Agreement:

"Acquisition Covenant Date" means the date ninety (90) days after any Permitted Acquisition Date.

"Basic Rate" is, as of the Funding Date, the per annum rate of interest (based on a year of 360 days) equal to the sum of (a) the U.S. Treasury note yield to maturity for a four-year term as quoted in The Wall Street Journal on the day the Loan Supplement is prepared, plus (b) three and one-half percent (3.5%).

"Committed Term Loan Line" is a Credit Extension of up to \$4,000,000.

"Funding Date" is any date on which an Term Loan Advance is made to or on account of Borrower.

"Maturity Date" is, with respect to each Term Loan Advance, the last day of the Repayment Period for such Term Loan Advance, or if earlier, the date of acceleration of such Term Loan Advance by Bank following an Event of Default.

"Net Income" means for any period, such Person's after-tax net income for such period on a consolidated basis, as determined in accordance with GAAP.

"**Permitted Acquisition**" means an acquisition of all or substantially all of the capital stock or property of another Person, as permitted by Section 7.3 of this Agreement.

"Permitted Acquisition Date" means the date of any Permitted Acquisition for which a Term Loan Advance is made

"Pro Forma Adjustments" means upward adjustments to Net Income made to reflect (a) an amount equal to the actual cost savings, retroactively applied to the Permitted Acquisition Date, of any operating expense that has been permanently eliminated prior to the Acquisition Covenant Date, and (b) other demonstrable adjustments, each as submitted in writing to, and approved by, Bank, in Bank's sole discretion.

"Repayment Period" as to each Term Loan Advance, thirty-nine (39) months.

"Term Loan Advance" has the meaning set forth in Section 2.1.3(a).

"Term Loan Commitment Termination Date" is June 8, 2005

"**Term Loan Fee**" has the meaning set forth Section 2.4(d)

"Total Acquisition Cost" means the total cost of any Permitted Acquisition, including all direct and indirect costs associated with such Permitted Acquisition.

- H. The definition of the term "Credit Extension" in Section 13.1 of the Loan Agreement is hereby deleted and restated as follows:
 - "Credit Extension" is each Advance, Letter of Credit, Term Loan Advance or any other extension of credit by Bank for Borrower's benefit.
- I. The Compliance Certificate attached to the Loan Agreement as Exhibit D is hereby deleted and replaced with the Compliance Certificate attached hereto as Exhibit A.

- J. The Loan Supplement attached hereto as Exhibit B is hereby incorporated into the Loan Agreement as Exhibit F to the Loan Agreement.
- K. The form of Stock Power Certificate attached hereto as Exhibit C is hereby incorporated into the Loan Agreement as Exhibit G to the Loan Agreement.
- L. The form of Joinder Agreement attached hereto as Exhibit D is hereby incorporated into the Loan Agreement as Exhibit H.
- L. <u>CONDITIONS TO CLOSING</u>. As conditions precedent to the effectiveness of this Amendment, Borrower shall have:
 - A. Executed and delivered to Lender this Amendment;
 - B. Executed, and shall have had executed by Genisys, and delivered to Bank that certain Joinder Agreement dated of even date herewith by and among Bank, Borrower and Genisys, and all other resolutions, certificates, and documents required therein;
 - C. Delivered to Bank stock certificates for all of the capital stock of Canada and Genisys and a fully-executed stock power certificate in the form attached hereto as Exhibit C for each certificate.
 - D. Paid to Bank \$5,000, which such amount shall be one-half of the Term Loan Fee; and
 - E. Delivered to Bank all resolutions, certificates or documents as Bank may request relating to the existence of Borrower, the corporate authority for the execution and delivery of this Amendment, and all other documents, instruments and agreements and any other matters relevant hereto or thereto.
- 5. <u>CONSISTENT CHANGES</u>. The Existing Loan Documents are hereby amended wherever necessary to reflect the changes described above.
- 6. NO DEFENSES OF BORROWER. Borrower agrees that, as of the date hereof, it has no defenses against paying any of the Obligations.
- 7. <u>PAYMENT OF EXPENSES</u>. Borrower shall pay Bank all out-of-pocket expenses for preparing and negotiating this Agreement not to exceed \$5,000.
- 8. <u>CHOICE OF LAW; VENUE</u>. Texas law governs this Amendment and the Loan Documents, without regard to principles of conflicts of law, as if performed entirely within the State of Texas by Texas residents. Borrower and Bank each submit to the exclusive jurisdiction of the State and Federal courts in Travis County, Texas.
- 9. <u>CONTINUING VALIDITY</u>. Borrower understands and agrees that in modifying the existing Indebtedness, Bank is relying upon Borrower's representations, warranties, and agreements, as set forth in the Existing Loan Documents. Except as expressly modified pursuant to this Agreement, the terms of the Existing Loan Documents remain unchanged and in full force and effect. Bank's agreement to modifications to the existing Obligations pursuant to this Amendment in no way shall obligate Bank to make any future modifications to the Obligations. Nothing in this Amendment shall constitute a satisfaction of the Obligations. It is the intention of Bank and Borrower to retain as liable parties all makers and endorsers of Existing Loan Documents,

unless the party is expressly released by Bank in writing. Unless expressly released herein, no maker, endorser, or guarantor will be released by virtue of this Agreement. The terms of this paragraph apply not only to this Agreement, but also to all subsequent loan modification agreements.

THIS AMENDMENT AND THE EXISTING LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR,

5

CONTEMPORANEOUS OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

EXHIBIT A

COMPLIANCE CERTIFICATE

1

EXHIBIT B - Loan Supplement

EXHIBIT F

FORM OF LOAN AGREEMENT SUPPLEMENT

1

EXHIBIT C – Form of Stock Power Certificate

IRREVOCABLE STOCK POWER

2

EXHIBIT D - Form of Joinder Agreement

PRO FORMA INVOICE - LOAN MODIFICATION AGREEMENT

1

EXHIBIT A

EXHIBIT B

INTELLECTUAL PROPERTY SECURITY AGREEMENT

3

EXHIBIT A to Intellectual Property Security Agreement

Copyrights

1

EXHIBIT B to Intellectual Property Security Agreement

Patents

5

EXHIBIT C to Intellectual Property Security Agreement

Trademarks

6

EXHIBIT D to Intellectual Property Security Agreement

Mask Works

7

EXHIBIT C

CORPORATE BORROWING RESOLUTION

FOR IMMEDIATE RELEASE

Contact: Bill Davis Perficient, Inc. (314) 995-8822 bill.davis@perficient.com

Perficient Acquires Meritage Technologies

Deal Creates Leading IBM WebSphere® Solutions Provider in Midwest

AUSTIN, Texas, June 21, 2004—Perficient, Inc. (NASDAQ: PRFT), a leading eBusiness solutions provider in the central United States, today announced it has agreed to acquire Meritage Technologies, Inc., a privately-held Midwestern eBusiness solutions firm with approximately \$12 million in annual revenues. Upon closing the acquisition, Perficient will be the leading IBM WebSphere solutions provider in the Midwest, with annualized revenues of approximately \$52 million, more than 320 consulting, technology, sales and support professionals in eight offices in the Midwest (10 in North America) and client relationships with more than 350 Global 3000 companies. The transaction, which will close today, is expected to be accretive to earnings.

"In addition to making us the unrivaled IBM WebSphere leader in the Midwest, this deal propels us through the critical \$50 million revenue mark well ahead of schedule," said Jack McDonald, chairman and chief executive of Perficient. "With our acquisition strategy and strong organic growth, we are well on our way to reaching our goal of becoming a \$100 million eBusiness leader by the end of 2006."

The acquisition of Meritage:

- Establishes Perficient as the leading IBM WebSphere solutions provider in the Midwest;
- Adds key office locations in Cincinnati, Columbus, Detroit and Indianapolis, creating a network of 10 Perficient offices throughout the Midwest and North America;
- Adds top-tier enterprise clients including, BankOne, DaimlerChrysler, Kroger, MeadWestvaco, Owens-Corning, Sara Lee, Zions Bancorp, and many
 others:
- Adds proven management leadership and more than 100 consulting, technology, sales and support professionals including a significant number of IBM WebSphere experts; and
- Increases the array of services and expertise Perficient can offer clients throughout its network.

"Perficient's track record of execution is impressive," said Scott Bok, Co-President of Greenhill & Co., the global merchant banking firm that manages Greenhill Capital Partners, which owns a controlling interest of Meritage. "They are well positioned to build additional value as IT services spending continues to rebound and IBM's offerings continue to gain share. Meritage complements Perficient's business well and we are pleased with the transaction and our position as shareholders of Perficient."

Meritage's senior managers, all IT services veterans, will assume leadership roles at Perficient. Chris Gianattasio, CEO of Meritage, will serve as general manager of

Perficient's Cincinnati, Columbus, Detroit and Indianapolis offices. He will report directly to Jeff Davis, Perficient's president and chief operating officer.

"Perficient has always had a strong reputation for excellence in the Midwest and we are delighted now to be joining forces with the company to create one of the region's leading industry players," said Gianattasio. "In addition to providing our people with new and meaningful growth opportunities, Perficient's WebSphere expertise, breadth of services, commitment to customer service and scale will bring strong benefits to Meritage's clients."

The consideration paid to selling shareholders in the transaction is approximately \$7.1 million, and includes \$2.9 million in cash and up to \$4.2 million worth of Perficient common stock, subject to satisfying certain conditions in the first three years following the closing of the transaction. Perficient will also assume \$2.4 million of debt, \$1.5 million of which is related to an accounts receivable credit facility that will be retired upon the closing of the acquisition. The operating management of Meritage will receive only Perficient common stock and Perficient stock options in the transaction. DecisionPoint International served as an advisor to Perficient on the transaction.

In connection with the acquisition, Perficient also announced that it has raised approximately \$2.5 million through the issuance of 800,000 shares of Perficient common stock to a group of institutional investors led by Tate Capital Partners. The terms of the financing include an option for the investors to purchase an additional 160,000 shares over the next 24 months through warrants for Perficient common stock, priced at 150% of their purchase price.

"We are excited to become shareholders in Perficient," said Tate Capital Partner Kurt Winters. "We believe that the Meritage acquisition will help Perficient attract the increased investor interest that it deserves, and that today's investment will strengthen the company's balance sheet and fuel its continued growth. We look forward to mutual long-term success as Perficient achieves its articulated goal of reaching \$100 million in revenues by the end of 2006."

The Meritage acquisition is the second of the year for Perficient. On April 5, 2004, the company acquired Chicago-based Genisys Consulting, a \$10 million revenue technology services firm.

About Perficient

Perficient is a leading provider of eBusiness solutions in the central United States. Perficient helps companies acquire and strengthen customer relationships, reduce costs and empower employees by creating Enabled EnterprisesTM, Web-based infrastructures with dynamically-integrated business applications that extend enterprise technology assets to customers, employees, suppliers and partners. Perficient is an award-winning "Premier Level" IBM business partner and a recognized expert in IBM's WebSphere[®] software. Perficient's other partners include leading eBusiness technology and services providers Microsoft, Stellent, Bowstreet, Wily Technology, Tibco, Mainline, Digex, Fusion and others. For more information about Perficient, which has more than 210 professionals in the Central US and Canada, please visit www.perficient.com. IBM and WebSphere are trademarks of

International Business Machines Corporation in the United States, other countries, or both.

About Meritage Technologies

Meritage Technologies provides application technology planning, design and implementation services for the Global 2000. Meritage's Portal Practice helps their clients adopt Portal Technology solutions, like IBM's WebSphere Portal Server and Bowstreet Portlet Factory, to bring business value to their organizations by reducing operating costs, improving employee productivity and tightly integrating with their suppliers and customers. Meritage's offices are located in Cincinnatti, Columbus, Detroit and Indianapolis. For more information about Meritage Technologies, please visit www.meritagetech.com.

About Greenhill & Co.

Greenhill & Co. is a leading independent investment bank that provides financial advisory and merchant banking fund management services. It acts for clients located throughout the world from its offices in New York, London and Frankfurt. For more information about Greenhill & Co., please visit www.greenhill-co.com.

About Tate Capital Partners

Tate Capital Partners is a private equity investor with a focus on the public markets. By making direct investments in small cap public companies, Tate seeks to catalyze change – enabling higher sales, higher profits, and a higher valuation multiple. For more information about Tate Capital Partners, please visit www.tatecapital.com.

Safe Harbor Statement

"Safe Harbor" statement under the Private Securities Litigation Reform Act of 1995: This news release contains forward-looking statements that are subject to risk and uncertainties. These forward-looking statements are based on management's current expectations and are subject to certain risks and uncertainties that could cause actual results to differ materially from management's current expectations and the forward-looking statements made in this press release. These risks and uncertainties include, but not limited to, the impact of competitive services, demand for services like those provided by the company and market acceptance risks, fluctuations in operating results, cyclical market pressures on the technology industry, the ability to manage strains associated with the company's growth, credit risks associated with the company's accounts receivable, the company's ability to continue to attract and retain high quality employees, accurately set fees for and timely complete its current and future client projects, and other risks detailed from time to time in the company's filings with Securities and Exchange Commission, including the most recent Form 10-KSB and Form 10-QSB. The foregoing information concerning Perficient's business outlook represents our outlook as of the date of this news release, and Perficient undertakes no obligation to update or revise any forward-looking statements whetheras a result of new developments or otherwise.